Kansas State University Libraries

New Prairie Press

NPP eBooks Monographs

4-2-2015

Human Issues in Horticulture: A Bibliography

Diana M. Farmer Kansas State University, dmfarmer@ksu.edu

Follow this and additional works at: https://newprairiepress.org/ebooks

Part of the Agriculture Commons, Horticulture Commons, Mental and Social Health Commons, Place and Environment Commons, Psychology Commons, Recreational Therapy Commons, and the Therapeutics Commons

Recommended Citation

Farmer, Diana M., "Human Issues in Horticulture: A Bibliography" (2015). *NPP eBooks*. 5. https://newprairiepress.org/ebooks/5

This Book is brought to you for free and open access by the Monographs at New Prairie Press. It has been accepted for inclusion in NPP eBooks by an authorized administrator of New Prairie Press. For more information, please contact cads@k-state.edu.

HUMAN ISSUES IN HORTICULTURE

A BIBLIOGRAPHY

DIANA M. FARMER

Human Issues in Horticulture: a Bibliography

DIANA M. FARMER



Copyright © 2015 Diana Farmer

Published by New Prairie Press, Kansas State University Libraries Manhattan, Kansas

Cover design by Kansas State University Libraries

Electronic edition available online at: http://newprairiepress.org/ebooks/

This work is licensed under a Creative Commons
Attribution 4.0 International License
http://creativecommons.org/licenses/by/4.0/

ISBN: 978-0-9915482-4-8

CONTENTS

Foreword by Candice A. Shoemaker page iv

Author's Note page vi

Introduction

page viii

1
Books
page 1

2
Pamphlets
page 63

3 Curriculum Materials page 72

4
<u>Dissertations and Theses</u>
page 81

5 <u>Media</u> page 98

6
<u>Journals, Periodicals and Serials</u>
page 103

7 <u>Titles Most Commonly Held</u> page 105

8 Index of Subject Terms page 116

About the Author page 149

FOREWORD

I am a horticulturist that has always been interested in the connection people have with plants, so much so that I have been studying the topic for over 25 years. Human issues in horticulture (HIH) is concerned with the interface between the user of horticultural crops and services and the horticulture industry and professionals (Relf, 1995). It looks at the influences of plants on people in all aspects of their lives with topics such as enabling or accessible gardens, children's and school gardening, healing landscapes, garden/plant therapy, horticultural therapy, plus economic and marketing issues. Other terms used to refer to this broad concept include socio-horticulture, people-plant interaction, and human dimensions in horticulture. This bibliography of books, pamphlets, curriculum materials, dissertations and theses, media, and journals, periodicals, and serials is the most comprehensive listing on the topic.

Look in any introductory horticulture textbook and you will find horticulture defined as the science and art of growing fruits, vegetables, flowers or ornamental plants. Thus, horticulture research and training is commonly seen as a science and technology focused on the role of horticulture in crop production. Charlie Lewis, one of the pioneers in defining a broader definition of horticulture in terms of human well-being, would often tell a wonderful story from early in his distinguished career at the Morton Arboretum that introduced him to these human values. In 1962, Charlie volunteered to assist the New York City Housing Authority in initiating a garden contest for its residents (which still continues). As a judge, he talked with the many gardeners that created beauty in areas where automobiles and buildings were regularly vandalized. What was revealed to him was "a new world, one in which the subtle human benefits of gardening perhaps outweighed the esthetic and material value of the flowers and vegetables produced" (Lewis, 1996, p.56).

Diana Farmer has done an excellent job in preparing a comprehensive bibliography of a rather broad construct. The considerable number of citations on horticultural therapy found throughout the book is a significant contribution to a profession that is evolving. However, what I think is most significant, and not found anywhere else, are the citations on gardening. These are not just 'how-to garden' references, but are garden references with a human focus. A few examples are *Gardening without sight* (Fleet, 1989) and *Prison garden book* (Flinn, 1986). Jointly with these gardening references, are references on associated topics such as universal design, accessibility, human perception and environment and behavior.

From a pioneer of human issues in horticulture, thank you Diana for completing this monumental project (and as an academic, nothing really counts until it is published).

Candice A. Shoemaker, Ph.D.
Department Head, Horticulture, Forestry, and Recreation Resources
Professor, Horticulture and Human Health
Kansas State University
Manhattan, KS USA

Lewis, C. 1996. Green Nature/Human Nature. University of Illinois Press, Urbana and Chicago, IL.

Relf, P.D. 1995. The significance of horticulture-human interaction to the horticulture industry and researchers. In: E. Matsuo and P.D. Relf, eds., Horticulture in Human Life, Culture, and Environment, A National Symposium (proceedings), Acta Horticulturae, IHC Proceedings Number 391, pp. 89-100.

AUTHOR'S NOTE

The idea for this bibliography began in 1996. A colleague (Elizabeth Deger) and I collaborated on writing a grant for funds to broaden a monograph collection that would better support the degree in horticultural therapy offered at Kansas State University. We developed a list of possible titles, compared it with titles currently in the Kansas State University Library's collections and made a prioritized list of the titles to purchase. Titles continued to be added to this list after the grant was over. In 2004, bibliographers were asked to analyze the Library's collections, especially collections that supported specific university programs. The goal was to determine the strengths/weaknesses of that collection in comparison to other educational institutions with similar programs. At that time the existing list of titles was compared to titles in the online library catalogs of Texas A&M University, Rutgers University and Virginia Tech because those institutions had strong programs in horticultural therapy. The title list was converted to a database at that time. The database included the title information as well as information on which institution(s) had a title in their collections. Additional titles were found as these catalogs were searched and these were added to the database. About this time Kansas State University obtained RefWorks, an online bibliographic tool. All titles listed in the database were added to RefWorks and additional information for each title such as subject descriptors was added. RefWorks allowed for easier de-duplication, formatting for various bibliographic styles and also allowed the bibliography to be shared with faculty and students.

The subject of horticultural therapy goes beyond just gardening/horticulture and the effect of that activity on the physical, emotional, mental, and social well-being of an individual. Texts on human physiology, cognition, philosophy, psychology, etc. were included in the bibliography that provided an improved understanding of why horticulture/gardening is an effective therapy. Hopefully the expanded coverage from academic theory to practical applications will enable

students and practitioners to broaden their skills, improve their understanding of why and how gardening can be an effective therapy and point to areas of further research.

Horticultural therapy is a helping profession. It utilizes gardening activities to initiate and achieve specific therapeutic goals for individuals. As the field has matured there has been more focus on the social and community benefits of horticulture/gardening on individuals and groups of people. The descriptive labels range from horticultural therapy to social and therapeutic horticulture to people-plant connection to human issues in horticulture. As the labels change, the focus broadens beyond the effects of gardening on an individual's physical, mental, or emotional health to include the effects upon a society and community.

INTRODUCTION

This bibliography of works about human issues in horticulture includes monographs, treatises, pamphlets and media published prior to and through 1999. It is not exhaustive nor does it include journal articles. Although the works listed are primarily in English, works in other languages are included. The compiler has not examined all items listed. The bibliography itself is divided into six chapters. Chapter 1 includes books over 50 pages in length. Chapter 2 includes pamphlets and brochures (items 50 or fewer pages in length). Chapter 3, Curriculum Materials, includes instructional and related or supportive material, including materials using advanced learning technology that is designed to strengthen the academic foundation of teaching gardening in grade levels K-12. Chapter 4 includes research reports, master's theses and doctoral dissertations written to fulfill the requirements for the attainment of a bachelor, master or doctorate degree at an educational institution. Chapter 5 includes media sound recordings, films, slides, etc. The last bibliography section, Chapter 6, is a listing of journals, periodicals, newsletters and other serials whose coverage was primarily horticultural therapy or some aspect of the topic.

The initial list of titles was compiled from a search of the library catalogs of four institutions – Kansas State University, Rutgers University, Texas A&M University, and VirginiaTech. At the time this bibliography was started, these institutions offered either a degree in horticultural therapy or an emphasis in horticultural therapy in their horticulture department. Titles held by the University of Washington, Seattle were added three years later at the request/suggestion of faculty in K-State's Department of Horticulture, Forestry and Recreational Resources. Changes and advances in technology led to an increase in the number of and access to online library catalogs. As a result, WorldCat and the library catalogs of other institutions have been used to identify the reports, theses and dissertations listed here. Throughout chapters 1-6, all works in the chapter

have been assigned unique codes for ease of cross reference or identification.

The intent of this work is to be as complete and accurate as possible for non-journal article literature for the period covered. This was not possible for Chapter 4, Theses and Dissertations. It has only been in the last 20 years that many institutions have included in their online library catalogs and sources such as WorldCat, the reports, theses and dissertations completed in fulfillment of attaining a degree. So this section is as complete as was possible at the time of the compilation. This is also true of Chapter 5, Media, especially in regard to the output of educational institutions and other organizations and associations engaged in the research or practice of the field.

1 BOOKS

- 1. Ableman, M. (1998). On good land: The autobiography of an urban farm. C. Wisehart, (Ed.). San Francisco, CA: Chronicle Books.
- 2. Abraham, G., & Abraham, K. (1975). Raise vegetables without a garden (Rev. ed.). Barrington, IL: A.B. Morse Co.
- 3. Abram, D. (1996). Spell of the sensuous: Perception and language in a more-than-human world. New York, NY: Pantheon Books.
- 4. Ackerman, D. (1990). *Natural history of the senses*. New York, NY: Random House.
- 5. Adil, J. R. (1994). Accessible gardening for people with physical disabilities: A guide to methods, tools, and plants. Bethesda, MD: Woodbine House.
- 6. Agricultural tools, equipment, machinery & buildings for farmers & ranchers with physical disabilities (1986-). West Lafayette, IN: Purdue University. Additional and updated materials for volume 2, October 26, 1992, inserted. With v. 2 (1991), last word of title is changed from 'handicaps' to 'disabilities'. Includes bibliographical references.
- 7. Agricultural tools, equipment, machinery & buildings for farmers & ranchers with physical disabilities (1991-). West Lafayette, IN: Purdue Research Foundation.
- 8. Aiello, B. (Ed.). (1976). *Places and spaces: Facility planning for handicapped children and adults.* Reston, VA: Council for Exceptional Children.
- 9. Allardice, P. (1992). Scented garden: A complete guide to growing and using fragrant plants. Pymble, New South Wales, Australia; New York, NY: Angus & Robertson; Distributed in the United States of America by HarperCollins

Publishers.

- 10. Allen, C. K. (1985). Occupational therapy for psychiatric diseases: Measurement and management of cognitive disabilities (1st ed.). Boston, MA: Little, Brown. Includes bibliographies and index
- 11. Altman, I., & Chemers, M. M. (1980). *Culture and environment*. Monterey, CA: Brooks/Cole Pub. Co.
- 12. Altman, I., & Chemers, M. M. (1984). *Culture and environment*. Monterey, CA: Brooks/Cole Pub. Col. Includes bibliographical references (p. 319-327) and indexes.
- 13. Altman, I., & Chemers, M. M. (1986). *Culture and environment*. Cambridge, England; New York, NY: Cambridge University Press.
- 14. Altman, I., Lawton, M. P., & Wohlwill, J. F. (Eds.). (1984). *Elderly people and the environment*. New York, NY: Plenum Press.
- 15. Altman, I., Rapoport, A., & Wohlwill, J. F. (Eds.). (1980). *Environment and culture*. New York, NY: Plenum Press.
- 16. Altman, I., & Tsertsman, A. (Eds.). (1994). Women and the environment. New York, NY: Plenum Press.
- 17. Altman, I., & Wohlwill, J. F. (Eds.). (1976-). Human behavior and environment: Advances in theory and research. New York, NY: Plenum Press.
- 18. Altman, I., & Wohlwill, J. F. (Eds.). (1978). *Children and the environment*. New York, NY: Plenum Press.
- 19. Altman, I., & Wohlwill, J. F. (Eds.). (1983). Behavior and the natural environment. New York, NY: Plenum Press.
- 20. American Horticultural Therapy Association. (1989). Horticulture hiring the disabled transitions project (HHD transitions): A transitions employment initiative for students with developmental disabilities. Gaithersburg, MD: American Horticultural Therapy Association. Includes bibliographical references (p. 43-44).
- 21. American Horticultural Therapy Association Annual Conference (23rd: 1995: Montreal, Quebec). (1995). *Horticulture as therapy: Proceedings AHTA Montreal* (95. D. L. Airhart, (Ed.). Gaithersburg, MD: American Horticultural Therapy Association.
- 22. American Society of Landscape Architects Foundation. (1975). Barrier free site design. Washington, DC: U.S. Dept. of Housing and Urban

Development, Office of Policy Development and Research.

- 23. American Society of Landscape Architects Foundation. (1977). Access to the environment: A series of reference documents on the design and development of site facilities to make them barrier free to the physically handicapped and disabled. Washington, DC: Dept. of Housing and Urban Development, Office of Policy Development and Research.
- 24. Ammann, R. (1999). Zauber des Gartens und was er unserer Seele schenkt. Munchen: Kosel. Includes bibliographical references (p. 173-175).
- 25. Ammon, G. B. (1972). Adventures with older adults in outdoor settings: A manual of guidance. Philadelphia, PA: Published for the Cooperative Publication Association by United Church Press. Includes bibliographical references (p. 55-57).
- 26. Andrews, V. (1990). Passion for this earth: Exploring a new partnership of man, woman & nature. San Francisco: Harper & Row. Includes bibliographical references.
- 27. Appleton, J. (1975). Experience of landscape. London, England; New York, NY: Wiley. Includes bibliographical references (p. -278).
- 28. Appleton, J. (1996). Experience of landscape (Rev. ed.). New York, NY: Wiley.
- 29. Architectural and Transportation Barriers Compliance Board. (1992). Recommendations for accessibility standards for children's environments [microform]. Washington, D.C: Architectural and Transportation Barriers Compliance Board. Includes bibliographical references.
- 30. Architectural and Transportation Barriers Compliance Board. (1994). Recommendations for accessibility guidelines: Recreational facilities and outdoor developed areas. Washington, D.C: Recreation Access Advisory Committee.
- 31. Arreaga-Mayer, C. (Ed.). (1987). Eco-behavioral approach to research in special education: The Juniper Gardens Children's Project. Lawrence, KS: University of Kansas, Bureau of Child Research. Includes bibliographical references.
- 32. Asquith, P. J., & Kalland, A. (Eds.). (1997). *Japanese images of nature: Cultural perspectives*. Richmond, Surrey: Curzon Press. Includes bibliographical references and index
- 33. Atkinson, R. E. (1973). Spot gardens: A guide for creating and planting miniature gardens, indoors and outdoors. New York, NY: McKay.

Human Issues in Horticulture: A Bibliography

- 34. Aubock, M., & Hagmuller, R. (1986). *Handbuch fur Wiener Kleingartner*. Wien: Verlag Falter. Includes bibliographical references.
- 35. Austin, D. R. (1982). *Therapeutic recreation processes and techniques*. New York, NY: Wiley. Includes bibliographies and indexes
- 36. Austin, D. R. (1994). *Comprehensive glossary of recreational therapy*. Hattiesburg, MS: American Therapeutic Recreation Association.
- 37. Austin, D. R. (1997). *Therapeutic recreation: Processes and techniques* (3rd ed.). Champaign, IL: Sagamore Pub.
- 38. Austin, D. R. (Ed.). (1980). *Therapeutic recreation curriculum: Philosophy, strategy, and concepts.* Bloomington, IN: Indiana University School of Health, Physical Education, and Recreation. Includes bibliographies
- 39. Avery, L. F. (1997). *Activity programming in long-term care*. New York, NY: Springer Pub.
- 40. Baer, D. M., & Pinkston, E. M. (Eds.). (1997). Environment and behavior. Boulder, CO: Westview Press.
- 41. Bailes, M. (1994). Healing garden. Australia: Kangaroo Press.
- 42. Baker, M. L., Gang, S. G., & O'Morrow, G. S. (1979). *Prototypical park design access for the handicapped.* Athens, GA: Institute of Community and Area Development. Includes bibliographical references (pp. 93-96).
- 43. Baldwin, B. (1988). Carry on gardening: A handbook for experienced gardeners who want to carry on gardening in spite of disablement or increasing age. Wellington, New Zealand: GP Books.
- 44. Bales, S. F. (1993). *Container gardening*. New York, NY: Prentice Hall. Includes index.
- 45. Balick, M. J. (1996). *Plants, people, and culture: The science of ethnobotany*. New York, NY: Scientific American Library. Includes bibliographical references (p. [209]-218) and index.
- 46. Ball, S. (1996). Flower remedies: A complete guide to Dr. Bach's natural healing system. Leicester, England: Blitz Editions.
- 47. Ball, S. (1998). Bach remedies workbook. Saffron Walden, England: C. W. Daniel.
- 48. Balmori, D., & Morton, M. (1993). *Transitory gardens, uprooted lives*. New Haven, CN: Yale University Press.

- 49. Barbour, N. (1995). Gardening for the less active. Chatswood, N.S.W., Australia: Reed Books. Includes index.
- 50. Barrett, M. (1992). Creating Eden: The garden as a healing space. San Francisco: HarperSanFrancisco.
- 51. Barrett, M. (1997). Creating Eden: The garden as a healing space (1st HarperCollins paperback ed.). San Francisco: HarperSanFrancisco. Includes bibliographical references.
- 52. Barrier free design: Access to and use of buildings by physically disabled people (1982). Ottawa, Canada: Public Works Canada, Design and Construction Technology. Includes indexes.
- 53. Barrier Free Environments, Inc., Architectural and Transportation Barriers Compliance Board, & Adaptive Environments Center. (1990). UFAS accessibility checklist. Washington, D.C.: The Board. Prepared for U.S. Architectural & Transportation Barriers Compliance Board; prepared by Barrier Free Environments, in conjunction with Adaptive Environments Center.
- 54. Barrier-free recreation on the Gifford Pinchot National Forest (1994). Washington, D.C.: U.S. Dept. of Agriculture, Forest Service, Pacific Northwest Region, Gifford Pinchot National Forest.
- 55. Barrier-free recreation opportunities: Huron-Manistee National Forests (1996). Washington, D.C.: Forest Service, U.S. Dept. of Agriculture.
- 56. Bartholomew, M. (1981). Square foot gardening. Emmaus, PA: Rodale Press.
- 57. Bayes, K., & Francklin, S. (Eds.). (1971). Designing for the handicapped: The mentally retarded, the mentally ill, the maladjusted, the blind, the deaf, those with learning difficulties, the gifted or exceptional child. London, England: George Godwin Limited.
- 58. Baylis, M. (1975). *Maggie Baylis on practicing plant parenthood*. San Francisco: 101 Productions. Includes index.
- 59. Bechtel, R. (1997). *Environment & behavior: An introduction*. Thousand Oaks, CA: Sage Publications. Includes bibliographical references and indexes.
- 60. Beckett, K. A., Carr, D., & Stevens, D. (1983). *Contained garden*. New York, NY: Viking Press. Includes indexes.

- 61. Beckett, K. A., Carr, D., & Stevens, D. (1993). *Contained garden: The complete guide to growing outdoor plants in pots* (1st American rev. ed.). New York, NY: Viking. Includes index
- 62. Beckett, K. A., Carr, D., & Stevens, D. (1993). *Contained garden: The complete guide to growing outdoor plants in pots* (Rev. ed.). New York, NY: Penguin Books. Includes index.
- 63. Bednar, M. J. (Ed.). (1977). *Barrier-free environments*. Stroudsburg, PA: Dowden, Hutchinson & Ross.
- 64. Beds & borders: 40 professional designs for do-it-yourselfers (1998). Tucson, AR: Home Planners. Includes index.
- 65. Beechel, J. (1975). *Interpretation for handicapped persons: A handbook for outdoor recreation personnel.* Seattle, WA: National Park Service, Pacific Northwest Region, Cooperative Park Studies Unit, College of Forest Resources. Bibliography: pp. 31-35.
- 66. Bell, G. (1994). *Permaculture garden*. London, England: Thorsons. Includes bibliographical references (p. 163-167) and index.
- 67. Bell, G., & MacCreevey, P. (1970). Behavior and environment: A bibliography of social activities in urban space. Monticello, IL: Council of Planning Librarians.
- 68. Bell, P. A., Fisher, J. D., & Loomis, R. J. (Eds.). (1978). *Environmental psychology*. Philadelphia, PA: Saunders.
- 69. Bell, P. A., & Fisher, J. D. (Eds.). (1990). *Environmental psychology* (3rd ed.). Fort Worth, TX: Harcourt Brace Jovanovich.
- 70. Bennett, J. (1997). *Container garden: Month-by-month.* Newton Abbot, Devon, United Kingdom: David & Charles. Includes bibliographical references (p. [141]) and index.
- 71. Bennett, J. (1991). *Lilies of the hearth: The historical relationship between women and plants*. Camden East, Ont., Canada: Camden House. Includes index and bibliography (p. 157-181).
- 72. Bennett, T. L. (1978). Sensory world: An introduction to sensation and perception. Monterey, CA: Brooks/Cole Pub. Co. Includes indexes and bibliography (p. 237-248).
- 73. Bernhardt, P. (1993). Natural affairs: A botanist looks at the attachments between plants and people. New York, NY: Villard Books.
- 74. Berrisford, J. M. (1974). Window box and container gardening. London,

- England: Faber.
- 75. Berry, J. S. (1975). *Hortitherapy program for substance abusers*. Clemson, SC: South Carolina Agricultural Experiment Station.
- 76. Berry, N. H., & Texas Garden Clubs, Garden Therapy Manual Committee. (1961). *National garden therapy manual* J. Hulburd, (Ed.). Philadelphia, PA: National Council Books.
- 77. Berry, S. (1995). *Contained gardens: Creative designs and projects*. Pownal, VT: Storey Communications. Includes index.
- 78. Best-Martini, E., Weeks, M. A., & Wirth, P. (1994). Long term care: Interpretation and inspiration for activity and social service professionals. Ravensdale, WA: Idyll Arbor, Inc. Includes bibliographical references (p. 367-369) and index.
- 79. Best-Martini, E., Weeks, M. A., & Wirth, P. (1996). Long term care: For activity and social service professionals (2nd ed.). Ravensdale, WA: Idyll Arbor, Inc. Includes bibliographical references (p. 389-391) and index.
- 80. Better homes and gardens container plants you can grow (1978). Des Moines, IA: Meredith Corp.
- 81. Better homes and gardens favorite houseplants and how to grow them (1976). Des Moines, IA: Meredith Corp.
- 82. Better homes and gardens. Landscape planning (1963). Des Moines, IA: Meredith Press.
- 83. Bilton, H. (1998). Outdoor play in the early years: Management and innovation. London, England: David Fulton Publishers.
- 84. Bird, R. (1998). *Beds and borders*. New York, NY: Stewart, Tabori & Chang. Includes indexes.
- 85. Bix, C. O., & Editors of Sunset Books. (1998). *Container gardening*. Menlo Park, CA: Sunset Pub. Corp. Includes index.
- 86. Bonnes, M., & Secchiaroli, G. (1995). *Environmental psychology: A psychosocial introduction* (C. Montagna, Trans.). London, England; Thousand Oaks, CA: Sage. Originally published in Italian as Psicologia ambientale. Includes index and bibliography (pp. [198]-220).
- 87. Bowlby, C. (1993). Therapeutic activities with persons disabled by Alzheimer's disease and related disorders. Gaithersburg, MD: Aspen Publishers.

Human Issues in Horticulture: A Bibliography

- 88. Bremner, E., & Posey, J. (1980s). *Children's gardens: A field guide for teachers, parents and volunteers.* Los Angeles, CA: University of California Cooperative Extension, Common Ground Garden Program. Includes bibliographical references (p. 179-187).
- 89. Brendbekken, M. (1998). Hablando con la mata: Estudio antropologicao de la interdependencia entry "la vida social de las plants" y la construccion de la identidad campesina en el area fronteriza de la Republica Dominicana. Santo Domingo, Republica Dominicana: Instituto de Medicina Dominicana. Includes bibliographical references (p. 251-266) and index.
- 90. Brennan, G., & Luebbermann, M. (1993). Little herb gardens: Simple secrets for glorious gardens--indoors and out. San Francisco: Chronicle Books.
- 91. Brett, A., Moore, R. C., & Provenzo, E. F. (1993). *Complete playground book*. Syracuse, NY: Syracuse University Press.
- 92. Brilmayer, B. (1962). All about vines and hanging plants for interior decoration, in the landscape, in container gardening. Garden City, NY: Doubleday.
- 93. Brino, G. (1982). Orti urbani a Torino: Un esperimento di autogestione. Firenze: ALINEA. Includes bibliographical references.
- 94. Bristow, A. (1988). *Gardening in retirement*. London, England: Cassell; Royal Horticultural Society.
- 95. Bristow, A. (1992). Gardening in retirement (2nd ed.). London, England: Cassell.
- 96. Britz, R. (1981). Edible city resource manual (Experimental ed.). Los Altos, CA: W. Kaufmann.
- 97. Brooklyn Botanic Garden. (1986). Beds & borders: Annuals, perennials, herbs, grasses, plant combinations, sources. Brooklyn, NY: Brooklyn Botanic Garden Record.
- 98. Brooks, C. V. W. (1974). *Sensory awareness: The rediscovery of experiencing*. New York, NY: Viking Press. Includes bibliographical references.
- 99. Brooks, H. D., & Oppenheim, C. J. (1973). *Horticulture as a therapeutic aid.* New York, NY: Institute of Rehabilitation Medicine, New York University Medical Center. Bibliography: pp. 46-47.
- 100. Brown, E. L., & Deffenbacher, K. (1979). *Perception and the senses*. New York, NY: Oxford University Press.
- 101. Bruce, H. (1999). Gardens for the senses: Gardening as therapy. Altamonte

- Springs, FL: Winner Enterprises.
- 102. Bryant, C. R., & Johnson, T. (1992). *Agriculture in the city's countryside*. London, England: Belhaven Press.
- 103. Buchanan, C. (1997). *Brother crow, sister corn.* Berkeley, CA: Ten Speed Press.
- 104. Buchanan, C. (1999). *Wildflife sanctuary garden*. Berkeley, CA: Ten Speed Press. Includes bibliographical references (p. 203-204) and index.
- 105. Burchett, M., Tarran, J., & Wood, R. A. (Eds.). (1999). *Towards a new millennium in people-plant relationships*. Sydney, Australia: University of Technology. Includes bibliographical references.
- 106. Buck, C. A. (Ed.). (1999). *Tips on growing bedding plants* (4th ed.). Columbus, OH: Ohio Florists' Association.
- 107. Budiansky, S. (1995). *Nature's keepers: The new science of nature management*. North Pomfret, VT: Trafalgar Square. Includes bibliographical references (p. 273-296) and index.
- 108. Buettner, L., & Martin, S. L. (1995). Therapeutic recreation in the nursing home. State College, PA: Venture Pub.
- 109. Bullock, C. C., & Mahon, M. J. (1997). *Introduction to recreation services for people with disabilities: A person-centered approach*. Champaign, IL: Sagamore Pub. Includes bibliographical references and index.
- 110. Burlingame, A. W. (1974). Hoe for health: Guidelines for successful horticultural therapy programs. Birmingham, MI: Bibliography: pp. 195-197.
- 111. Burlingame, J., & Blaschko, T. M. (1990-). Assessment tools for recreational therapy. Ravensdale, WA: Idyll Arbor, Inc. Includes bibliographical references.
- 112. Burlingame, J., & Blaschko, T. M. (1997). Assessment tools for recreational therapy (2nd ed.). Ravensdale, WA: Idyll Arbor, Inc. Includes bibliographical references.
- 113. Burns, G. W. (1998). *Nature-guided therapy: Brief integrative strategies for health and well-being*. Philadelphia, PA: Brunner/Mazel. Includes bibliographical references and indexes.
- 114. Bush-Brown, L. C. (1969). *Garden blocks for urban America*. New York, NY: Scribner.

- 115. Butler, R.,& Parr, H. (Eds.). (1999). Mind and body spaces: Geographies of illness, impairment and disability. London, England; New York, NY: Routledge.
- 116. Byerts, T. O., Howell, S. C., & Pastalan, L. A. (Eds.). (1979). Environmental context of aging: Life-styles, environmental quality, and living arrangements. New York, NY: Garland STPM Press.
- 117. Calkins, M. P. (1988). Design for dementia: Planning environments for the elderly and confused. Owings Mills, MD: National Health Publishing. Includes index and bibliography (p. 127-146).
- 118. Canter, D. V. (1977). *Psychology of place*. London, England: Architectural Press. Includes indexes and bibliography (p. 186-192).
- 119. Canter, D. V., & Stringer, P. (1975). *Environmental interaction: Psychological approaches to our physical surroundings*. London, England: Surrey University Press: Distributed by International Textbook Co. Includes indexes and bibliography (p. 343-362).
- 120. Canter, D. V., & Stringer, P. (1976). Environmental interaction: Psychological approaches to our physical surroundings (1st American ed.). New York, NY: International Universities Press. Includes indexes and bibliography (p. 343-362).
- 121. Canter, D. V., & Canter, S. (Eds.). (1979). *Designing for therapeutic environments: A review of research.* Chichester, England; New York, NY: J. Wiley. Includes bibliographies and indexes.
- 122. Carr, S., Francis, M., Rivlin, L., & Stone, A. (1992). *Public space*. Cambridge, MA: Cambridge University Press. Includes bibliographical references (p.317-383) and indexes.
- 123. Carrick, J. (1979). *Planting, growing, caring.* New Brunswick, NJ: Vocational-Technical Curriculum Laboratory, Rutgers—The State University.
- 124. Carstens, D. Y. (1985). Site planning and design for the elderly: Issues, guidelines and alternatives. New York, NY: Van Nostrand Reinhold. Includes index and bibliography (p. 164-165).
- 125. Carta, J. J., Greenwood, C. R., & Robinson, S. L. (1986). *Application of an eco-behavioral approach to the evaluation of early intervention programs*. Kansas City, MO: Bureau of Child Research, University of Kansas. Includes bibliographical references (leaves 43-50).

- 126. Carter, G. (1997). *Containers*. New York, NY: Stewart, Tabori & Chang. Includes index.
- 127. Carver, V., & Rodda, M. (1978). *Disability and the environment*. London, England: P. Elek. Includes index and bibliography (p. [116]-123).
- 128. Case-Smith, J., Allen, A. S., & Pratt, P. N. (Eds.) (1996). *Occupational therapy for children*, (3rd ed.). St. Louis, MO: Mosby. Includes bibliographical references and index.
- 129. CCI/Service pour l'innovation sociale, & Grimaud, V. (1982). *Jardins comme le votre: Hortitherapie et jardins de personnes handicapees* A. Mazas, (Ed.). Paris, France: Centre national d'art et de culture Georges Pompidou; Centre de création industrielle, Bibliography: p. 68.
- 130. Cecil, E. (1902). *Children's gardens*. London, England; New York, NY: Macmillan.
- 131. Cecil, E. (1902-1903). *Children's gardens*. London, England; New York, NY: Macmillan.
- 132. Cecil, E. (1903). *Children's gardens*. London, England; New York, NY: Macmillan.
- 133. Chadwick-Jones, J. K. (1979). *Brain, environment, and social psychology*. Baltimore, MD: University Park Press. Includes bibliographies and index.
- 134. Chamala, S., & Keith, K. J. (Eds.). (1995). *Participative approaches for landcare: Perspective, policies, programs*. Brisbane, Australia: Australian Academic Press. Includes bibliographical references and index.
- 135. Chancellor, P. M. (1980). Dr. Philip M. Chancellor's handbook of the Bach flower remedies. New Canaan, CT: Keats Pub. Includes bibliography (p. [244]) and index.
- 136. Chaplin, M., & Royal Horticultural Society. (1978). *Gardening for the physically handicapped and elderly*. London, England: Batsford. Includes bibliography (p. 137-138) and index.
- 137. Chaplin, M., & Royal Horticultural Society. (1980). Gardening for the physically handicapped and elderly (1st pbk. ed.). London, England: Batsford.
- 138. Chard, P. S. (1994). Healing earth: Nature's medicine for the troubled soul. Minocqua, WI: NorthWord.
- 139. Chicago Horticultural Society, Horticultural Therapy Department Staff, Rothert, E. A., & Daubert, J. R. (1981). Horticultural therapy for nursing

Human Issues in Horticulture: A Bibliography

- homes, senior centers, retirement living. Glencoe, IL: Chicago Horticultural Society. Includes bibliography (p. 126-128).
- 140. Chigier, E. (1970). Use of a group approach in the rehabilitation of severely retarded adolescents in agriculture in Israel: Final report. Tel Aviv, Israel: AKIM Israel Association for Rehabilitation of Mentally Handicapped.
- 141. Children and the environment (1990). New York, NY: United Nations Children's Fund. Includes bibliographical references (p. 67-73)
- 142. Children and the environment: Report of the Executive Director (1990). New York, NY: United Nations Children's Fund. Includes bibliographical references (p. 106-113)
- 143. *Children's play yards* (1994). Menlo Park, CA: Sunset Pub. Corp. Includes index.
- 144. Classen, C. (1993). Worlds of sense: Exploring the senses in history and across cultures. London, England; New York, NY: Routledge. Includes bibliographical references and index
- 145. Clinebell, H. J. (1996). *Ecotherapy: Healing ourselves, healing the Earth*. New York, NY: Haworth Press. Includes bibliographical references (p. 272-288) and index.
- 146. Cloet, A., & Underhill, C. (1982). *Gardening is for everyone: A week-by-week guide for people with handicaps*. London, England: Souvenir Press.
- 147. Cloet, A., & Underhill, C. (1990). Gardening is for everyone: A week-by-week guide for people with handicaps (Repr. with corrections ed.). London, England: Souvenir Press.
- 148. Cochrane, G. M., & Wilshere, E. R. (1983). Equipment for the disabled. Leisure and gardening G.M. Cochrane, & E. R. Wilshere (Eds.). Oxford, Oxfordshire, Great Britain: Oxfordshire Health Authority. Includes bibliographies and index.
- 149. Coe, M. L. (1978). *Growing with community gardening*. Taftsville, VT: Countryman Press. Includes bibliography (p. 144-145) and index.
- 150. Cohen, M. J. (1997). Reconnecting with nature: Finding wellness through restoring your bond with the earth. Corvallis, OR: Ecopress. Includes bibliographical references (p. [229]-220)
- 151. Colavita, F. B. (1978). Sensory changes in the elderly. Springfield, IL: Thomas.

- 152. Colborn, N. (1990). Container garden (1st U.S. ed.). Boston, MA: Little, Brown.
- 153. Colborn, N. (1992). *Container garden*. London, England: Conran Octopus.
- 154. Colborn, N. (1995). *Container garden.* Avenel, NJ: Crescent Books. Includes index.
- 155. Colby, D. (1987). City gardening: Planting, maintaining, and designing the urban garden K. Druse, M.C. Gleason, & H. Van Zelst (Eds.). New York, NY: Simon & Schuster. Includes bibliography (p. 93).
- 156. Cole, R. (1997). *Potted gardens: A fresh approach to container gardening*. New York, NY: C. Potter. Includes bibliographical references (p. 188-189) and index.
- 157. Coleman, E. (1989). New organic grower: A master's manual of tools and techniques for the home and market gardener. Chelsea, VT: Chelsea Green Pub. Co. Includes bibliographical references (p. 225-235).
- 158. Coleman, E. (1992). New organic grower's four-season harvest: How to harvest fresh organic vegetables from your home garden all year long. Post Mills, VT: Chelsea Green Pub. Includes bibliographical references (p. 204-205) and index.
- 159. Coleman, E. (1995). New organic grower: A master's manual of tools and techniques for the home and market gardener (Rev. and expanded ed.). White River Junction, VT: Chelsea Green Pub. Co. Includes bibliographical references (p. 326-334) and index.
- 160. Community Environmental Council, & Evarts, J. (1976). Agriculture in the city: El Mirasol Educational Farm. Santa Barbara, CA: Community Environmental Council. Includes bibliography (p. 73-74).
- 161. Community gardening procedural manual (1974). Burlington, VT: Gardens for All, Inc.
- 162. Community gardens: A growing resource (1998). Bristol, England: National Federation of City Farms. Includes bibliographical references.
- 163. Conference on Occupational Therapy for the Multiply Handicapped Child (1965: Chicago, Ill.). (1965). Occupational therapy for the multiply handicapped child; proceedings W. L. West, (Ed.). Chicago, IL: Board of Trustees of the Univ. of Illinois.
- 164. Container gardening encyclopedia (1998). Godalming, Surrey, England; New

- York, NY: CLB International; Distributed in the US by Quadrillion Pub.
- 165. Contini, R. (Ed.). (1958). Contributions of the physical, biological, and psychological sciences in human disability. New York, NY: New York Academy of Sciences. Includes bibliographical references.
- 166. Cooper Marcus, C., & Barnes, M. (1995). Gardens in healthcare facilities: Uses, therapeutic benefits, and design recommendations. Martinez, CA: Center for Health Design. Includes bibliographical references (p. 69-70).
- 167. Cooper Marcus, C., & Francis, C. (1998). *People places: Design guidelines for urban open space* (2nd ed.). New York, NY: John Wiley & Sons, Inc. Includes bibliographical references and index.
- 168. Cooper Marcus, C., & Barnes, M. (Eds.). (1999). *Healing gardens: Therapeutic benefits and design recommendations*. New York, NY: Wiley. Received the 2000 EDRA/Places "place research" award.
- 169. Copeland, M. E., Ford, L., & Solon, N. (1976). Occupational therapy for mentally retarded children: Guidelines for occupational therapy aides and certified occupational therapy assistants. Baltimore, MD: University Park Press. Includes index and bibliography: p. 223-224.
- 170. Corbett, L. C. (1905). School garden. Washington, DC: Gov't. Printing Office.
- 171. Corbett, L. C. (1913). *Garden farming*. Boston,MA; New York etc.: Ginn and company.
- 172. Corbett, L. C. (1922). *School garden* (Rev. ed.). Washington, DC: U.S. Dept. of Agriculture.
- 173. Coren, S., Porac, C., & Ward, L. M. (1979). *Sensation and perception*. New York, NY: Academic Press. Includes bibliography and indexes.
- 174. Coren, S., Porac, C., & Ward, L. M. (1984). Sensation and perception. Orlando, FL: Academic Press. Includes indexes and bibliography (p. 519-569).
- 175. Coren, S., Ward, L. M., & Enns, J. T. (1994). *Sensation and perception* (4th ed.). Fort Worth, TX: Harcourt Brace College Publishers. Includes bibliographical references (p. [649]-716) and indexes.
- 176. Corso, J. F. (1967). Experimental psychology of sensory behavior. New York, NY: Holt, Rinehart and Winston.
- 177. Cottom-Winslow, M. (1991). International landscape design: Architecture of

- gardens, parks, playgrounds & open spaces. New York, NY: PBC International.
- 178. Covington, G., & Hannah, B. (1997). Access by design. New York, NY: Van Nostrand Reinhold.
- 179. Cox, J. (1993). Creating a garden for the senses. New York, NY: Abbeville Press.
- 180. Craft, S. F., & Harris, W. W. (1996). *Hortitherapy and history: A botanical tour of S.C. State Hospital.* Columbia, SC: S.C. Department of Mental Health.
- 181. Craig Hospital (Englewood, CO). (1988). Raised bed gardening: A resource manual for patients, families, and professionals. Englewood, CO: Craig Hospital. Includes bibliography (p. 59-60).
- 182. Crandall, C., & Crandall, B. (1995). *Movable harvests: Fruits, vegetables, berries: The simplicity & bounty of container gardens.* Shelburne, VT; Buffalo, NY: Chapters. Includes bibliographical references (p. 122) and index.
- 183. Crandall, C., & Crandall, B. (1997). *Creating privacy in the garden*. New York, NY: Rizzoli. Includes bibliographical references (p. 140-141) and index.
- 184. Crandall, C., & Crandall, B. (1998). Gardener's guide to planters, containers & raised beds. New York, NY: Sterling Publ. Co.
- 185. Crisp, B. (1998). Human spaces: Life-enhancing designs for healing, working, and living. Gloucester, MA; Cincinnati, OH: Rockport Pub.
- 186. Crockett, J. U. (1979). *Crockett's Tool shed.* Boston, MA: Little, Brown. Includes index.
- 187. Crouch, D., & Ward, C. (1997). *Allotment: Its landscape and culture.* Nottingham, United Kingdom: Five Leaves.
- 188. Cunningham, D. (1992). Flower remedies handbook: Emotional healing & growth with Bach & other flower essences. New York, NY: Sterling Pub. Co. Includes bibliographical references (p. 181-182) and index.
- 189. Damian, P. (1995). Aromatherapy: Scent and psyche: Using essential oils for physical and emotional well-being. Rochester, VT: Healing Arts Press.
- 190. Dannenmaier, M. (1998). Child's garden: Enchanting outdoor spaces for children and parents. New York, NY: Simon & Schuster.
- 191. Davern, J. M. (Ed.). (1976). Places for people: Architects design hotels, motels, restaurants, bars, clubs, community recreation facilities, camps, parks, plazas,

- playgrounds. New York, NY: McGraw-Hill.
- 192. Davis, J. M. (1975). *About community gardening* J. M. Davis, (Ed.). Columbia, MD: Bridge Pub. Co. Includes index.
- 193. Davis, M. (1997). Balcony, terrace, and patio gardening. Golden, CO: Fulcrum Pub.
- 194. Davis, W. M. (1981). *Aids to make you able* (3rd ed.). New York, NY: Beaufort Books. Includes bibliography (p. 149-152).
- 195. Davis, W. M. (1981). Aids to make you able: Self-help devices and ideas for the disabled (Rev. and expanded 3rd ed.). Don Mills, Ontario, Canada: General Pub. Co. Includes bibliography (p. 149-152)
- 196. Deans, E. (1991). Leaves of life: Creating therapy gardens for people with disabilities. North Ryde, N.S.W., Australia: Angus & Robertson. Includes index.
- 197. DeCristoforo, R. J. (Ed.). (1976). Wood projects for the garden. San Francisco: Ortho Book Division, Chevron Chemical Co. Includes index.
- 198. Dennery, P. (1995). *Cities feeding people project fact sheets*. Ottawa, Canada: International Development Research Centre.
- 199. Dennis, L. (1994). Garden for life: Horticulture for people with special needs. Saskatoon, Canada: University Extension Press, Extension Division, University of Saskatchewan.
- 200. Dennison, S. T. (1988). Activities for adolescents in therapy: A handbook of facilitating guidelines and planning ideas for group therapy with troubled adolescents. Springfield, IL: Thomas. Includes bibliography (p. 163-171).
- 201. Desmond, R. (1995). Kew: The history of the Royal Botanic Gardens. London, England: Harvill Press with the Royal Botanic Gardens, Kew.
- 202. Deutsch, F. (1963). Body, mind, and the sensory gateways. New York, NY: Basic Books.
- 203. Diamond, D. (1982). Living with the flowers: A guide to bringing flowers into your daily life. New York, NY: Morrow.
- 204. Dixon, J. T. (1981). Adapting activities for therapeutic recreation service: Concepts & applications. San Diego, CA: Campanile Press, San Diego State University. Includes bibliography (p. 76).
- 205. Don, M. (1997). Sensuous garden. New York, NY: Simon & Schuster

- Editions. Includes index.
- 206. Donaldson, S. (1997). Hanging baskets. New York, NY: Lorenz Books.
- 207. Donaldson, S. (1999). Boxes, baskets, pots & planters: A practical guide to 100 inspirational containers (Updated ed.). London, England: Lorenz.
- 208. Donaldson, S. (1999). *Country containers*. Newton Abbot: David & Charles.
- 209. Donaldson, S., & Berry, S. (1998). *Plants for small spaces*. North Pomfret, VT: Trafalgar Square. Includes index.
- 210. Dorsey, J. (1978). *Introducing your kids to the outdoors*. Boston, MA: Stone Wall Press. Includes bibliography (p. 123-126).
- 211. Dowling, J. R. (1995). Keeping busy: A handbook of activities for persons with dementia. Baltimore, MD: Johns Hopkins University Press.
- 212. Drake, S. Y., & Lawrence, R. L. (1976). Recreational community gardening. Ann Arbor, MI: U.S. Bureau of Outdoor Recreation, Lake Central Region.
- 213. Dr. Edward Bach Centre. (Ed.). (1997). *Bach flower remedies*. (Updated, rev. and enlarged ed.). New Canaan, CN: Keats Pub.
- 214. Duensing, E., & Brune, C. (1979). *Urban community gardens*. Monticello, IL: Vance Bibliographies.
- 215. Easy path to gardening: For all who want to go on enjoying a rewarding hobby (1972). London, England: Reader's Digest Association; Disabled Living Foundation.
- 216. Edinburgh. University. Dept. of Urban Design and Regional Planning. Planning Research Unit. (1969). *Planning for disabled people in the urban environment*. London, England: Central Council for the Disabled.
- 217. Editorial staffs of Sunset books and magazine. (Ed.). (1959). *Gardening in containers*. Menlo Park, CA: Lane Books.
- 218. Editors of Sunset Books. (Ed.). *Children's play areas* (1998). Menlo Park, CA: Sunset Books, Inc.
- 219. Editors of Sunset Books and Sunset Magazine. (Ed.). (1984). *Container gardening*. Menlo Park, CA: Lane Pub. Co. Includes indexes.
- 220. Editors of Sunset magazine and Sunset books. (1967). Gardening in containers. (2d ed.). Menlo Park, CA: Lane Books.

- 221. Edmonds, J., & Browse, P. M. (1980). *Container plant manual*. London, England: Grower Books.
- 222. Edwards, C. H. (1991). Human ecology: Interactions of man with his environments: An introduction to the academic discipline of human ecology. Dubuque, IA: Kendall/Hunt.
- 223. Egziabher, A. G. (1994). Cities feeding people: An examination of urban agriculture in East Africa. Ottawa, Canada: International Development Research Centre. Includes bibliographical references (p. [117]-128) and index.
- 224. Elliott, P. (1978). *Gardening and the handicapped child*. London, England: Disabled Living Foundation. Includes index and bibliography (p. 190-197).
- 225. Elliott, R. (1971). Life and leisure for the physically handicapped. London, England: Elek.
- 226. Euser, B. J. (1987). *Take 'em along: Sharing the wilderness with your children*. Evergreen, CO: Cordillera Press. Includes index and bibliography (p. [109]-111).
- 227. Evans, P. S., & Donnelly, B. (1993). *Accessible landscapes: Designing for inclusion*. San Francisco, CA: Dept. of Plant Operations, San Francisco State University.
- 228. Fairchild, T. (1722). City gardener. containing the most experienced method of cultivating and ordering such ever-greens, fruit-trees, flowering shrubs, flowers, exotick plants, &c. as will be ornamental, and thrive best in the London gardens. London, England: printed for T. Woodward, and J. Peele.
- 229. Fairchild, T. (1760). London gardener. containing the most experienced method of cultivating and ordering such ever-greens, ...exotick plants, &c. as will be ornemental [sic], and thrive best in the London garden. London, England: printed for R. Davis.
- 230. Fairfax, K. (1999). Tranquil garden: Creating peaceful spaces outdoors. New York, NY: Hearst Books. Includes index.
- 231. Federation for the Promotion of Horticulture for the Disabled. (1980). *Proceedings 1980 conference* D.R. Evans, (Ed.). Bath, England: University of Bath.
- 232. Federation for the Promotion of Horticulture for the Disabled. (1981). 1981 proceedings D. R. Evans, (Ed.). Bath, England: University of Bath. Includes bibliographical references.

- 233. Feenstra, G. W., McGrew, S., & Campbell, D. (1999). *Entrepreneurial community gardens: Growing food, skills, jobs and communities.* Oakland, CA: University of California, Division of Agriculture and Natural Resources.
- 234. Feimer, N. R., & Geller, E. S. (Eds.). (1983). Environmental psychology: Directions and perspectives. New York, NY: Praeger.
- 235. Fischer, K. (1997). Green places in small spaces: A practical guide to designing and planting a small-space garden. Newtown, CT: Taunton Press. Includes index.
- 236. Fisher, J. D., Bell, P. A., & Baum, A. (1984). *Environmental psychology* (2nd ed.). New York, NY: Holt, Rinehart, and Winston. Includes indexes and bibliography (p. [407]-447).
- 237. Fisher, S. (1996). *Hanging garden: Creative displays for every garden*. North Pomfret, VT: Trafalgar Square. Includes bibliographical references (p. 157) and index.
- 238. Flagler, J., & Poincelot, R. P. (Eds.). (1994). *People-plant relationships: Setting research priorities*. New York, NY: Food Products Press.
- 239. Flanagan, T. R. (1973). *Growing food & flowers in containers*. Charlotte, VT: Garden Way Pub. Co. Includes bibliography (p. 109-110).
- 240. Fleet, K. (1982). *Gardening without sight*. London, England: Royal National Institute for the Blind.
- 241. Fleet, K. (1989). *Gardening without sight*. Royal National Institute for the Blind.
- 242. Flinn, N. (1986). *Prison garden book*. Burlington, VT: National Gardening Association.
- 243. Floyd, J. E. (1995). Sociological portrait of the homeless population in a moderate sized city: Macon, Georgia. Lewiston, NY: E. Mellen.
- 244. Floyd, J. A., & Senn, T. L. (197?). Investigation into the physical and psychological response of the visually handicapped to some selected woody and herbaceous plant materials. Clemson, SC: South Carolina Agricultural Experiment Station.
- 245. Fogg, G. E., & Fulton, R. F. (1994). Leisure site guidelines for people over 55. Arlington, VA: National Recreation and Park Association.
- 246. Fogg, J. (1999). *Creating privacy in the garden*. London, England: Ward Lock. Includes index.

- 247. Fordham, R., & Biggs, A. G. (1985). *Principles of vegetable crop production*. London, England: Collins. Includes bibliographies and index.
- 248. Foster, P. M. (Ed.). (1986). *Therapeutic activities with the impaired elderly*. New York, NY: Haworth Press. Includes bibliographies.
- 249. Fountain, K. (1982). Community gardening: A study of their feasibility and potential benefits to the Waco community and its citizens. Waco, TX: Kathy Fountain.
- 250. Fowler, M. L. C. (1978). *Growing with community gardening*. Countryman Press: Taftsville, VT. Includes index and bibliography (p.144-145).
- 251. Fox, M. W. (1980). *One earth, one mind.* New York, NY: Coward, McCann & Geoghegan. Includes bibliography (p. 263-264).
- 252. Fox, T., Koeppel, I., & Kellam, S. (1985). *Struggle for space: The greening of New York City, 1970-1984*. New York, NY: Neighborhood Open Space Coalition. Includes indexes and bibliography (p. 154-162).
- 253. Francis, M., & Hester, R. T. (Eds.). (1990). Meaning of gardens: Idea, place, and action. Cambridge, MA: MIT Press.
- 254. Francis, M., & Hester, R. T., Jr. (Eds.). (1987). Meanings of the garden: Proceedings of a working conference to explore the social, psychological and cultural dimensions of gardens: University of California, Davis, May 14-17, 1987. Davis, CA: University of California, Davis. Center for Design Research. Includes bibliographical references.
- 255. Francis, M., Lindsey, P., & Rice, J. S. (Eds.). (1994). Healing dimensions of people-plant relations: Proceedings of a research symposium: March 24-27, 1994, University of California, Davis. Davis, CA: Center for Design Research, Dept. of Environmental Design, University of California.
- 256. Freeman, D. B. (1991). *City of farmers: Informal urban agriculture in the open spaces of Nairobi, Kenya*. Montreal, Canada: McGill-Queen's University Press. Includes bibliographical references (p. [149]-153) and index.
- 257. French, S. (Ed.). (1994). On equal terms: Working with disabled people. Oxford, England; Boston, MA: Butterworth-Heinemann.
- 258. Frost, J. L., & Klein, B. L. (1979). *Children's play and playgrounds*. Boston, MA: Allyn and Bacon.
- 259. Funston, R., & Richards, P. A. (1988). Garden programme for everyone including blind, visually impaired & disabled people. Brighton Beach, Victoria,

- Australia: Association for the Blind. Includes bibliographical references (p. 59-60).
- 260. Furst Survey Research Center. (1967). Psychology of homeowner reaction to landscaping; a horticultural research institute summary. Washington, DC: Horticultural Research Institute.
- 261. Gallagher, W. (1994). Power of place: How our surroundings shape our thoughts, emotions, and actions (1st HarperPerennial ed.). New York, NY: HarperPerennial.
- 262. Gallon, C., Oxley, P. R., & Simms, B. (1991). *Tactile footway surfaces for the blind*. Crowthorne, Berkshire: Transport and Road Research Laboratory. Includes bibliographical references (p. 59).
- 263. Garden therapy manual (1998). (Rev. 1998 ed.). St. Louis, MO: National Garden Clubs, Inc. Includes bibliographical references.
- 264. Garden tools & equipment (1995). Newton, CT: Taunton Press. Includes index.
- 265. Gardening: School, community, home (1940). New York, NY: National Recreation Association.
- 266. Gardening in small spaces (1989). Alexandria, VA: Time-Life Books. Includes bibliographical references (p. 153) and index.
- 267. Gardening with children: A handbook (1984). Brooklyn, NY: Brooklyn Botanic Garden. Includes bibliographical references.
- 268. Gardening, mental health and community care: Seminar '95 proceedings (1995). Frome, Somerset: Horticultural Therapy.
- 269. Gardens for everybody (1992). Tokyo: Japan Greenery Research and Development Center.
- 270. Gaston, M. L. (Ed.). (1994). *Tips on growing and marketing hanging baskets*. Columbus, OH: Ohio Florists' Association.
- 271. Gaston, M. L. (Ed.). (1996). *Tips on growing and marketing garden mums*. Columbus, OH: Ohio Florists' Association. Includes bibliographical references.
- 272. Gaston, M. L. (Ed.). (1997). Tips on growing specialty potted crops. Columbus, OH: Ohio Florists' Association. Includes bibliographical references.

- 273. George, V. (1995). Gardening, mental health and community care. Frome, Somerset: Horticultural Therapy.
- 274. Gerlach-Spriggs, N., Kaufman, R. E., & Warner, S. B. (1998). Restorative gardens: The healing landscape. New Haven, CT: Yale University Press.
- 275. Gerontological Society, & Syracuse University. Gerontology Center. (1975). Barrier-free design for the elderly and the disabled. Syracuse, NY: Syracuse University.
- 276. Gething, L. (1992). Person to person: A guide for professionals working with people with disabilities (2nd ed.). Baltimore, MD: P.H. Brookes Pub. Co. Includes bibliographical references (p. 259-268) and index.
- 277. Gething, L. (1997). Person to person: A guide for professionals working with people with disabilities (3rd ed.). Baltimore, MD: Paul H. Brookes. Includes bibliographical references (p. 349-366) and index.
- 278. Gething, L., Leonard, R., & O'Loughlin, K. (1986). *Person to person: Community awareness of disability*. Sydney, Australia: Williams & Wilkins. Includes index and bibliography (p. 171-172).
- 279. Gibson, M. (1982). *Gardening for handicapped and elderly persons*. Washington, DC: National Library Service for the Blind and Physically Handicapped, Library of Congress,.
- 280. Gilbert, H. (1985). *Horticultural therapy, 1979-85: 41 citations*. Beltsville, MD: U. S. Dept. of Agriculture, National Agricultural Library. Includes index.
- 281. Gillespie, L. E. (1980). *Influences of horticultural therapy intervention on self-concept of institutionalized moderately retarded adolescents*. Clemson, SC: South Caroline Agricultural Experiment Station. Horticulture Department. Includes bibliography (p. [71]-74).
- 282. Gilmer, M. (1997). Rooted in the spirit: Exploring inspirational gardens. Dallas, TX: Taylor Publishing Company.
- 283. Ginsburg, N. S., Koppel, B., & McGee, T. G. (Eds.). (1991). Extended metropolis: Settlement transition in Asia. Honolulu,HI: University of Hawaii.
- 284. Gilreath, P. R. (1976). *Hortitherapy program for the visually handicapped*. Clemson, SC: Clemson University, AES. Includes bibliographical references.

- 285. Girling, C. L., & Helphand, K. I. (1994). Yard, street, park: The design of suburban open space. New York, NY: J. Wiley. Includes bibliographical references (p. 229-239) and index.
- 286. Gleeson, B. (1999). *Geographies of disability*. London, England: New York, NY: Routledge.
- 287. Glendinning, C. (1994). "My name is Chellis & I'm in recovery from Western civilization". Boston, MA; New York, NY: Shambhala; Distributed in the United States by Random House. Includes bibliographical references (p. 215-232) and index.
- 288. Glenn, A. M. (1998). Entrepreneurial community greening: A model for the Delridge valley of Seattle. Seattle, WA: University of Washington. Includes bibliographical references (p.[98]-99).
- 289. Golby, R. (1993). Container gardener. London, England: Headline.
- 290. Goldsmith, S. (1963). Designing for the disabled: A manual of technical information. London, England: Technical Information Service, Royal Institute of British Architects.
- 291. Goldsmith, S. (1967). *Designing for the disabled* (2d, rev. and expanded ed.). New York, NY: McGraw-Hill.
- 292. Goldsmith, S. (1976). *Designing for the disabled* (3rd, fully rev. ed.). London, England: RIBA Publications.
- 293. Goldsmith, S. (1984). *Designing for the disabled* (3rd, fully rev. ed.; reprinted with updates, October 1984.). London, England: RIBA Pub. Includes index.
- 294. Goldsmith, S. (1997). *Designing for the disabled: The new paradigm*. Oxford, United Kingdom; Boston, MA: Architectural Press.
- 295. Goldstein, E. B. (1980). *Sensation and perception*. Belmont, CA: Wadsworth. Includes indexes and bibliography (p. 455-477).
- 296. Goldstein, E. B. (1980). *Sensation and perception* (2nd ed.).Belmont, CA: Wadsworth. Includes indexes and bibliography (p. 447-466).
- 297. Goldstein, E. B. (1989). *Sensation and perception* (3rd ed.). Belmont, CA: Wadsworth. Includes indexes and bibliography (p. 551-575).
- 298. Goldstein, E. B. (1996). *Sensation and perception* (4th ed.). Pacific Grove, CA: Brooks/Cole Pub. Includes bibliographical references (p. 622-652) and indexes.

Human Issues in Horticulture: A Bibliography

- 299. Goldstein, E. B. (1999). *Sensation and perception* (5th ed.). Pacific Grove, CA: Brooks/Cole Pub. Includes bibliographical references (p. 589-628) and indexes.
- 300. Golledge, R. G., & Stimson, R. J. (1987). *Analytical behavioural geography*. London, England: New York, NY: Croom Helm.
- 301. Goltsman, S. M., Gilbert, T. A., & Wohlford, S. D. (1992). Accessibility checklist: An evaluation system for buildings and outdoor settings. Berkeley, CA: MIG Communications.
- 302. Goltsman, S. M., Gilbert, T. A., & Wohlford, S. D. (1993). Accessibility checklist: An evaluation system for buildings and outdoor settings (2nd ed.). Berkeley, CA: MIG Communications.
- 303. Göncü, A. (Ed.). (1999). *Children's engagement in the world: Sociocultural perspectives.* Cambridge; New York, NY: Cambridge University Press.
- 304. Goodenough, U. (1998). Sacred depths of nature. New York, NY: Oxford University Press. Includes bibliographical references (p. [175]-182) and index.
- 305. Graber, L. H. (1976). *Wilderness as sacred space*. Washington, DC: Association of American Geographers.
- 306. Grant White, M. (1969). *Pots and pot gardens*. London, England: Abelard-Schuman.
- 307. Greenbie, B. B. (1988). *Space and spirit in modern Japan*. New Haven, CT: Yale University Press.
- 308. Greenblatt, F. S. (1988). *Therapeutic recreation for long-term care facilities*. New York, NY: Human Sciences Press. Includes index and bibliography (p. 187-192).
- 309. Greene, M. L. (1910). *Among school gardens*. New York, NY: Charities Publication Committee.
- 310. Greenman, J. T. (1988). Caring spaces, learning places: Children's environments that work. Redmond, WA: Exchange Press. Includes bibliography (p. 211-215).
- 311. Greenwood, C. R., Arreaga-Mayer, C., & Clark-Preston, D. (Eds.). (1985). *Eco-behavioral approach to research in special education*. Lawrence, KS: Juniper Gardens Children's Project, Bureau of Child Research: Depts. of Special Education and Human Development and Family Life, University of

- Kansas. Includes bibliographical references.
- 312. Gregory, R. L., & Colman, A. M. (Eds.). (1995). Sensation and perception. New York, NY: Longman Group. Includes bibliographical references and index.
- 313. Gremillion, K. J. (Ed.). (1997). *People, plants, and landscapes: Studies in paleoethnobotany*. Tuscaloosa, AL: University of Alabama Press.
- 314. Grene, M. G. (1968). *Approaches to a philosophical biology*. New York, NY: Basic Books. Includes bibliographical references.
- 315. Grene, M. G. (1974). Understanding of nature: Essays in the philosophy of biology. Dordrecht; Boston, MA: Reidel Pub. Co.
- 316. Griffiths, P. (Ed.). (1992). Trees of life: Essays in philosophy of biology. Dordrecht; Boston, MA: Kluwer Academic Publishers.
- 317. Groning, G., & Wolschke-Bulmahn, J. (1995). Von Ackermann bis Ziegelhutte: Ein jahrhundert kKleingartenkultur in Frankfurt am Main. Frankfurt am Main, Germany: Kramer.
- 318. Grossman, D., Berg, L. v. d., & Ajaegbu, H. I. (Eds.). (1999). *Urban and peri-urban agriculture in Africa: Proceeding of a workshop, Netanya, Israel, 23-27 June 1996*. Aldershot, United Kingdom: Ashgate.
- 319. Grubinger, V. P. (1999). Sustainable vegetable production from start-up to market. Ithaca, NY: Natural Resource, Agriculture, and Engineering Service, Cooperative Extension. Includes bibliographical references (p. 257-265).
- 320. Guide to designing accessible outdoor recreation facilities (1980). Ann Arbor, MI: Heritage Conservation and Recreation Service, U.S. Department of Interior, Lake Central Regional Office.
- 321. Guinness, B. (1996). *Creating a family garden*. New York, NY: Abbeville Press Publishers. Includes index.
- 322. Guinness, B. (1996). Family gardens. Newton Abbot, Devon: David & Charles. Includes index.
- 323. Haakenson, D. (1995). Small commercial garden: How to make \$10,000 a year in your backyard. Bismarck, ND: PC-Services.
- 324. Hackman, R. M. (1986). Senior gardening and nutrition project. Eugene, OR: College of Human Development and Performance, University of Oregon.

- 325. Hagedorn, R. (1987). *Therapeutic horticulture*. Bicester, Oxon, United Kingdom: Winslow Press. Includes bibliography (p. 156-161).
- 326. Hall, K. (1988). Growing and everlasting experience: [project, Diploma of Recreation and Sport, New Zealand]. Canterbury, New Zealand: Lincoln University.
- 327. Hall, W. (1983). Barnacle Parp's guide to garden & yard power tools: Selection, maintenance & repair. Emmaus, PA: Rodale Press. Includes index.
- 328. Hall, W. L. (1901). *Tree planting on rural school grounds*. Washington, DC: U.S. Dept. of Agriculture.
- 329. Halpin, A. M. (1989). Window box book. New York, NY: Simon and Schuster.
- 330. Ham, J. (1998). Designing playgrounds. Menlo Park, CA: Dale Seymour.
- 331. Hamill, C. M., Oliver, R. C., Benes, N., & Krieger, A. (1980). *Therapeutic activities for the handicapped elderly*. Rockville, MD: Aspen Systems Corp. Includes bibliographies and index.
- 332. Hamilton, B. (1987). Report on training needs in horticultural therapy in Victoria [microform]. Melbourne, Australia: Victorian TAFE Clearinghouse.
- 333. Hamlyn, D. W. (1961). Sensation and perception: A history of the philosophy of perception. London, England; New York, NY: Routledge & K.Paul; Humanities Press;.
- 334. Handelsman, J. F. (1996). *Growing myself: A spiritual journey through gardening*. New York, NY: Dutton.
- 335. Haraszty, E., & Colen, B. D. (1980). *Living with flowers*. New York, NY: Liveright. Includes index.
- 336. Harkness, S. P., & Groom, J. N. (1976). Building without barriers for the disabled. New York, NY: Whitney Library of Design.
- 337. Harper, P., Madsen, C., & Light, J. (1994). *Natural garden book: A holistic approach to gardening*. New York, NY: Simon & Schuster.
- 338. Harris, M. (1996). *Healing garden*. Toronto, Canada: Harper Collins Publishers.
- 339. Hart, C. H. (Ed.). (1993). *Children on playgrounds: Research perspectives and applications*. Albany, NY: State University of New York Press.

- 340. Hartshorne, C. (1934). *Philosophy and psychology of sensation*. Chicago, IL: University of Chicago Press. Includes index and bibliography: p. 280-282
- 341. Harvey, J. H. (Ed.). (1981). *Cognition, social behavior, and the environment.* Hillsdale, NJ: L. Erlbaum.
- 342. Hassler, D., & Gregor, L. (Eds.). (1998). *Place to grow: Voices and images of urban gardeners*. Cleveland, OH: Pilgrim Press. Includes bibliographical references (p. 104).
- 343. Hawkins, B. A., May, M. E., & Rogers, N. B. (1996). *Therapeutic activity intervention with the elderly: Foundations & practices*. State College, PA: Venture Pub. Includes bibliographical references.
- 344. Hayden, D. (1995). *Power of place: Urban landscapes as public history*. Cambridge, MA: MIT Press.
- 345. Heimstra, N. W., & McFarling, L. H. (1974). *Environmental psychology*. Monterey, CA: Brooks/Cole Pub. Co.
- 346. Heimstra, N. W., & McFarling, L. H. (1978). *Environmental psychology* (2d ed.). Monterey, CA: Brooks/Cole Pub. Co
- 347. Heitz, H. (1992). Container plants: For patios, balconies, and window boxes (K. Luft, Trans.). Hauppauge, NY: Barron's.
- 348. Hemenway, H. D. (1909). How to make school gardens: A manual for teachers and pupils. New York, NY: Doubleday, Page & Company. "School garden bibliography": p. [97]-107
- 349. Hemenway, H. D. (1912). How to make school gardens: A manual for teachers and pupils. New York, NY: Doubleday, Page & Company. "School garden bibliography": p. [97]-107.
- 350. Henderson, E., & Van En, R. (1999). Sharing the harvest: A guide to community supported agriculture. White River Junction, VT: Chelsea Green.
- 351. Henty, C. (1989). For the people's pleasure: Australia's botanic gardens. New York, NY: Rizzoli.
- 352. Herd, M. (1995). *City gardener: Beautiful gardens for small spaces.* Pymble, Sydney, New South Wales, Australia: Angus & Robertson. Includes index.
- 353. Heseltine, P., & Holborn, J. (1987). *Playgrounds: The planning, design and construction of play environments.* New York, NY: Nichols Pub. Co.
- 354. Hessayon, D. G. (1996). Container expert. London, England: New York,

- NY;: Expert Books; Distributed in the United States by Sterling Pub. Co.
- 355. Hewson, M. L. (1994). Horticulture as therapy: A practical guide to using horticulture as a therapeutic tool. Guelph, Ontario, Canada: M.L. Hewson.
- 356. Hieke, K. (1976). *Window-box, balcony and patio gardening* (O. Kuthanova, Trans.). London, England; New York, NY: Hamlyn. Includes indexes.
- 357. Hillier, M. (1991). *Book of container gardening*. New York, NY: Simon & Schuster. Includes index.
- 358. Hillier, M. (1995). *Container gardening through the year* (1st American ed.). London, England; New York, NY; Boston, MA: Dorling Kindersley.
- 359. Hillier, M. (1998). *Container gardening through the year* (1st pbk. ed.). New York, NY: Dorling Kindersley. Includes index.
- 360. Hillman, P. (1985). *How does your garden grow?*. London, England; Dover, NH; Portland, OR: Croom Helm. Includes index.
- 361. Hiott, J. A. (1979). *Hortitherapy program for the mentally handicapped*. Clemson, SC: South Carolina Agricultural Experiment Station.
- 362. Hiss, T. (1990). Experience of place. New York, NY: Knopf.
- 363. Hobhouse, P. (1992). Penelope Hobhouse's gardening through the ages: An illustrated history of plants and their influence on garden styles--from ancient Egypt to the present day. New York, NY: Simon & Schuster.
- 364. Hogg, J., & Cavet, J. (Eds.). (1995). Making leisure provision for people with profound learning and multiple disabilities, (1st ed.). London, England; New York, NY: Chapman & Hall.
- 365. Holahan, C. J. (1978). *Environment and behavior: A dynamic perspective*. New York, NY: Plenum Press.
- 366. Holahan, C. J. (1982). *Environmental psychology*. New York, NY: Random House.
- 367. Hollingsworth, B. (1968). *Gardening on Main Street*. New Brunswick, NJ: Rutgers University Press.
- 368. Hollinrake, D. (1992). Gardening: Equipment for disabled people (Wassenaar, I. P., & Cochrane, G. M., Eds.), (2nd ed.). Oxford, England: Disability Information Trust.
- 369. Holmes-Siedle, J. (1996). Barrier-free design: A manual for building designers

- and managers. Oxford, England: Butterworth Architecture. Includes index.
- 370. Hood, G. W. (1916). *Practical school and home gardens*. Lincoln, NE: Long and Company. "References" at end of most of the chapters.
- 371. Horticultural therapy. (1952). East Lansing, MI: Michigan State College. Syllabus and notes from horticultural therapy special course, August 4 August 8, 1952 at Michigan State College, East Lansing, Michigan.
- 372. Horticultural Therapy Department Staff of the Chicago Horticultural Society, Daubert, J. R., & Rothert, E. A., Jr. (1981a). *Horticultural therapy at a psychiatric hospital*. Glencoe, IL: Chicago Horticultural Society.
- 373. Horticultural Therapy Department Staff of the Chicago Horticultural Society, Daubert, J. R., & Rothert, E. A., Jr. (1981b). *Horticultural therapy for the mentally handicapped*. Glencoe, IL: Chicago Horticultural Society.
- 374. Horticultural Therapy Department Staff of the Chicago Horticultural Society, Rothert, E. A., Jr., & Daubert, J. R. (1981a). *Horticultural therapy at a physical rehabilitation facility*. Glencoe, IL: Chicago Horticultural Society.
- 375. Horticultural Therapy Department Staff of the Chicago Horticultural Society, Rothert, E. A., Jr., & Daubert, J. R. (1981b). *Horticultural therapy for nursing homes, senior centers, retirement living*. Glencoe, IL: Chicago Horticultural Society.
- 376. Horticultural Therapy Professional Development Symposium (Oct. 26-28, 1990: Topeka, Kans.). (1990). Proceedings of the Horticultural Therapy Professional Development Symposium, "Competencies Defined" Mattson R. H., (Ed.). Manhattan, KS: Kansas State University.
- 377. Horticultural Therapy Society of N.S.W. (1990). *Ideas for less active gardeners*. Ryde, New South Wales, Australia: Horticultural Therapy Society of N.S.W.
- 378. Horticultural therapy second workshop, Michigan State College, June 29-30, July 1-3, 1953. (1953). East Lansing, MI: Michigan State College.
- 379. Horton, A. (Ed.). (1984). Gardening in containers. San Ramon, CA: Ortho Books.
- 380. Hottes, A. C. (Ed.). (1935). Better homes and gardens. My Better Homes & Gardens's garden helper. Des Moines, IA: Meredith Pub. Co.
- 381. Hou, J., Johnson, J., & Lawson, L. J. Greening cities, growing communities: Learning from Seattle's urban community gardens. Washington, DC: Landscape

- Architecture Foundation in association with University of Washington Press. Includes bibliographical references and index
- 382. Howard, E. (1898). *To-morrow: A peaceful path to real reform.* London, England: Swann Sonnenschein.
- 383. Howard, G. S. (1997). *Ecological psychology: Creating a more earth-friendly human nature.* Notre Dame, IN: University of Notre Dame Press. Includes bibliographical references (p. 157-161)
- 384. Howard, J. (1992). *Bach flower remedies for women.* Saffron Walden, Essex, England: C.W. Daniel.
- 385. Hudak, J. l., Mallory, D., & United States. Dept. of Health and Human Services, Administration on Developmental Disabilities. (1980). *Melmood manual: A planning and operations manual for horticultural training and work co-op programs*. Menomonie, WI: Materials Development Center, Stout Vocational Rehabilitation Institute, University of Wisconsin-Stout. Inclufes bibliography (p. 179-181).
- 386. Hudson, H., & Fish, M. (1944). Occupational therapy in the treatment of the tuberculous patient. New York, NY: National Tuberculosis Association.
- 387. Hull, D. L., & Ruse, M. (Eds.). (1998). *Philosophy of biology*. New York, NY: Oxford University Press. Includes bibliographical references (p.[760]-768) and index.
- 388. Hunter, C. (1994). *Everyone's nature*. Helena, MT: Falcon. Includes bibliographical references (p. 174-176) and index.
- 389. Hynes, H. P. (1996). *Patch of Eden: America's inner city gardeners*. White River Junction, VT: Chelsea Green Pub.
- 390. *Ideas for less active gardeners* (1990). Ryde, New South Wales, Australia: Horticultural Therapy Society of N.S.W.
- 391. Improving farmstead accessibility (1994). West Lafayette, IN: Breaking New Ground Resource Center, Purdue University.
- 392. International Conference on Play and Play Environments (1983: University of Texas at Austin). (1985). When children play: Proceedings of the International Conference on Play and Play Environments (J. L. Frost, & S. Sunderlin, Eds.). Wheaton, MD: Association for Childhood Education International. Includes bibliographies.
- 393. International Horticultural Congress (24th: 1994: Kyoto, Japan).

- (1995). Horticulture in human life, culture, and environment: International Symposium, 22 August 1994: XXIV th International Horticultural Congress, Kyoto International Conference Hall, 21-27 August 1994, Kyoto, Japan (E. Matsuo, & D. Relf, Eds.). Leuven, Belgium: International Society for Horticultural Science. Includes bibliographical references and index.
- 394. International Seminar on Vocational Rehabilitation for Mentally Retarded Persons (2d : 1976 : Philadelphia, etc.). (1978). Proceedings of the Second International Seminar on Vocational Rehabilitation for Mentally Retarded Persons (E. K. Hollander, & G. Soloyanis, Eds.). Washington, DC: American Association on Mental Deficiency. Includes bibliographies.
- 395. International Symposium on the Use of Space by Animals and Men (1968: Dallas). (1971). Behavior and environment; the use of space by animals and men. proceedings of an international symposium held at the 1968 meeting of the American Association for the Advancement of Science A. H. Esser, (Ed.). New York, NY: Plenum Press.
- 396. Jansson, B. (1984). *Children's play and nature in an urban environment*. Frankfurt am Main; New York, NY: P. Lang. Includes bibliography (p. 204-217).
- 397. Jarvis, C. D. (1916). *Gardening in elementary city schools*. Washington, DC: Govt. print. off.
- 398. Jay, R. (1998). Gardens of the spirit: Create your own sacred spaces. New York, NY: Sterling Pub. Includes bibliographical references (p. 140) and index.
- 399. Jay, R. (1998). Sacred gardens: Creating a space for contemplation and meditation. London, England: Thorsons. Includes bibliographical references (p. 140) and index.
- 400. Jeavons, J., & Cox, C. (1999). Sustainable vegetable garden: A backyard guide to healthy soil and higher yields. Berkeley, CA: Ten Speed Press. Includes bibliographical references and index.
- 401. Jekyll, G. (1982). *Children and gardens*. (reprint ed.). Woodbridge, Suffolk, U.K.: Antique Collectors' Club.
- 402. Jensen, J. V. (1949). *Garden colonies in Denmark*. Copenhagen: Det Danske Selskab.
- 403. Jensen, M. H., & Malter, A. J. (1995). *Protected agriculture: A global review.* Washington, DC: World Bank. Includes bibliographical references

- 404. Jewell, J. R. (1907). Agricultural education, including nature study and school gardens. Washington, DC: Govt. Print. Off. Includes bibliographical references and index.
- 405. Jewell, J. R. (1908). Agricultural education, including nature study and school gardens (2d, rev ed.). Washington, DC: Govt. print. off. Includes bibliography (p. 128-133).
- 406. Jobb, J. (1979). Complete book of community gardening. New York, NY: Morrow.
- 407. Johns, L. (1974). *Plants in tubs, pots, boxes, and baskets*. New York, NY: Van Nostrand Reinhold Co.
- 408. Johnson, M. (1978). *Tub farming: Grow vegetables anywhere in containers*. Charlotte, VT: Garden Way Pub.
- 409. Jorgensen, J. (1975). Landscape design for the disabled. McLean, VA: distributed by American Society of Landscape Architects Foundation.
- 410. Joseph, L. E. (1990). *Gaia the growth of an idea*. New York, NY: St. Martin's Press. Includes index.
- 411. Joyce, D. (1991). Hanging baskets, window boxes, and other container gardens: A guide to creative small-scale gardening. New York, NY: Summit Books.
- 412. Joyce, D. (1993). Windowbox gardening (American ed.). Old Saybrook, CT: Globe Pequot Press.
- 413. Joyce, D. (1996). *Complete container garden*. Pleasantville, NY: Reader's Digest Association.
- 414. Kahn, P. H. (1999). Human relationship with nature: Development and culture. Cambridge, MA: MIT Press.
- 415. Kane, D., Brunton, L., & Fournier, E. (1999). Sanctuary: Gardening for the soul. New York, NY: Friedman/Fairfax Publishers. Includes bibliographical references (p. 142).
- 416. Kaplan, R., & Kaplan, S. (1989). Experience of nature: A psychological perspective. Cambridge, England; New York, NY: Cambridge University Press.
- 417. Kaplan, R., & Kaplan, S. (1995). Experience of nature: A psychological perspective. Ann Arbor, MI: Ulrich's Bookstore.
- 418. Kaplan, R., & Kaplan, S. (Eds.). (1978). Humanscape: Environments for

- people. N. Scituate, MA: Duxbury Press.
- 419. Kaplan, R., Kaplan, S., & Ryan, R. L. (1998). With people in mind: Design and management of everyday nature. Washington, DC: Island Press.
- 420. Kaplan, S., & Kaplan, R. (1982). Cognition and environment: Functioning in an uncertain world. New York, NY: Praeger.
- 421. Kaufman, P. B., & LaCroix, J. D. (Eds.). (1979). *Plants, people, and environment*. New York, NY: Macmillan. Includes bibliographies and index.
- 422. Keane, J. (1992). *Plants for people: The psychological and physiological effects of plants.* Beltsville, MD: National Agricultural Library.
- 423. Keeling, J. (1995). New terracotta gardener: Creative ideas from leading gardeners (Rev. ed.). North Pomfret, VT: Trafalgar Square Pub.
- 424. Kellert S. R., & Wilson E. O. (Eds.). (1993). *Biophilia hypothesis*. Washington, DC: Island Press.
- 425. Kelsen, H. (1943). *Society and nature; a sociological inquiry*. Chicago, IL: The University of Chicago Press. Includes bibliographical "notes" (p. 267-384).
- 426. Kelsen, H. (1946). *Society and nature, a sociological inquiry*. London, England: K. Paul, Trench, Trubner & Co., ltd. Includes "Notes" (bibliographical) (p. 267-384).
- 427. Kelsen, H. (1974). Society and nature (Reprint ed.). New York, NY: Arno Press.
- 428. Kevin Heinze Garden Centre. (1989). Kevin Heinze Garden Centre Inc. 10th anniversary 1979-1989. Doncaster, Victoria, Australia: Kevin Heinze Garden Centre..
- 429. Kevin Heinze Garden Centre. (1999). *A dream bears fruit.* Doncaster, Victoria, Australia: Kevin Heinze Garden Centre.
- 430. Kiefer, J., & Kemple, M. (1998). Digging deeper: Integrating youth gardens into schools & communities; a comprehensive guide. Montpelier, VT: Food Works; Common Roots Press.
- 431. Kirby, A. (1997). *Gardening in small spaces*. London, England: Pavilion. Includes index and bibliography (p. 158).
- 432. Kirschbaum, J. B. (1999). Designing sidewalks and trails for access [microform]. Washington, DC: U.S. Dept. of Transportation, Federal

- Highway Administration. Includes bibliographical references and index.
- 433. Klein, R. M. (1979). *Green world: An introduction to plants and people.* New York, NY: Harper & Row. Includes bibliographies and index.
- 434. Klein, R. M. (1987). *Green world: An introduction to plants and people* (2nd ed.). New York, NY: Harper & Row. Includes bibliographies and index
- 435. Koch, M. (1998). Seed leaf flower fruit. New York, NY: Smithmark Publishers.
- 436. Kramer, J. (1970). Complete book of patio gardening. New York, NY: Putnam.
- 437. Kramer, J. (1971). Container gardening indoors and out. Garden City, NY: Doubleday.
- 438. Kramer, J. (1971). Hanging gardens; basket plants, indoors and out. New York, NY: Scribner. Includes bibliography (p. [111]).
- 439. Kramer, J. (1973). Gardening without stress and strain. New York, NY: Charles Scribner's Sons.
- 440. Kramer, J. (1979). *Gardening in small spaces*. Tuscon, AZ: H.P. Books. Includes index.
- 441. Kramer, J. (1982). *Hanging gardens: Basket plant, indoors and out.* Piscataway, NJ: New Century publishers.
- 442. Kramer, J. (1991). Easy gardening: No stress--no strain (Updated and expanded ed.). Golden, CO: Fulcrum Publishing.
- 443. Kramer, J. (1997). Earthly delights: Tubs of tomatoes and buckets of beans. Golden, CO: Fulcrum Pub.
- 444. Kreig, M. (1964). *Green medicine; the search for plants that heal.* Chicago, IL: Rand McNally. Includes bibliography (p. 433-446).
- 445. Kriegsman, K. H., Zaslow, E. L., & D'Zmura-Rechsteiner, J. (1992). *Taking charge: Teenagers talk about life & physical disabilities.* Rockville, MD: Woodbine House.
- 446. Kruse, L., & Arlt, R. (1984). Environment and behavior: An international and multidisciplinary bibliography, 1970-1981. Paris, France; New York, NY: K.G. Saur. Includes index.
- 447. Kruse, L., & Schwarz, V. (1988). Environment and behavior, part II: An

- international and multidisciplinary bibliography, 1982-1987. Paris, France; New York, NY: K.G. Saur. Includes index.
- 448. Lalli, V. A., Tennessen, D. J., & Lockhart, K. (1998). *Using plants to bridge the generations: Horticulture and Intergenerational Learning as Therapy (HILT)*. Ithaca, NY: Cornell Cooperative Extension.
- 449. Landman, R. H. (1993). Creating community in the city: Cooperatives and community gardens in Washington, D.C. Westport, CT: Bergin & Garvey.
- 450. Landreth, B. (1893). Market gardening and farm notes. Experiences and observations in the garden and field, of interest to the amateur gardener, trucker and farmer. New York, NY: Orange Judd Company.
- 451. Landreth, B. (1897). Market gardening and farm notes: Experiences and observations in the garden and field, of interest to the amateur gardener, trucker and farmer. New York, NY: O. Judd. Includes index.
- 452. Langer, R. W. (1969). *After-dinner gardening book*. New York, NY: Macmillan.
- 453. Langer, R. W. (1971). *After-dinner gardening book*. New York, NY: Collier Books. Includes index and bibliography (p. 193-195).
- 454. Langer, R. W. (1975). Grow it indoors; a practical, personal how-to and why guide to growing successful houseplants. New York, NY: Saturday Review Press.
- 455. Latham, C. A. T. (1983). Occupational therapy for physical dysfunction, (2nd ed.). C. A. T. Latham, (Ed.). Baltimore, MD: Williams & Wilkins. Includes bibliographical references and index.
- 456. Latham, C. A. T. (1989). Occupational therapy for physical dysfunction (3rd ed.). C. A. T. Latham, (Ed.). Baltimore, MD: Williams & Wilkins. Includes bibliographies and index
- 457. Latham, C. A. T. (1995). Occupational therapy for physical dysfunction (5th ed.). C. A. T. Latham, (Ed.). Baltimore, MD: Williams & Wilkins. Includes bibliographies and index
- 458. Lauder, R. (1969). Goal is: Mobility! Environmental and transportation barriers encountered by the disabled. Washington, DC: Published for the National Citizens Conference on Rehabilitation of the Disabled and Disadvantaged, by U.S. Social rehabilitation Service. Includes bibliography (p. 65-72).
- 459. Lawson, J. E. (1992). Community-supported agriculture: Farming that works.

- Santa Cruz, CA: Includes bibliographical references.
- 460. Lawton, M. P. (1980). *Environment and aging*. Monterey, CA: Brooks/Cole Pub. Co. Includes indexes and bibliography (p. 167-177).
- 461. Lawton, M. P. (1986). *Environment and aging* (2nd ed.). Albany, NY: Center for the Study of Aging. Includes indexes and bibliography (p. 185-196).
- 462. Lebenswerte Umwelt für Senioren und Behinderte = A quality environment for the elderly and handicapped (1989). Stuttgart: K. Kramer.
- 463. Ledermann, A. (1960). *Creative playgrounds and recreation centers*. New York, NY: F.A. Praeger.
- 464. Ledermann, A., & Trachsel, A. (1959). *Creative playgrounds and recreation centers* (E. Priefert, Trans.). New York, NY: F. A. Praeger.
- 465. Ledermann, A., & Trachsel, A. (1960). *Playgrounds and recreation spaces* (E. Priefert, Trans.). London, England: Architectural Press.
- 466. Ledermann, A., & Trachsel, A. (1968). *Creative playgrounds and recreation centers* (J. C. Palmes, E. Priefert, Trans.). (Rev. ed.). New York, NY: Praeger.
- 467. Ledermann, A., & Trachsel, A. (1968). *Playgrounds and recreation spaces* (J. C. Palmes, E. Priefert, Trans.). (2nd ed.). London, England: Architectural Press.
- 468. Lee, A. W. (1993). Backyard market gardening: The entrepreneur's guide to selling what you grow G. DeVault, (Ed.), . Burlington, VT: Good Earth Publications.
- 469. Lees, C. B. (1970). *Gardens, plants, and man.* Englewood Cliffs, NJ: Prentice-Hall.
- 470. Levine, M. W., & Shefner, J. M. (1981). Fundamentals of sensation and perception. Reading, MA: Addison-Wesley Pub. Co.
- 471. Levine, M. W., & Shefner, J. M. (1991). Fundamentals of sensation and perception (2nd ed.). Pacific Grove, CA: Brooks/Cole.
- 472. Levitan, L. (1980). *Improve your gardening with backyard research*. Emmaus, PA: Rodale.
- 473. Lewington, A. (1990). *Plants for people*. New York, NY: Oxford University Press.

- 474. Lewis, C. A. (1996). Green nature/human nature: The meaning of plants in our lives. Urbana, IL: University of Illinois Press..
- 475. Lidoff, L. (1983). Promoting self-help ventures in food production and distribution. Washington, DC: National Council on the Aging. Includes bibliography (p. 30-33).
- 476. Lifelong leisure skills and lifestyles for persons with developmental disabilities (1995). Baltimore, MD: Paul H. Brookes. Includes bibliographical references (p. 231-240) and index.
- 477. Lindayati, R. (1996). *Urban agriculture: A survey of academic expertise and programs in Canada*. Ottawa, Canada: International Development Research Centre.
- 478. Lindholm, G. (1995). *Skolgården: Vuxnas bilder, barnets miljö*. Alnarp, Sweden: Sveriges lantbruksuniversitet. Includes bibliographical references.
- 479. Link, C. B. (1974). *Plants for pots* (U. Robertshaw, Picknett, L., Link, C. B., & Stark, F. C., Eds.), (American ed.). Secaucus, NJ: Castle Books.
- 480. Llewellyn, R., & Davies, A. (1993). *Grow it yourself: Gardening with a physical disability*. London, England: Cedar. Includes bibliographical references (p. 91) and index.
- 481. Lopez, R. K. (1990). Gardens for growing people: A guide to gardening with children. Point Reyes, CA: Gardens for Growing People.
- 482. Lord, D. (1998). Little big book of Bach flower remedies. Hod Hasharon, Israel: Astrolog.
- 483. Lovelock, J. E. (1979). *Gaia, a new look at life on earth.* Oxford, England; New York, NY: Oxford University Press. Includes bibliography (p. 157).
- 484. Lovelock, J. E. (1987). *Gaia: A new look at life on earth.* Oxford, Oxfordshire; New York, NY: Oxford University Press. Includes bibliography (p. [155]-157).
- 485. Lovelock, J. (1982). *Gaia, a new look at life on earth*(1st paperback ed.). Oxford, Oxfordshire; New York, NY;: Oxford University Press.4 Includes bibliography (p. [155]-157).
- 486. Lovelock, J. (1995). *Gaia: A new look at life on earth.* Oxford, England; New York, NY: Oxford University Press.
- 487. Low-maintenance gardening (1998). Menlo Park, CA: Sunset Books.

- 488. MacCaskey, M., & Sinnes, A. C. (1982). *Gardening in small spaces*. Mount Vernon, VA: American Horticultural Society. Includes index.
- 489. MacLean, J. T. (1985). *Urban horticulture*. Beltsville, MD: U.S. Dept. of Agriculture, National Agricultural Library.
- 490. Mahner, M., & Bunge, M. A. (1997). Foundations of biophilosophy. Berlin, Germany; New York, NY: Springer.
- 491. Malassigne, P. (1977?). Design for all people: A compendium of selected publications. Blacksburg, VA: College of Architecture and Urban Studies, Virginia Polytechnic Institute and State University. Includes bibliographies.
- 492. Mandelbaum, M. (1964). *Philosophy, science, and sense perception: Historical and critical studies.* Baltimore, MD: Johns Hopkins Press.
- 493. Marinelli, J. (1998). *Stalking the wild amaranth: Gardening in an age of extinction*. New York, NY: Henry Holt. Includes bibliographical references (p. 215-228) and index.
- 494. Marken, B., & National Gardening Association. (1998). *Container gardening for dummies*. Foster City, CA: IDG Books Worldwide. Includes bibliographical references and index.
- 495. Market gardening (1977). Rome, Italy: Food and Agriculture Organization of the United Nations.
- 496. Massey, W. F. (Wilbur Fisk). (1900-1909). *Truck farming* (Improved and revised ed.). Southern Pines, NC: Supervising Committee of the Experimental Farms of the North Carolina State Horticultural Society.
- 497. Massey, W. F. (Wilbur Fisk). (1900-1939). *Truck farming* (Reprint ed.) New York, NY: German Kali Works.
- 498. Massingham, B. (1972). *Gardening for the handicapped*. Aylesbury, Buckinghamshire, UK: Shire Publications. Includes bibliography (p. 57).
- 499. Masterton, R. B. (Ed.). (1978). Sensory integration. New York, NY: Plenum Press.
- 500. Matthen M., & Ware R. X. (Eds.). (1994). Biology & society: Reflections on methodology. Calgary, Alta., Canada: University of Calgary Press. Includes bibliographical references and index
- 501. Mattson, R. H., & Romig, C. A. (Eds.). (1982). Horticultural therapy: An annotated bibliography. Manhattan, KS: Kansas State University.

- 502. May, L. (1995). In my father's garden. Atlanta, GA: Longstreet Press.
- 503. Mayr, E. (1988). Toward a new philosophy of biology: Observations of an evolutionist. Cambridge, MA: Belknap Press of Harvard University Press.
- 504. Mbiba, B. (1995). *Urban agriculture in Zimbabwe: Implications for urban management and poverty*. Aldershot, Hampshire, UK; Brookfield, VT: Avebury. Includes bibliography (p. 213-220).
- 505. McDonald, E. (1976). *Plants as therapy*. New York, NY: Praeger Publishers.
- 506. McDowell, C. F., & Clark-McDowell, T. (1998). Sanctuary garden: Creating a place of refuge in your yard or garden. New York, NY: Simon & Schuster.
- 507. McHarg, I. (1969). *Design with nature*. Washington, DC: Published for the American Museum of Natural History [by] the Natural History Press. Includes bibliographical footnotes.
- 508. McHarg, I. (1992). Design with nature. New York, NY: J. Wiley.
- 509. McHoy, P. (1986). *Container gardening*. Poole, Dorset, UK; New York, NY: Blandford Press. Includes index.
- 510. McHoy, P. (1990). Weekend gardener: All you need to plan and make a good-looking low-maintenance and fast-maturing garden. New York, NY: Gallery Books. Includes index.
- 511. McHoy, P. (1993). *Containers and baskets for year round colour.* London, England: Ward Lock. Includes index.
- 512. McIntyre, A. (1996). Flower power: Flower remedies for healing body and soul through herbalism, homeopathy, aromatherapy, and flower essences (1st American ed.). New York, NY: Henry Holt and Co. Includes bibliographical references (p. 282) and index.
- 513. Mehrabian, A. (1976). Public places and private spaces: The psychology of work, play, and living environments. New York, NY: Basic Books.
- 514. Mehrabian, A., & Russell, J. A. (1974). *Approach to environmental psychology*. Cambridge, MA: M.I.T. Press.
- 515. Meier, W. H. D. (1913). School and home gardens. Boston, MA; New York, NY: Ginn and Company.
- 516. Messervy, J. M. (1990). Contemplative gardens. Charlottesville, VA:

- Howell Press. Includes bibliographical references (p. 206).
- 517. Messervy, J. M. (1995). *Inward garden: Creating a place of beauty and meaning*. Boston, MA: Little, Brown. Includes bibliographical references and index.
- 518. Michigan State College. Horticulture Dept. (1952). Horticultural therapy. East Lansing, MI: Syllabus and notes from horticultural therapy special course, August 4 August 8, 1952 at Michigan State College, East Lansing, Michigan.
- 519. Miller, K. (1989). *Outside play and learning book: Activities for your children.* Mt. Rainier, MD: Gryphon House. Includes bibliographical references.
- 520. Miller, L. K. (1904). *Children's gardens for school and home: A manual of cooperative gardening.* New York, NY: Appleton. Includes index.
- 521. Miller, L. K. (1914). *Children's gardens for school and home: A manual of cooperative gardening*. New York, NY: D. Appleton and company.
- 522. Miller, M. (1983). *Gardening in small spaces*. New York, NY: Putnam. Includes index.
- 523. Miller, P. L. (1972). *Creative outdoor play areas*. Englewood Cliffs, NJ: Prentice-Hall.
- 524. Milner, M. (1977). Planning for accessibility: A guide to developing and implementing campus transition plans. Washington, DC: Association of Physical Plant Administrators of Universities and Colleges. Includes bibliography (p. 84).
- 525. Minter, S. (1993). Healing garden: A natural haven for body, senses and spirit. Boston, MA: C.E. Tuttle Co. Includes bibliographical references (p. 156) and index.
- 526. Minter, S. (1995). Healing garden: A natural haven for body, senses, and spirit (1st pbk. ed.). Boston, MA: Charles E. Tuttle.
- 527. Mittleider, J. R. (1981). Gardening by the foot: Mini grow-boxes for maxi yields. Bountiful, UT: Horizon Publishers & Distributors.
- 528. Mittleider, J. R. (1986). *Grow-bed gardening*. Santa Barbara, CA: Woodbridge Press.
- 529. Mobley, J., & Harris, N. W. (1991). City within a park: One hundred years of parks and boulevards in Kansas City, Missouri. Kansas City, MO: American Society of Landscape Architects and the Kansas City, Missouri Board of

- Parks & Recreation Commissioners.
- 530. Molen, S. L., Chambers, N., Wichrowski, M., & Fried, G. (1999). *Growth through nature: A preschool program for children with disabilities* H. Loomis, (Ed.). Sagaponack, NY; Portland, OR: Sagapress. Includes bibliographical references (p. 112-113).
- 531. Mollison, B. C. (1979). Permaculture two: Practical design for town and country in permanent agriculture. Stanley, Australia: Tagari.
- 532. Mollison, B. C. (1988). *Permaculture: A designers' manual.* Tyalgum, Australia: Tagari.
- 533. Mollison, B. C., & Holmgren, D. (1978). Permaculture 1: A perennial agricultural system for human settlements. Hobart, Australia: Environmental Psychology, University of Tasmania.
- 534. Mollison, B. C., & Holmgren, D. (1987). Perma-culture one: A perennial agriculture for human settlements. Tyalgum, Australia: Tagari.
- 535. Mooney, P. F., & Errett, M. C. (1991). Introduction to horticultural therapy: A horticultural therapy program for seniors in intermediate care facilities. Vancouver, British Columbia, Canada: Botanical Garden, Faculty of Agricultural Sciences, University of British Columbia. Includes bibliographical references.
- 536. Moore, B. (1987). Growing with gardens: A horticultural therapy training manual. Chapel Hill, NC: North Carolina Botanical Garden, University of North Carolina. Includes bibliography (p. 258-269).
- 537. Moore, B. (1989). Growing with gardening: A twelve-month guide for therapy, recreation, and education (Rev. ed.). Chapel Hill, NC: University of North Carolina Press.
- 538. Moore, R. C. (1986). *Childhood's domain: Play and place in child development.* London, England; Dover, NH: Croom Helm.
- 539. Moore, R. C. (1993). Plants for play: A plant selection guide for children's outdoor environments. Berkeley, CA: MIG Communications.
- 540. Moore, R. C., Goltsman, S. M., Iacofano, D. S., Beckwith, J., & Schneekloth, L. (Eds.). (1987). *Play for all guidelines: Planning, design and management of outdoor play settings for all children*, (Version 2.0 ed.). Berkeley, CA: MIG Communications.
- 541. Moore, R. C., Goltsman, S. M., & Iacofano, D. S. (Eds.) (1992). Play

- for all guidelines: Planning, design, and management of outdoor play settings for all children, (2nd ed.). Berkeley, CA: MIG Communications.
- 542. Moran, D. M. (1990). *Allotment movement in Britain*. New York, NY: P. Lang. Includes bibliographical references (p. 173-177) and index.
- 543. Morgan, B. (1989). Growing together: Activities to use in your horticulture and horticultural therapy programs for children. Pittsburgh, PA: Pittsburgh Child Guidance Foundation. Includes bibliography (p. 297-305).
- 544. Mullett, B. H. (1985). *Horticulture therapy: An application of experiential education*. Mankato, MN: Mankato State University. Includes bibliography (leaves 67-72).
- 545. Munson, R. (1971). Man and nature: Philosophical issues in biology. New York, NY: Dell Pub. Co.
- 546. Murphy, W. B. (1990). Beds and borders: Traditional and original garden designs. Boston, MA: Houghton Mifflin.
- 547. Murray, E. (1997). *Cultivating sacred space: Gardening for the soul.* San Francisco, CA: Pomegranite.
- 548. Naimark, S. (Ed.). (1982). *Handbook of community gardening*. New York, NY: Scribner.
- 549. Nash, H., & Speichert, C. G. (1996). Water gardening in containers: Small ponds, indoors & out. New York, NY: Sterling Pub. Co. Includes index.
- 550. National community gardening survey (1998). Philadelphia, PA: American Community Gardening Association.
- 551. National Council for Therapy and Rehabilitation Through Horticulture. (1979). *Comprehensive view of horticulture and the blind.* Mt. Vernon, VA: National Council for Therapy and Rehabilitation Through Horticulture. Issued as *HortTherapy*, Vol. 1, no. 1 (March 1979). Includes bibliographical references.
- 552. National Council for Therapy and Rehabilitation Through Horticulture. (1981). *Comprehensive view of horticulture and the aging.* Gaithersburg, MD: American Horticultural Therapy Association. Includes bibliographical references.
- 553. National Council for Therapy and Rehabilitation Through Horticulture. (1986). *Project plantwork: A horticulture employment initiative for workers with developmental disabilities.* Gaithersburg, MD: National Council for

- Therapy and Rehabilitation Through Horticulture. Includes bibliography (p. 77-89).
- 554. National Council for Therapy and Rehabilitation Through Horticulture (6th: 1978: Topeka, Kansas). (1978). *Directions '78: 6th annual conference, September 17-20 Topeka, Kansas*. Topeka, KS: National Council for Therapy and Rehabilitation Through Horticulture.
- 555. National Council for Therapy and Rehabilitation Through Horticulture (7th: 1979: Glencoe, Illinois). (1979). Growth and professionalism: 7th annual conference [of] National Council for Therapy and Rehabilitation Through Horticulture [held at] Chicago Horticultural Society Botanic Garden, Glencoe, Illinois, October 9-12, 1979 M. E. Mandeville, (Ed.), National Council for Therapy and Rehabilitation Through Horticulture. Includes bibliographical references.
- 556. National Council for Therapy and Rehabilitation through Horticulture. Horticultural Therapy Research Project, Relf, P. D., & Snoddy, G. D. (1978). *Study of horticultural therapy in Fairfax County, Virginia*. Mount Vernon, VA: The Council.
- 557. National Council for Therapy and Rehabilitation through Horticulture. Horticultural Therapy Research Project, Snoddy, G. D., & Relf, P. D. (1978). *Innovative horticultural therapy programs* W. J. Cummings, (Ed.). Mount Vernon, VA: The Council.
- 558. Nebbe, L. L. (1995). *Nature as a guide: Nature in counseling, therapy, and education* (2nd ed.). Minneapolis, MN: Educational Media Corp.
- 559. Neville, R. C. (1989). Recovery of the measure: Interpretation and nature. Albany, NY: State University of New York Press.
- 560. Newbury, T. (1996). *Ultimate garden designer* (1st paperback ed.). London, England; New York, NY: Ward Lock.
- 561. Newcomb, D. (1975). Postage stamp garden book: How to grow all the food you can eat in very little space. Los Angeles, CA; New York, NY: J.P. Tarcher. Includes bibliographical references (p. 142-143) and index.
- 562. Nicolaou, L. (1984). Sport and recreational opportunities for disabled people of immigrant origin: An exploratory study on disabled people of Greek background in Sydney. Sydney, Australia: Released by the Greek Welfare Centre. Includes bibliography (p. 35-37).
- 563. Nollman, J. (1994). Why we garden: Cultivating a sense of place (1st ed.). New York, NY: Holt.

- 564. Noyes, S. J. (1978). Effects of living and non-living horticultural materials on academic achievement and conduct of moderately retarded institutionalized individuals. Clemson, SC: Clemson University, AES.
- 565. Nursery Industry Association of Australia. (1992). Understanding the underlying needs & motivations for gardening: A market report for the Nursery Industry Association of Australia. Sydney, Australia: Australian Horticultural Corporation.
- 566. Obosu-Mensah, K. (1999). Food production in urban areas: A study of urban agriculture in Accra, Ghana. Aldershot, Hampshire, UK; Brookfield, VT. Includes bibliographical references and index.
- 567. O'Brien, J. (1992). Schoolyard garden designs: A guide to gardening with children (1st ed.). Rochester, NY: Praying Mantis Press. Includes bibliogrpahy (p. 89-94) and index.
- 568. Occupational therapy for children (1985). St. Louis, MO: Mosby.
- 569. Occupational therapy for children (1989). (2nd ed.). St. Louis, MO: C.V. Mosby.
- 570. Oelschlaeger, M. (1991). *Idea of wilderness: From prehistory to the age of ecology*. New Haven, CT: Yale University Press.
- 571. Olson, A. P. (1976). Development and implementation of an evaluated study of the effect of horticultural therapy on certain physically disabled patients. Clemson, SC: South Carolina Agricultural Experiment Station. Includes bibliography (p. 55-58).
- 572. Olson, M. (1994). MetroFarm: The guide to growing for big profit on a small parcel of land. Santa Cruz, CA: TS Books.
- 573. Olszowy, D. R. (1978). Horticulture for the disabled and disadvantaged. Springfield, IL: Thomas.
- 574. Olwell, C. (1990). *Gardening from the heart: Why gardeners garden*. Berkeley, CA: Antelope Island Press.
- 575. Ontario Public Interest Research Group. Toronto Chapter. (1986). *Growing city: A guide to community gardening in Toronto*. Toronto, Ontario, Canada: Ontario Public Interest Research Group.
- 576. Orcutt, G. (1977). Successful planters. Farmington, MI: Structures Pub. Co. Includes index.
- 577. Our little men and women: Modern methods of character building, a manual of

- work and recreation in which many valuable lessons are taught that prepare boys and girls for future usefulness (1912). Neil, H., (Ed.). Chicago, IL: Bible House.
- 578. Pack, C. L. (1919). War garden victorious. Philadelphia, PA: J. P. Lippincott Company.
- 579. Paludan, M. (1995). *Jord under neglene: Og fred i sjoelen* (2. opl. ed.). Copenhagen, Denmark: Samleren. Includes index.
- 580. Parker, S. D., Will, C., & Burke, C. (1993). *Activities for the elderly*. Ravensdale, WA: Idyll Arbor.
- 581. Parks Council. (1969). Little about lots: The do-it-yourself book on improving vacant lots and neighborhoods. New York, NY: The Parks Council. Cover title.
- 582. Parsons, A. (1985). Resource guide for horticultural therapy programs for botanical gardens and arboreta. Gaithersburg, MD: National Council for Therapy and Rehabilitation through Horticulture.
- 583. Pays, I. (1985). *Gardening in retirement*. Mitcham, Surrey, UK: Age Concern England. Includes index.
- 584. Pedneault, A., & Grenier, R. (1996). Creer un jardin communautaire: L'amenager, le gerer, l'animer. Montreal, Canada: Ville de Montréal; MAB metro.
- 585. Perlman, M. (1994). *Power of trees: The reforesting of the soul.* Dallas, TX: Spring Publications.
- 586. Phillips, R., & Rix, M. (1998). Random house book of plants for pots and patios: [and how to grow them] (1st U.S. ed.). New York, NY: Random House. Includes index.
- 587. Phillips, S. (1994). *Creative step-by-step guide to container gardening*. North Vancouver, British Columbia, Canada: Whitecap Books.
- 588. Pierce, J. H., & Barnsley, R. (1993). Easy lifelong gardening: A practical guide for seniors. North Pomfret, VT: Trafalgar Square Publishing.
- 589. Playgrounds and playspaces for the handicapped (1974). s.l.: Theraplan.
- 590. Please P. (Ed.). (1990). Able to garden: A practical guide for disabled and elderly gardeners. London, England: Batsford.
- 591. Plummer, B., & Shewan, D. (1992). City gardens: An open spaces survey in the city of London. London, England: Belhaven Press.

- 592. Pollan, M. (1991). Second nature: A gardener's education. New York, NY: Atlantic Monthly Press. Includes bibliographical references.
- 593. Ponting, C. (1991). *Green history of the world*. London, England: Sinclair-Stevenson.
- 594. Porteous, J. D. (1990). Landscapes of the mind: Worlds of sense and metaphor. Toronto, Canada: University of Toronto Press.
- 595. Porter, E. R. (1994). *All visitors welcome: Accessibility in state park interpretive programs and facilities* M. A. Helmich, & D. C. Pozzi, (Eds.). Sacramento, CA: California Dept. of Parks and Recreation.
- 596. Portnoy, E. J. (1999). *Make their days: Activities for residents in long-term care*. Springfield, IL: Charles C. Thomas.
- 597. Potter, J. (1998). Secret gardens. London, England: Conran Octopus. Includes index.
- 598. Practical step-by-step guide to complete container gardening (1997). North Vancouver, British Columbia, Canada: Whitecap Books. Includes index.
- 599. Proshansky, H. M., Ittelson, W. H., & Rivlin L. G. (Eds.). (1970). *Environmental psychology: Man and his physical setting.* New York, NY: Holt, Rinehart and Winston.
- 600. Proshansky, H. M., Ittelson, W. H., & Rivlin L. G. (Eds.). (1976). Environmental psychology: People and their physical settings (2nd ed.). New York, NY: Holt, Rinehart and Winston.
- 601. Putting health into place: Landscape, identity, and well-heing (1998). (1st ed.). Syracuse, NY: Syracuse University Press. Includes bibliographical references (p. 297-336).
- 602. Rahel, E. M. (1986). Garden organizer: A guide to community gardening in Cleveland. Pownal, VT; Cleveland, OH: National Gardening Association; City of Cleveland Community Development Department Summer Sprout Program, Cuyahoga County Cooperative Extension Service Seed to Shelf Urban Gardening Program.
- 603. Rawlings, R. (1998). Healing gardens. Minocqua, WI: Willow Creek Press.
- 604. Rawson, H. (1901). Success in market gardening: A new vegetable growers' manual (Rev. and enl. ed.). New York, NY: Doubleday, Page & company.
- 605. Rawson, W. W. (Warren W.). (1887). Success in market gardening: And

- vegetable growers' manual. Boston, MA: Pub. by the author.
- 606. Rawson, W. W. (Warren W.). (1892). Success in market gardening: A new vegetable growers' manual .. (7th rev. and enl. .. ed.). Boston, MA: Rawson.
- 607. Rees, Y., & Palliser, D. (1990). *Container gardening*. Ramsbury, England: Crowood Press Ltd.
- 608. Relf, D. Horticultural therapy bibliography The Author?.
- 609. Relf, D. (Ed.). (1992). Role of horticulture in human well-being and social development: A national symposium, 19-21 April 1990, Arlington, Virginia. Portland, OR: Timber Press.
- 610. Relf, D. (Ed.). (1994). *Virginia master gardener handbook*, (rev. ed.). Blacksburg, VA: Virginia Cooperative Extension Service. Includes index.
- 611. Relf, P. D. (1980). Establishment of horticultural training centers for the mentally retarded. Mt. Vernon, VA: National Council for Therapy and Rehabilitation Through Horticulture.
- 612. Relph, E. C. (1976). Place and placelessness. London, England: Pion.
- 613. Rencontre Europeenne Environnement et Handicap (1st: 1987: Jouyen-Josas, France). (1988). Environnement et handicap: Les jardins dans la vill: Recontre de jouy-en-Josas, France organisee par le Relais nature Jouy-Velizy; 5-6-7 Novembre 1987 A. Mazas, (Ed.). France:
- 614. Rensch, B. (1971). *Biophilosophy* (C. A. M. Sym, Trans.). New York, NY: Columbia University Press.
- 615. Resident gardening in social housing environments: Identifying the need (1983). . Ottawa, Canada: Canada Mortgage and Housing Corporation.
- 616. Restuccio, J. P. (1992). Fitness the dynamic gardening way: A health and wellness lifestyle. Cordova, TN; Traverse City, MI: Balance of Nature Pub; Publishers Distribution Service.
- 617. Richey, D. (1976). *Child's introduction to the outdoors.* Toronto; New York, NY: Pagurian Press.
- 618. Rider, K. (Ed.). (1987). Wood projects for the garden. (Rev. ed.). San Francisco: Ortho Books.
- 619. Riotte, L. (1987). Sleeping with a sunflower: A treasury of old-time gardening lore. Pownal, VT: Storey Communications.

- 620. Robinette, G. O. (1972). Plants, people, and environmental quality: A study of plants and their environmental functions. Washington, DC: U.S. Dept. of the Interior, National Park Service.
- 621. Robinette, G. O. (1976). Roofscape: Roof top landscape development. Reston, VA: Environmental Design Press.
- 622. Robinette G. O. (Ed.). (1984). How to make cities liveable: Design guidelines for urban homesteading. New York, NY: Van Nostrand Reinhold Co.
- 623. Robinette G. O. (Ed.). (1985). Barrier-free exterior design: Anyone can go anywhere. New York, NY: Van Nostrand Reinhold.
- 624. Rodaway, P. (1994). Sensuous geographies: Body, sense, and place. London, England; New York, NY: Routledge.
- 625. Rohde, C. L. E., & Kendle, A. D. (1994). *Human well-being, natural landscapes and wildlife in urban areas: A review.* Peterborough, Cambridgeshire, UK: English Nature. Includes bibliographical references (p. 162-181).
- 626. Rohde, E. S. (1931). Scented garden. London, England: Medici Society.
- 627. Rohde, E. S. (1948). *Scented garden* (rev. & enl. ed.). London, England: Medici Society.
- 628. Roopnarine, J. L., Johnson, J. E. and Hooper, F. H. (Eds.). (1994). *Children's play in diverse cultures*. Albany, NY: State University of New York Press.
- 629. Ross, A. (1980). *Gardening with children*. London, England; Boston, MA: Faber and Faber. Includes index.
- 630. Roszak, T. (1992). *Voice of the earth.* New York, NY: Simon & Schuster. Includes bibliographical references (p. 353-358) and index.
- 631. Roszak, T., Gomes, M. E., & Kanner, A. D. (Eds.). (1995). *Ecopsychology: Restoring the earth, healing the mind.* San Francisco, CA: Sierra Club Books.
- 632. Roth, S. A. (1991). Weekend garden guide: Work-saving ways to a beautiful backyard. Emmaus, PA: Rodale Press.
- 633. Rothert, G. (1994). Enabling garden: A guide to lifelong gardening. Dallas, TX: Taylor Pub. Co.
- 634. Rouard, M., & Simon, J. (1977). *Children's play spaces: From sandbox to adventure playground* ((L. Geiser, Trans.). Woodstock, NY: Overlook Press.

- 635. Rouner, L. S. (Ed.). (1984). *On nature* (Reprint ed.). Notre Dame, IN: University of Notre Dame Press.
- 636. Rowles, G. D., & Ohta, R. J. (Eds.). (1983). Aging and milieu: Environmental perspectives on growing old. New York, NY: Academic Press. Includes bibliographies and indexes
- 637. Rowntree, B. S., & Lasker, B. (1911). *Unemployment: A social study*. London, England: MacMillan.
- 638. Rowntree, B. S., & Lasker, B. (1980). *Unemployment: A social study*. New York, NY: Garland Pub.
- 639. Rowson, N. J., & Thoday, P. R. (1981). Raised bed planters for the disabled: A study of the design, construction and use of raised planters for disabled people. Bath, England: Rehabilitation and Medical Research Trust, University of Bath. Includes bibliographical references (leaves 132-134)
- 640. Rowson, N. J. & Thoday, P. R. (1997). Accessible gardening: A guide to raised planters and modified growing techniques. (Rev. ed.). J. Stoneham & P. R. Thoday, (Eds.). Gillingham, Dorset, England: Horticulture for All.
- 641. Rowson, N. J., Thoday, P. R., & Croft, P. M. (1985). Landscape design for disabled people in public open space P. M. Croft, (Ed.). Bath, England: University of Bath. Includes bibliography (p. 127-130).
- 642. Ruse, M. (1973). *Philosophy of biology*. London, England: Hutchinson. Includes index and bibliography (p. [219]-224).
- 643. Ruse, M. (1988). *Philosophy of biology today*. Albany, NY: State University of New York Press.
- 644. Ruse, M. (Ed.). (1989). What the philosophy of biology is: Essays dedicated to David Hull. Dordrecht; Boston, MA; Norwell, MA: Kluwer Academic. Includes bibliographical references and index.
- 645. Rutledge, A. J. (1971). Anatomy of a park: The essentials of recreation area planning and design. New York, NY: McGraw-Hill. Includes bibliography (p. 173-176).
- 646. Rutledge, A. J. (1981). *Visual approach to park design*. New York, NY: Garland STPM Press. Includes index and bibliography (p. 169-176).
- 647. Rutledge, A. J., & Molnar, D. J. (1986). *Anatomy of a park: The essentials of recreation area planning and design* (2nd ed.). New York, NY: McGraw-Hill. Includes bibliography (p. 183-185).

- 648. Sandhu, J. S., & Hendriks-Jansen, H. (1976). Environmental design for handicapped children. Farnborough, Hampshire, UK: Saxon House. Includes bibliography (p. 219-222).
- 649. Sarafino, E. P. (1990). *Health psychology: Biopsychosocial interactions*. New York, NY: Wiley. Includes bibliographical references.
- 650. Sarafino, E. P. (1994). *Health psychology: Biopsychosocial interactions* (2nd ed.). New York, NY: Wiley.
- 651. Sattler, R. (1986). *Biophilosophy: Analytic and holistic perspectives*. Berlin, Germany; New York, NY: Springer-Verlag. Includes index and bibliography (p. [257]-277).
- 652. Schaal, H. D. (1994). *Landscape as inspiration*. London, England; Berlin, Germany: Academy Editions; Ernst & Sohn. Includes bibliographical references (p. [379]).
- 653. Schaier, A. (1986). *Gartenarbeit fur Korperbehinderte und Senioren*. Dortmund: Verlag Modernes Lernen.
- 654. Schar, D. (1994). *Dump your stress in the compost pile!*. Washington, DC: Elliott & Clark Pub.
- 655. Scharf, B., & Reynolds, G. S. (Eds.). (1975). Experimental sensory psychology. Glenview, IL: Scott, Foresman.
- 656. Scheffer, M. (1988). *Bach flower therapy: Theory and practice* ((A. R. Meuss, Trans.). Rochester, VT: Healing Arts Press.
- 657. Schiffman, H. R. (1976). Sensation and perception: An integrated approach. New York, NY: Wiley.
- 658. Schiffman, H. R. (1982). Sensation and perception: An integrated approach (2nd ed.). New York, NY: Wiley. Includes indexes and bibliography (p. 463-518).
- 659. Schiffman, H. R. (1990). Sensation and perception: An integrated approach (3rd ed.). New York, NY: Wiley. Includes bibliographical references (p. 493-536) and index.
- 660. Schiffman, H. R. (1996). Sensation and perception: An integrated approach (4th ed.). New York, NY: Wiley. Includes bibliographical references and index.
- 661 Schleien, S. J., & Ray, M. T. (1988). Community recreation and persons with disabilities. Baltimore, MD: P.H. Brooks Publ. Includes index and

- bibliography (p. 263-269).
- 662. Schleien, S. J., Ray, M. T., & Green, F. P. (1997). *Community recreation and people with disabilities: Strategies for inclusion* (2nd ed.). Baltimore, MD: P.H. Brooks Publ. Includes bibliographical references (p. 317-326) and index
- 663. Schmidt, R. F. (1973). Grundriss der Sinnesphysiologie: Mit 109 Testfragen z. selbstkontrolle. Berlin, Germany; Heidelberg, Germany; New York, NY: Springer.
- 664. Schmidt, R. F. (1977). *Grundriss der Sinnesphysiologie* (3., uberarb. und erw. Aufl. ed.). Berlin, Germany; New York, NY: Springer-Verlag.
- 665. Schmidt, R. F., & Altner H. (Eds.). (1978). Fundamentals of sensory physiology (M. A. Biederman-Thorson, Trans.). (Springer study ed.). New York, NY: Springer-Verlag.
- 666. Schmidt, R. F., & Altner, H. (Eds.) (1986). Fundamentals of sensory physiology (M. A. Biederman-Thorson, Trans.). (3rd., rev. and expand ed.). Berlin, Germany; New York, NY: Springer-Verlag.
- 667. Schoggen, P. (1989). Behavior settings: A revision and extension of Roger G. Barker's Ecological psychology. Stanford, CA: Stanford University Press. Includes bibliographical references (p. [387]-403).
- 668. Schroeder, S. R. (Ed.). (1990). *Ecobehavioral analysis and developmental disabilities: The twenty-first century*. New York, NY: Springer-Verlag.
- 669. Schultz, W. (1995). City gardens. Boston, MA: Little, Brown and Company. Includes index.
- 670. Schuman, H. (1998). Gardening within arm's reach: Gardening and experiencing nature for the visually handicapped. how to set up a garden with this in mind (G. Broadribb, Trans.). The Netherlands: Bartimeus.
- 671. Swartz J. L., & Martin, W. E. (Eds.). (1997). Applied ecological psychology for schools within communities: Assessment and intervention. Mahwah, NJ: Lawrence Erlbaum Associates.
- 672. Schwarz, C. (1977). *Get growing*. Washington, DC: International City Management Association. Includes bibliographical references.
- 673. Schweller, F. J. (1989). *Container gardening for the handicapped.* Phoenix, AZ: Hand-D-Cap Publishing.
- 674. Senda, M. (1992). Design of children's play environments. New York, NY: McGraw-Hill. Includes index.

- 675. Sensory Trust, & Horticulture for All. (1998). Plants and human well-being: Proceedings of a conference held at the University of Reading, 18th-19th September 1996, organised by the Sensory Trust & Horticulture for All (J. Stoneham, & T. Kendle, Eds.). Bath, England: Sensory Trust. Includes bibliographical references.
- 676. Shanahan, J., & McComas, K. (1999). *Nature stories: Depictions of the environment and their effects*. Cresskill, N.J.: Hampton Press. Includes bibliographical references (p. 179-189) and indexes.
- 677. Shapiro, J. G. (1999). Flower remedy book: A comprehensive guide to over 700 flower essences. Berkeley, CA: North Atlantic Books.
- 678. Shaw, N. (1998). Bach flower remedies: A step-by-step guide. Shaftesbury, Dorset, UK; Boston, MA: Element.
- 679. Sheen, J., & Alexander, C. (1991). *Dried flower gardening*. London, England: Ward Lock.
- 680. Sheen, J., & Alexander, C. (1996). *Dried flower gardening* (New ed.). London, England: Ward Lock.
- 681. Shipley, T. (1980). Sensory integration in children: Evoked potentials and intersensory functions in pediatrics and psychology. Springfield, IL: Thomas. Includes index and bibliography (p. 145-152).
- 682. Shivers, J. S., & Fait, H. F. (1975). *Therapeutic and adapted recreational services*. Philadelphia, PA: Lea & Febiger. Includes bibliographies and index.
- 683. Shoemaker, J., & Mattson, R. H. (1982). *Defining horticulture as a therapeutic modality*. Manhattan, KS: Dept. of Horticulture, Kansas State University. Includes bibliographies.
- 684. Simmons, A. F. (1975). *Potted orchards: Growing fruit in small spaces*. Newton Abbot, England; North Pomfret, VT: David and Charles.
- 685. Simmons, G. A., Cannon, E. C., American Association for Leisure and Recreation, & American Alliance for Health, Physical Education, Recreation and Dance. (1991). *It is outdoors: A guide to experiential activities*. Reston, VA: The Association.
- 686. Simon, Y. R. M. (1970). *Great dialogue of nature and space*. G. J. Dalcourt, (Ed.). Albany, NY: Magi Books. Includes bibliographical references.
- 687. Simoons, F. J. (1998). *Plants of life, plants of death*. Madison, WI: University of Wisconsin Press. Includes bibliographical references (p. 460-

- 535) and index.
- 688. Simson, S. P., & Straus, M. C. (Eds.). (1998). *Horticulture as therapy: Principles and practice*. New York, NY: Food Products Press.
- 689. Sinnott, E. W. (1966). *Bridge of life: From matter to spirit*. New York, NY: Simon and Schuster. Includes bibliographical footnotes.
- 690. Skelsey, A. F. (1971). Cucumbers in a flowerpot. New York, NY: Workman Pub. Col.
- 691. Skelsey, A. F. (1975). Farming in a flowerpot = formerly Cucumbers in a flowerpot (New and rev. ed.). New York, NY: Workman Pub. Col.
- 692. Slesin, S. (1996). *Garden tools*. New York, NY: Abbeville Press Publishers.
- 693. Smith, E. (1985). Therapeutic horticulture, a growing field: The role of horticulture in the rehabilitation of disabled people. Pearce, A.C.T., Australia: E. Smith. Includes bibliography (p. 361-371).
- 694. Smith, O. B. (Ed.). (1999). Agriculture urbaine en afrique de l'ouest: Une contribution à la sécurité alimentaire et à l'assainissement des villes = Urban agriculture in West Africa: Contributing to food security and urban sanitation. Ottawa, Ontario, Canada; Wageningen, Netherlands: International Development Research Centre; Technical Centre for Agricultural and Rural Cooperation. Includes bibliography.
- 695. Somekh, A. (1994). *Homeless garden project, 1991-1994*. Santa Cruz, CA: Includes bibliographical references.
- 696. Sommers, L. (1984). Community garden book: New directions for creating and managing neighborhood food gardens in your town. Burlington, VT: Gardens for All/National Association for Gardening. Includes bibliography (p. 108-109).
- 697. Squire, D. (1998). Healing garden: Natural healing for mind, hody, and soul. Lincolnwood, IL: Contemporary Books. Includes bibliographical references (p. 140-141) and index.
- 698. Squire, D., & Newdick, J. (1989). Scented garden: How to grow and use beautiful plants to create a harmony of fragrances for garden and home. Emmaus, PA: Rodale Press. Includes index.
- 699. Squire, D., & Newdick, J. (1996). Scented garden: Creating fragrance and beauty in the home and the garden with a rich diversity of plants and flowers. London,

- England: Salamander Book.
- 700. Staines, R. (1991). *Market gardening: Growing and selling produce*. Golden, CO: Fulcrum Pub. Includes bibliographical references (p. 153) and index.
- 701. Staples, K. (1997). *Plant parenthood for urban gardeners*. New York, NY: Macmillan. Includes index.
- 702. Stauffer, M. (1979). Gemusebau fur Gemeinschaften = Vegetable gardening for the community. Dornack, Schweiz: R. Geering.
- 703. Stebbins, C. A. (1913). Principles of agriculture through the school and the home garden. New York, NY: Macmillan Company.
- 704. Steinfeld, E., & Danford, G. S. (Eds.). (1999). *Enabling environments: Measuring the impact of environment on disability and rehabilitation.* New York, NY: Kluwer Academic/Plenum Publishers. Includes bibliographical references and index
- 705. Stensrud, C. J. (1993). Training manual for Americans with Disabilities Act compliance in parks and recreation settings. State College, PA: Venture Publishing. Includes bibliographical references.
- 706. Step-by-step guide to creative patio and container gardening (1996). North Vancouver, British Columbia, Canada: Whitecap Books.
- 707. Stine, S. (1997). Landscapes for learning: Creating outdoor environments for children and youth. New York, NY: John Wiley & Sons. Includes bibliographical references (p. 231-239) and index.
- 708. Stokols, D. (Ed.). (1977). Perspectives on environment and behavior: Theory, research, and applications. New York, NY: Plenum Press.
- 709. Stokols, D., Altman, I. (Eds.). (1987). Handbook of environmental psychology. New York, NY: Wiley. Includes bibliographies and indexes.
- 710. Stoneham, J., & Thoday, P. R. (1994). Landscape design for elderly & disabled people. Chichester, England: Packard.
- 711. Stoneham, J., & Thoday, P. R. (1996). Landscape design for elderly and disabled people. Woodbridge, Suffolk, UK: Garden Art Press. Includes bibliographical references (p. 224-226) and index
- 712. Stout, R. (1973). How to have a green thumb without an aching back: A new method of mulch gardening. New York, NY: Cornerstone Library.
- 713. Stout, R. (1998). Gardening without work: For the aging, the busy and the

- indolent. New York, NY: Lyons Press.
- 714. Stout, R., & Clemence, R. V. (1971). Ruth Stout no-work garden book G. F. Johns, (Ed.). New York, NY: Bantam Books.
- 715. Sullivan, M. (1980). Where does your garden grow?: Horticulture programs with indoor activities. Buffalo, NY: Potentials Development for Health & Aging Services. Includes bibliography (p. 61).
- 716. Symposium on Horticulture for the People (1978: University of Sydney). (1981). Symposium on Horticulture for the People: XXth International Horticultural Congress, the University of Sydney, Sydney, Australia, 17 August 1978 H. B. Tukey, Jr, (Ed.). The Hague, Netherlands: International Society for Horticultural Science. Includes bibliographies.
- 717. Symposium on Human Adaptation to Environment. Relative Impacts of Physical Environments and Sociocultural Factors (11th: 1983: Vancouver, B.C.). (1985). Human adaptation to environment: Proceedings of the Symposium on Human Adaptation to Environment: Relative impacts of physical environments and sociocultural factors, XIth I.C.A.E.S., Vancouver, Canada, 20-25 August 1983 A. Basu, & B. Mukhopadhyay, (Eds.). Calcutta: Indian Anthropological Society. Includes bibliographies.
- 718. Taloumis, G. (1972). Container gardening outdoors. New York, NY: Simon and Schuster.
- 719. Taloumis, G. (Ed.). (1987). *Container gardening*. Brooklyn, NY: Brooklyn Botanic Garden.
- 720. Tarling, T. (1994). *Container garden: A practical guide to planning & planting*. New York, NY: Sterling Pub. Co. Includes index.
- 723. Tarrant, D. (1989). Highrise horticulture: A guide to gardening in small spaces. St. Croix, MN: Tew Press. Includes index.
- 722. Taylor, J. (1990). *Planning your garden for retirement. Melbourne: Lothian.* Includes bibliographical references (p. 63) and index.
- 723. Taylor, P. A. (1996). Easy care native plants: A guide to selecting and using beautiful American flowers, shrubs, and trees in gardens and landscapes. New York, NY: H. Holt.
- 724. Teuscher, H. (1956). Window-box gardening: An illustrated guide for the outdoor culture of plants in boxes, tubs, and hanging baskets. New York, NY: Macmillan.

- 725. Thoday, P. R., & Robinson, D. W. (Eds.). (1987). Scientific management of vegetation in the urban environment: University of Bath, England, July 8-11, 1985. Wageningen, Netherlands: Commission for Urban Horticulture. Includes bibliographies.
- 726. Thomas, W. H. (1996). Life worth living: How someone you love can still enjoy life in a nursing home: The Eden alternative in action. Acton, MA: VanderWyk & Burnham. Includes bibliographical references and index.
- 727. Thorpe, W. H. (1961). *Biology, psychology, and belief.* Cambridge, England: University Press. Includes bibliography.
- 728. Tilgner, L. (1985). Tips for the lazy gardener. Pownal, VT: Storey Communications.
- 729. Titman, W. (1994). Special places, special people: The hidden agenda of school grounds. Surrey, United Kingdom: WWF UK, World Wide Fund for Nature.
- 730. Tolliday, L., & Chivers, D. (1988). *Gardening for everyone*. Wembly, Western Australia, Australia: Western Australian Government Department for Sport and Recreation.
- 731. Tompkins, P., & Bird, C. (1973). Secret life of plants. New York, NY: Harper & Row. Includes bibliography (p. [375]-392).
- 732. Toogood, A. R. (1988). *Container gardening*. London, England: Ward Lock. Includes index.
- 733. Tower Hamlets. (1985). *Community gardening: Year two report, 1984/85*. London, England: London Borough Council. Directorate of Development.
- 734. Trombly, C. A., & Scott, A. D. (1977). Occupational therapy for physical dysfunction. Baltimore, MD: Williams & Wilkins Co. Includes bibliographical references and index.
- 735. Truex, P. (1963). *City gardener*. New York, NY: Knopf. Includes bibliography (p. 341, 344)..
- 736. Tuan, Y. (1993). Passing strange and wonderful: Aesthetics, nature, and culture. Washington, DC: Island Press/Shearwater Books.
- 737. Tufts University. School of Nutrition. (1987). Sustaining agriculture near cities W. Lockeretz, (Ed.). Ankeny, IA: Soil and Water Conservation Society.
- 738. Tyson, M. M. (1998). Healing landscape: Therapeutic outdoor environments. New York, NY: McGraw-Hill.

- 739. Ulvund, S. E. (1989). Cognitive development in infancy: A study with emphasis on physical environmental parameters. Oslo; Atlantic Highlands, NJ: Solum forlag; Humanities Press International. Includes bibliographical references.
- 740. United States. (1998). Act to require the Secretary of Agriculture and the Secretary of the Iinterior to conduct a study to improve the access for persons with disabilities to outdoor recreational opportunities made available to the public. Washington, DC: U.S. G.P.O.
- 741. United States. Forest Service. Alaska Region. (1998). *Alaska region regional accessibility strategy for recreation programs and administrative sites/facilities*. Juneau, AK: The Region.
- 742. United States. General Accounting Office. (1995). School facilities [microform]: Accessibility for the disabled still an issue: Report to congressional requesters. Washington, DC; Gaithersburg, MD: The Office; The Office distributor. Includes bibliographical references.
- 743. United States. National Park Service. (1971). National park guide for the handicapped. Washington, DC: U. S. Govt. Print. Off. Cover title.
- 744. Universal access to outdoor recreation: A design guide (1993). Berkeley, CA: PLAE.
- 745. University of British Columbia. Botanical Garden. Hortitherapy Committee, Walline, M. E. C., & Tarrant, D. (1978). *Gardening as therapy*. (Vo.ls. 1-4). Vancouver, British Columbia, Canada: Botanical Garden, University of British Columbia. Includes bibliographies.
- 746. Upson, S. (1997). Permanent raised bed gardening: Novel raised bed designs for home and market gardens. Ardmore, OK: The Samuel Roberts Noble Foundation Inc..
- 747. Upson, S. (1990). Permanent raised bed gardening: Advanced cultural technologies for home and market garden. Ardmore, OK: The Samuel Roberts Noble Foundation Inc..
- 748. Upton, A. J., & Thoday, P. R. (1980). *Bibliography of horticulture and out-door amenities for the handicapped and disadvantaged.* Bath, England: Department of Horticulture, University of Bath.
- 749. Urban agriculture: Food, jobs and sustainable cities (1996). New York, NY: United Nations Development Programme. Includes bibliographical references
- 750. Valnet, J. (1966). Aromatherapi: Traitement des maladies par les essences des

- plantes (2., revue, corr. et augm. ed.). Paris, France: Maloine.
- 751. Valnet, J. (1975). Aromatherapie: Traitement des maladies par les essences des plantes (7. revue, corr. et augm. ed.). Paris, France: Maloine. Includes bibliography (p. [429]-434).
- 752. van Epenhuijsen, C. W. (1974). *Growing native vegetables in Nigeria*. Rome; New York, NY: Food and Agriculture Organization of the United Nations.
- 753. Verey, R. (1981). Scented garden. New York, NY: Van Nostrand Reinhold Co.
- 754. Verey, R. (1989). *Scented garden*. New York, NY: Random House. Includes bibliographical references (p. 167)
- 755. Verey, R., & Lambert, K. (1994). Secret gardens: Revealed by their owners (R. Verey, & K. Lambert, Eds.), (1st North American ed.). Boston, MA: Little, Brown..; "A
- 756. Vlamis, G. (1986). Flowers to the rescue: The healing vision of Dr. Edward Bach, incorporating ye suffer from yourselves and free thyself. Wellingborough, Northamptonshire, United Kingdom; New York, NY: Thorsons Pub. Group. Includes indexes and bibliography (p. 159).
- 757. Walden, F.R.J. (1997). Garden for you: A practical guide to tools, equipment and design for older people and people with disabilities. London, England: Disabled Living Foundation. Includes bibliographical references and index.;
- 758. Walden, F.R.J., Baxendale, K.M. & Kelsall, A.D. (1997). *Gardening: an equipment guide*. K.M. Baxendale, A.D. Kelsall & G.M. Cochrane (Eds.). Oxford, England: Disability Information Trust.;
- 759. Wales Council for the Disabled. (1993). Accessible landscapes: Based on a review of Garden Festival Wales, 1992 C.Thomas, & J. Thompson (Eds.), (1st ed.). Caerphilly: Wales Council for the Disabled...
- 760. Wall, D. (1994). Green history: A reader in environmental literature, philosophy, and politics. London, England; New York, NY: Routledge. Includes bibliographical references (p. 258-262) and index.
- 761. Walmsley, D. J. (1976). Gardening in the city: Backyards, balconies, terraces, and penthouses. New York, NY: Harcourt Brace Jovanovich. Includes index and bibliography (p. [168]-171).
- 762. Walmsley, D. J., & Day, R. A. (1972). Perception and man-environment

- interaction: A bibliography and guide to the literature. Armidale, New South Wales, Australia: Geographical Society of New South Wales, New England Branch.
- 763. Wals, A. E. J. (1994). Pollution stinks!: Young adolescents' perceptions of nature and environmental issues with implications for education in urban settings. De Lier, The Netherlands: Academic Book Center.
- 764. Walsh, W. B., Craik, K. H., and Price, R. H. (Eds.). (1992). *Person-environment psychology: Models and perspectives*. Hillsdale, NJ: L. Erlbaum Associates.
- 765. Walter, E. V. (1988). *Placeways: A theory of the human environment*. Chapel Hill: University of North Carolina Press. Includes index and bibliography (p. 233-246).
- 766. Wapner, S., Cohen, S. B., & Kaplan, B. (Eds.). (1976). Experiencing the environment. New York, NY: Plenum Press.
- 767. Ward, R. A., La Gory, M., & Sherman, S. (1988). *Environment for aging: Interpersonal, social, and spatial contexts*. Tuscaloosa, AL: University of Alabama Press. Includes index and bibliography (p. 243-253).
- 768. Wardle, C. (1983). *City farming and community gardening. Book 1*. London, England; New York, NY: Inter-Action. Includes index.
- 769. Warner, S. B., & Durlach, H. (1987). *To dwell is to garden: A history of Boston's community gardens.* Boston, MA: Northeastern University Press. Includes bibliographies and index.
- 770. Waterman, M. P. (1995). Self-sufficiency gardening: Financial, physical, and emotional security from your own backyard: A practical guide for growing vegetables, fruits, nuts, herbs, medicines, and other useful products. Port Townsend, WA: Loompanics Unlimited.
- 771. Watson, D. P., & Burlingame, A. W. (1960). *Therapy through horticulture*. New York, NY: Macmillan. Includes bibliographical references.
- 772. Webber, R. (1972). Market gardening: The history of commercial flower, fruit and vegetable growing. Newton Abbot, United Kingdom: David and Charles. Includes bibliography (p. 205-208).
- 773. Wells, S. E., American Horticultural Therapy Association, & Friends of Horticultural Therapy. (Eds.). (1997). *Horticultural therapy and the older adult population*. New York, NY: Haworth Press.

- 774. Wellsted, T. (1987). *Patio and windowbox gardening*. New York, NY: Exeter Books.
- 775. Wheelwright, J. H., & Schmidt, L. W. (1991). Long shore: A psychological experience of the wilderness. San Francisco, CA: Sierra Club Books.
- 776. White, E. T. (1999). *Path-portal-place: Appreciating public space in urban environments*. Tallahassee, FL: Architectural Media.
- 777. Wilderness accessibility for people with disabilities [microform]: A report to the President and the Congress of the United States on Section 507(a) of the Americans with Disabilities Act (1992). Washington, DC: National Council on Disability.
- 778. Wildwood, C. (1996). *Encyclopedia of aromatherapy* (1st U.S. ed.). Rochester, VT: Healing Arts Press.
- 779. Williams, A. (Ed.). (1999). Therapeutic landscapes: The dynamic between place and wellness. Lanham, MD: University Press of America. Includes index.
- 780. Williams, C. (1999). *Bringing a garden to life* (Bantam trade paperback ed.). New York, NY: Bantam Books. Includes bibliographical references (p. [255]-264) and index.
- 781. Williams, D. (1911). *Gardens and their meaning*. Boston, MA; New York, NY; etc.: Ginn and company.
- 782. Williams, P. N., Zajicek, J. M., Bradley, J. C., & Texas A & M University. Dept. of Horticultural Sciences (Eds.). (1996). *People-plant interactions in urban areas: Proceedings of a research and education symposium, San Antonio, Texas, May 23-26, 1996*. College Station, TX: Dept. of Horticultural Sciences, Texas A&M University.
- 783. Williams, T. J., & Editorial staff of Ortho Books. (1981). How to select, use & maintain garden equipment. San Francisco, CA: Ortho.
- 784. Wilson, A. (1949). How does your garden grow?. Menlo Park, CA: Happy Hours.
- 785. Wilson, E. O. (1975). *Sociobiology: The new synthesis*. Cambridge, MA: Belknap Press of Harvard University Press. Includes bibliographical references (p. 599-663) and index.
- 786. Wilson, E. O. (1984). *Biophilia*. Cambridge, MA: Harvard University Press.
- 787. Wilson, E. O. (1996). *In search of nature*. Washington, DC: Island Press. Includes bibliographical references (p. 201-202) and index.

- 788. Wilson, J. W. (1990). Landscaping with container plants. Boston, MA: Houghton Mifflin. Includes index.
- 789. Winslow R. M., Halberg, K. J. (Eds.). (1992). *Management of therapeutic recreation services*. Arlington, VA: National Recreation and Park Association.
- 790. Winter, D. D. N. (1996). Ecological psychology: Healing the split between planet and self. New York, NY: HarperCollins College Publishers. Includes bibliographical references and index.
- 791. Wohlwill, J. F., & Weisman, G. D. (1981). *Physical environment and behavior: An annotated bibliography and guide to the literature.* New York, NY: Plenum Press. Includes index.
- 792. Worker's Educational Association of South Australia. Community Education Centre. (1980-1988). *Gardening the easy way: (for people with weak hands, short breath, no legs, or other difficulties)* (Vols. 1-2). Adelaide, South Australia; WEA Community Education Centre..
- 793. Woy, J. (1997). Accessible gardening: Tips and techniques for seniors and the disabled. Mechanicsburg, PA: Stackpole Books..
- 794. Wright M., & Brown D. (Eds.). (1975). Complete indoor gardener (1st American ed.). New York, NY: Random House.
- 795. Wright M., & Brown D. (Eds.). (1979). *Complete indoor gardener* (Rev. ed.). New York, NY: Random House.
- 796. Wyburn, G. M., Pickford, R. W., & Hirst, R. J. (1964). *Human senses and perception*. G. M. Wyburn, (Ed.). Toronto, Canada: University of Toronto Press. Includes bibliographies.
- 797. Wydra, N. (1997). Feng shui in the garden: Simple solutions for creating comforting, life-affirming gardens of the soul. Chicago, IL: Contemporary Books.
- 798. Yang, L. (1975). Terrace gardener's handbook: Raising plants on a rooftop, penthouse, or patio. Garden City, NY: Doubleday. Includes index and bibliography (p. 263-267).
- 799. Yang, L. (1982). Terrace gardener's handbook: Raising plants on a balcony, terrace, rooftop, penthouse, or patio. Beaverton, OR: Timber Press.
- 800. Yang, L. (1990). *City gardener's handbook: From balcony to backyard.* New York, NY: Random House. Includes bibliographical references (p. 297-299) and index.
- 801. Yang, L. (1995). City and town gardener: A handbook for planting small spaces

- and containers (1st pbk. ed.). New York, NY: Random House.
- 802. Yeaw, F. L. (1915). Market gardening. New York, NY: J. Wiley & sons, inc.
- 803. Yeomans, K. (1992). Able gardener: Overcoming barriers of age & physical limitations. Pownal, VT: Storey Communications. Includes bibliographical references and index.
- 804. Zandstra, P. J., McCurry, E., & Mitchell S. (Eds.). (1982). Horticultural therapy: Meeting the needs of special populations, the elderly, the mentally impaired, the mentally ill, substance abusers, criminal offenders. East Lansing, MI: National Council for Therapy and Rehabilitation through Horticulture. Michigan Chapter. Includes bibliographies.
- 805. Zeisel, J. (1981). *Inquiry by design: Tools for environment-behavior research.* Monterey, CA: Brooks/Cole Pub. Co.4 Includes bibliographical references (p. 232-239) and indexes.
- 806. Zgola, J. M. (1987). *Doing things: A guide to programming activities for persons with Alzheimer's disease and related disorders.* Baltimore, MD: Johns Hopkins University Press. Includes bibliography (p. 129-131).

Braille books

- 807. Eyre, F. (1988?). Manual for blind gardeners. [s.l.: s.n.]
- 808. Stevens, L. (1962). *Garden unseen: A manual for blind gardeners.* (Vols. 1-2). London, England: Royal National Institute for the Blind.

2 PAMPHLETS

(Items with 50 or fewer pages)

- P1. Adaptive garden equipment: A resource manual for patients, families, and professionals (1986). Englewood, CO: Craig Hospital.
- P2. Aiken, K., Bowns, C., Fellinger, M., & Francis, M. (1995). Healing environments: A collection of case studies. University of California, Davis, CA: Center for Design Research.
- P3. Allan, N. (1993). Personal philosophy and practical approach to horticultural therapy. Glen Waverly, Victoria, Australia: Horticultural Therapy Association of Victoria.
- P4. Austin, D. R. (Ed.). (1980). Therapeutic recreation curriculum: Philosophy, strategy, and concepts. Bloomington, IN: Indiana University School of Health, Physical Education, and Recreation. Includes bibliographies.
- P5. Barrier-free recreation on the Gifford Pinchot National Forest (1994). Washington, DC: U.S. Dept. of Agriculture, Forest Service, Pacific Northwest Region, Gifford Pinchot National Forest.
- P6. Barrier-free recreation opportunities: Huron-Manistee National Forests (1996). Washington, DC: Forest Service, U.S. Dept. of Agriculture.
- P7. Beattie, W. R. (1911). Frames as a factor in truck growing. Washington, DC: U.S. Dept. of Agriculture.
- P8. Beechel, J. (1980). Interpretation for handicapped persons: A handbook for outdoor recreation personnel. Washington, DC: distributed by Hawkins

- and Associates. Includes bibliography (p. 31-35).
- P9. Central Council for the Disabled, London. (1968). Gardening for the disabled: Some hints and suggestions. London, England: Disabled Living Activities Group of the Central Council for the Disabled.
- P10. Cole, N. (1964). Special gardening for special people. Andover, Hampshire, England: Author, the Thatch, Longparish Common.
- P11. Container gardening (1978). Ottawa, Canada: Agriculture Canada. Includes bibliographical references.
- P12. Crafts from the garden (1995). Ryde, New South Wales, Australia: Horticultural Therapy Society of NSW.
- P13. Down to Earth: Horticultural Therapy in Corrections (1982: Kansas City, Mo.). (1982). Proceedings of Down to Earth: Horticultural therapy in corrections: March 5, 1982. J. Shoemaker, & R. H. Mattson, (Eds.). [s.l., s.n.].
- P14. Evans, R. (1984). Short cuts for less active gardeners: A guide to low-maintenance gardening. Ryde, New South Wales, Australia: Ryde and Hunter's Hill Councils; Ryde Horticultural College; Ryde Home Care Service.
- P15. Ezell, D.O., Jones, E.V., & Olson, A.P. (1981). Outdoor gardening for the handicapped. Clemson, SC: Clemson University, Cooperative Extension Service. Includes bibliographical references (p. 8).
- P16. Femal, J. T. (1967). Children's gardening. Washington, DC: National Recreation and Park Association. Includes bibliography (p. 32).
- P17. Fleet, K. (197?). Methods and aids for blind gardeners. London, England: Royal National Institute for the Blind.
- P18. Fleet, K. (1978). Manual for blind gardeners: Some suggested aids (Rev. ed.). London, England: Royal National Institute for the Blind.
- P19. Fleet, K. (1979). Manual for Blind gardeners: Suggested aids. [London, England: Royal National Institute for the Blind?] .
- P20. Fleet, K. (198?). Methods and aids for blind gardeners. London, England: Royal National Institute for the Blind.
- P21. Foster, S. Duvall, C. & Powell, J. (1992). Adaptive gardening techniques for the visually impaired. Corvallis, OR: Oregon State University. Extension Service. . . Includes bibliographical references (p.3-4).

- P22. Foster, S. & Powell, J. (1992a). Gardening ideas for children with special needs. Corvallis, OR: Oregon State University. Extension Service. . . Includes bibliographical references (p. 2).
- P23. Foster, S. & Powell, J. (1992b). Gardening strategies for people with heart and lung problems. Corvallis, OR: Oregon State University. Extension Service. . . Includes bibliographical references (p. 2).
- P24. Fox, S. (1975). Special education classes: For use with elementary grades 1-6 by hortitherapy volunteer groups. Clemson, SC: Horticulture Dept., South Carolina Agricultural Experiment Station, Clemson University.
- P25. Fredericks, D. (1987). Issues in botanic garden design. Seattle, WA: University of Washington. Includes bibliography (p. 25).
- P26. Frojmovic, M. (1996). Urban agriculture in Canada: A survey of municipal initiatives in Canada and abroad. Ottawa, Canada: International Development Research Centre.
- P27. Garden Club of Toronto. (1997). Civic garden center in metropolitan Toronto, 1958-1997: A history. Toronto, E.M. Bryce, (Ed.). Ontario, Canada: Civic Garden Centre.
- P28. Gardening for the disabled. (1983-1984?). London, England: Disabled Living Activities Group. . .
- P29. Gardening for the visually impaired (1995). Ryde, New South Wales, Australia: Horticultural Therapy Society of NSW.
- P30. Griffin, G. (1976). Grow your own garden: Pedal power. College Station, TX: Energy Conservation Resources for Education, Dept. of Industrial Education, Texas A & M University. Includes bibliography (p.23).
- P31. Hadfield, K. (1981). Gardening for the handicapped: and the disenchanted. Harwood, Bolton, England: Tell Tale.
- P32. Hamad, M. M. (1970). Report to the government of the Sudan on a school gardening and nutrition project. Rome, Italy: Food and Agriculture Organization of the United Nations.
- P33. Hammerman, S., & Duncan, B. (Eds.). (1975). Barrier-free design: Report of a United Nations expert group meeting on barrier-free design, held June 3-8, 1974, at the United Nations Secretariat, New York. New York, NY: Rehabilitation International.

- P34. Handelsman, J. F. (1993). Gardens from garbage: How to grow indoor plants from recycled kitchen scraps. Brookfield, CT: Milbrook Press. Includes bibliographical references (p. 46) and index.
- P35. Haywood, J. A. (1997). Hastings heritage permaculture garden information book: Creating community by building gardens. Rosebud, Victoria, Australia: Jo Anne Haywood.
- P36. Heymann, M. (1978). Places and spaces: Environmental psychology in education. Bloomington, IN: Phi Delta Kappa Educational Foundation. Includes bibliography (p. 44-46).
- P37. Hiott, J.A. (1974). Hortitherapy in a recreational program for the mentally retarded. Clemson, SC: Horticulture Department, South Carolina Agricultural Experiment Station, Clemson University. . . Includes bibliographical references (last unnumbered leaf).
- P38. Horticultural Therapy Association of Victoria. (1996). Introduction to raised garden beds. Glen Waverly, Victoria, Australia: Horticultural Therapy Association of Victoria, Inc. . . Includes bibliographical references.
- P39. Howlett, S. (1993). Children's play areas: Help yourselfl: How to use the neighborhood matching fund to build community. Seattle, WA: Dept. of Neighborhoods.
- P40. Hughes, C. (1997). Urban agriculture and sustainable urban development [microform]: A case of study of Nairobi, Kenya. Vancouver, British Columbia, Canada: University of British Columbia, Centre for Human Settlements; University of British Columbia, School of Community & Regional Planning. Includes bibliography.
- P41. International Society for Rehabilitation of the Disabled. Committee on Technical Aids, Housing and Transportation. (1968). Gardening for the disabled: Some hints and suggestions.Bromma, Sweden: International Society for Rehabilitation of the Disabled (ISRD). Committee on Technical Aids, Housing and Transportation (ICTA).
- P42. Jensen, N. (1996). Allotment guide: Copenhagen & surroundings [Kolonihave guide]. (L. Haugmark, Trans.). Copenhagen: De Frie Fugle Forlag. Includes bibliographical references (p. [48]).
- P43. Jewell, M. & Powell, J. (1962). Gardening hints for people with arthritis. Corvallis, OR: Oregon State University. Extension Service. Includes bibliographical references (p.2).
- P44. Jonas, G.J. (1955). Handbook on horticultural therapy. Hastings, MI:

- Printed by Hastings Banner.
- P45. Jorgensen, J. (1980). Modification of park and recreation facilities for handicapped individuals. Washington, DC: Distributed by Hawkins and Associates. Includes bibliography (p. 40.)
- P46. Kingman, A.R., Exzell, D.O., Jones, E.V. & Olson, A.P. (1992). Outdoor gardening for the handcapped. Clemson, SC: Cooperative Extension Service, Clemson University. Includes bibliographical references (p.8).
- P47. Kirk, L. (1975). Accent on access. McLean, VA: American Society of Landscape Architects Foundation.
- P48. Larson, J., Hanchek, A. & Vollmar, P. (1996). Accessible gardening for therapeutic horticulture. St. Paul, MN: Minnesota Extension Service, University of Minnesota, College of Agriculture, Food, and Environmental Sciences.
- P49. Macalister-Smith, E.F. (1982). Horticulture and gardening: Important resources for the occupational therapist in rehabilitation programmes. Frome, Somerset, England: Horticultural Therapy. Includes bibliographical references (p. 6-7).
- P50. McNeilan, R.A. & Powell, J. (1992). Master gardeners promote therapy through horticulture. Corvallis, OR: Oregon State University Extension Service. Includes bibliographical references (p. 2).
- P51. Miller, P. L. (1970). School gardens and farms: Aspects of outdoor education. Las Cruces, NM: Educational Resources Information Center (ERIC), Clearinghouse on Rural Education and Small Schools (CRESS), New Mexico State University.
- P52. Mollison, B. C. (1984). Permaculture community. E. Bevor, (Ed.). Worcester, MA: Yankee Permaculture. "Edited from transcript of the Permaculture Design Course, The Rural Education Center, Wilton, NH, 1981."
- P53. Mollison, B. C. (1985). Permaculture for urban areas and urban-rural links. Worcester, MA: Yankee Permaculture. "Edited from transcript of the Permaculture Design Course, The Rural Education Center, Wilton, NH, USA, 1981."
- P54. National Council for Therapy and Rehabilitation Through Horticulture Conference (8th: August 26-29, 1980: Boys Town, Neb.). (1980). Eighties: A decade of discovery, development and dedication. T. K.

- Titze, & G. S. Liddel, (Eds.). Mount Vernon, VA: The Council.
- P55. National Council for Therapy and Rehabilitation through Horticulture. Horticultural Therapy Research Project, Coorsh, R. P., & Snoddy, G. D. (1978). Funding sources for horticultural therapy programs. Mount Vernon, VA: The Council.
- P56. National Council for Therapy and Rehabilitation through Horticulture. Horticultural Therapy Research Project, Snoddy, G. D., & Relf, P. D. (1978). Organizing a horticultural therapy workshop. Mount Vernon, VA: The Council.
- P57. National Council for Therapy and Rehabilitation through Horticulture. Michigan Chapter. (1970-1979?). Horticultural therapy activities. [s.l.]: NCTRH, Michigan Chapter.
- P58. National Council for Therapy and Rehabilitation through Horticulture. Michigan Chapter. (1979, 1989). Annotated bibliography of the Michigan Chapter NCTRH Library. [s.l.]: NCTRH, Michigan Chapter.
- P59. New Jersey. Dept. of Education. Bureau of Facility Planning Services. (1977). Greenhouses for schools. Trenton, NJ: The Bureau. Includes bibliographical references (p. 24).
- P60. Newberg, J., Eberbach, C., New York (State) Office of Mental Health, Weston United Community Renewal Inc., & Cornell Cooperative Extension. (1991). Urban gardens, New York, N.Y. [microform]. New York, NY: Bruner Foundation.
- P61. Nicholls, K., & Miller, D. (1980). Grow your own: A handbook for backyard and community gardeners. K. Nicholls, (Ed.). Ann Arbor, MI: Project Grow Community Gardens. Includes bibliographical references.
- P62. Ocone, L., & Thabault, G. (Eds.). (1984). Tools & techniques for easier gardening. Burlington, VT: Gardens for All.
- P63. Parsons, H. G. (1917). City gardens. New York, NY: Division of Intelligence and Publicity of Columbia University.
- P64. Pascalides, E. (1978).Hortitherapy: A comprehensive approach to greenhouse and training systems. Clemson, SC: Horticutlure Department. South Carolina Agricultural Experiment Station, Clemson University. Includes bibliographical references.
- P65. Peoples Housing Inc. (1985). Accessibility design guidelines for public facilities serving children. Topanga, CA: Peoples Center for Housing

Change.

- P66. Planning for a gardening programme. (1994). Ryde, New South Wales, Australia: Horticultural Therapy Society of NSW.
- P67. Privette, C.V. (1970-1979?). Greenhouse designs for the handicapped. Clemson, SC: Cooperative Extension Service. Clemson University
- P68. Project WEY: Washington Environmental Yard! (1972). Berkeley, CA: Washington Elementary/University of California Laboratory School.
- P69. Prudencio Bohrt, J. (1998). Urban agriculture in community gardens. Rotterdam: Institute for Housing and Urban Development Studies.
- P70. Raised garden beds (19--). Ryde, New South Wales, Australia: Horticultural Therapy Society of NSW.
- P71. Relf, D. (1994). Gardening in raised beds and containers for individuals with physical disabilities. Blacksburg, VA: Virginia Cooperative Extension.
- P72. Relf, P. D., Cummings, W. J., & National Council for Therapy and Rehabilitation Through Horticulture. Horticultural Therapy Research Project. (1978). Annotated bibliography of horticultural therapy. Mt. Vernon, VA: National Council for Therapy and Rehabilitation Through Horticulture.
- P73. Richmond, K. C. (1920). Rural school playgrounds and equipment. Washington, DC: Govt. Print. Off.
- P74. Rogers, P. & Powell, J. (1992). Gardening with limited range of motion. Corvallis, OR; Oregon State University Extension Service. Includes bibliographical references (p.2).
- P75. Savill, T. A. (1996). Trials on platform edge tactile surfaces. Crowthorne, Berkshire, UK: Transport Research Laboratory. Includes bibliographical references (p. 38-39).
- P76. Simmons, D.R. (1996). Horticultural opportunities for people with physical disabilities: A study of horticultural fiacilities in Australia (ACT, South Australia, N.S.W.). Wellington, New Zealand: Winston Churchill Memorial Trust.
- P77. Slemp, P.A., & Stoutamire, K.T. (Eds.). (1980). Introduction to hortitherapy, Sunland Training Center, Marianna, Florida. Marianna, FL: Sunland Training Center.

- P78. Smit, J. (1996). Urban agriculture, progress and prospect: 1975-2005. Ottawa, Canada: IDRC.
- P79. Smith, C. O. (1917). Garden clubs in the schools of Englewood, New Jersey. Washington, DC: G.P.O. Includes bibliographical references.
- P80. Spastics Society. (1980). Thorngrove. Gillingham, Dorset, England: The Spastics Society.
- P81. Texas state park facilities accessible to & usable by the handicapped. (1986). Austin, TX: Texas Parks and Wildlife Dept.
- P82. Texas state park facilities accessible to & usable by the handicapped. (1989). Austin, TX: Texas Parks and Wildlife Dept.
- P83. Texas state park facilities accessible to and usable by the handicapped. (1983). Austin, TX: Texas Parks and Wildlife.
- P84. Thoday, P.R. (1977). Table-top planter for the handicapped: Planter handbook. Bath, England: University of Bath.
- P85. Throckmorton, E., & Powell, J. (1992a). Adapting garden tools to overcome physical challenges. Corvallis, OR: Oregon State University Extension Service. Includes bibliographical references (p. 2).
- P86. Throckmorton, E.. & Powell, J. (1992b). Gardening adaptations for people with gripping and lifting problems. Corvallis, OR: Oregon State University Extension Service. Includes bibliographical references (p. 2).
- P87. Towler, J. K. (19--). Horticulture and the patient in an acute care general hospital: A manual designed for staff persons volunteers. East Lansing, MI: Michigan State University Printing.
- P88. Underhill, C. (1981). Horticulture and gardening in therapy, leisure and employment for disabled and handicapped people. Frome, Somerset, England: Horticultural Therapy. Includes bibliographical references (p. 23-25).
- P89. University of British Columbia. Botanical Garden. (1984). Gardening as therapy: A resource manual for development of horticulture therapy programs for the fall season. Vancouver, British Columbia, Canada: Botanical Garden, UBC.
- P90. University of Connecticut. Bartlett Arboretum. (1977). Horticultural therapy handbook. Stamford, CT: University of Connecticut. College of Agriculture and Natural Resources. Bartlett Arboretum. Includes

- bibliographical references (p. 17).
- P91. Urban agriculture: An abbreviated list of references and resource guide (1997). Beltsville, MD: USDA, ARS, National Agricultural Library.
- P92. Virginia Polytechnic Institute and State University. College of Architecture. (1975). Design for the disabled. Blacksburg, VA: The College.
- P93. Virginia Polytechnic Institute and State University. College of Architecture. (1977). Toward a barrier-free environment. Blacksburg, VA: The College.
- P94. Wals, H. (1982). Town farms: New approaches in the use of active educational methods in the environment field. Strasbourg; Croton, NY: Council of Europe.
- P95. Walter, F. (19??). Raised flower beds: Plans produced for the Central Council for the Disabled. London, England: Disabled Living Activities Group.
- P96. White, A. S. (1981). Gardening for leisure: A series of seminars-workshops. Canberra, Australia: Royal Australian Institute of Parks and Recreation.
- P97. Williams Fork River boardwalk: Challenge cost share project for the physically challenged (1990). Washington, DC: The Service.
- P98. Winslow, A. G., & Virginia Polytechnic Institute and State University. College of Architecture and Urban Studies. Division of Architecture and Environmental Design. (1978). Towards barrier-free environments: Criteria and policies for site design. Blacksburg, VA: Division of Architecture and Environmental Design, College of Architecture and Urban Studies, Virginia Polytechnic Institute and State University.
- P99. Witt, P. A., & Compton, D. M. (1980-1983?). Manual for utilization of the "Assessment of environmental barriers to leisure experiences" instrument. Denton, TX: Division of Recreation and Leisure Studies, North Texas State University. Includes bibliography (leaf 18).

3 CURRICULUM MATERIALS

- C1. Agyeman, J. (1995). *People, plants and places*. Crediton, Devon, United Kingdom: Southgate.
- C2. Backyards and butterflies: Ways to include children with disabilities in outdoor activities (1995). Cambridge, MA: Brookline Book.
- C3. Bales, S. F. (1996). Ready, set, grow!: A guide to gardening with children. New York, NY: Macmillan USA.
- C4. Barnes, T., Haas, J. K., Madama, J., & Wicklein, J. (1982). Ladybugs and lettuce leaves: Teachers' manual S. Hackman & J. Madama (Eds.). Washington, D. C.: Center for Science in the Public Interest. Supplement to the text Ladybugs and lettuce leaves, originally produced by Project Outside/Inside, Somerville Public Schools, Somerville, Massachusetts.
- C5. Barrett, K., & Willard, C. (1998). *Schoolyard ecology: Teacher's guide*. Berkeley, CA: Lawrence Hall of Science, University of California.
- C6. Beaty, S., & Fountas, I. (1993). Butterflies abound!: A whole language resource guide for K-4. Menlo Park, CA: Addison-Wesley.
- C7. Beller, J. (1985). Experimenting with plants: Projects for home, garden, and classroom. New York, NY: Arco. Includes index and bibliography (p. 149).
- C8. Big book of nature projects (1997). New York, NY: Thames and Hudson.
- C9. Bonnet, R. L., & Keen, G. D. (1989). Botany: 49 science fair projects. New York, NY: Tab Books.
- C10. Bonnet, R. L., & Keen, G. D. (1991). Botany: 49 more science fair projects.

- Blue Ridge Summit, PA: TAB Books.
- C11. Bradley, C. (1996). Step-by-step 50 gardening projects for kids. New York, NY: Smithmark.
- C12. Bremner, E., & Pusey, J. (1990). *Children's gardens: A field guide for teachers, parents and volunteers* (2nd ed.). Los Angeles, CA: University of California Cooperative Extension, Common Ground Garden Program.
- C13. Bremner, E., & Pusey, J. (1999). *Children's gardens: A field guide for teachers, parents and volunteers* Y. Savio (Ed.), (3rd ed.). Oakland, CA: University of California Cooperative Extension, Common Ground Garden Program. Includes bibliographical references (p. 181-186g)
- C14. Brennan, G., Brennan, E., Barchechat, M., & East Bay French-American School. (1997). *Children's kitchen garden: A book of gardening, cooking and learning.* Berkeley, CA: Ten Speed Press. Includes bibliographical references (p. 141) and index
- C15. Brown, S. E. (1981). Bubbles, rainbows & worms: Science experiments for preschool children. Mt. Rainier, MD: Gryphon House.
- C16. Byles, M. (1994). Experiment with plants. Minneapolis, MN: Lerner Publications.
- C17. Caduto, M. J., & Bruchac, J. (1994). Keepers of life: Discovering plants through Native American stories and earth activities for children. Golden, CO: Fulcrum Pub.
- C18. Caduto, M. J., & Bruchac, J. (1996). *Native American gardening: Stories, projects, and recipes for families.* Golden, CO: Fulcrum Pub.
- C19. Carrick, J. (1979). *Planting, growing, caring*. New Brunswick, NJ: Vocational-Technical Curriculum Laboratory, Rutgers--The State University. "Prepared as a text for deaf students involved in an Ornamental Horticulture program"--p. i.
- C20. Clark, R., & Walters, P. (1992). Trees in the school grounds: Learning through landscapes. Crediton, Devon, United Kingdom: Southgate.
- C21. Cockram, M., Rakow, D., & Kline, R. (1991). Garden in the city: A 4-H leader's guide for urban gardening. New York, NY: Cornell Cooperative Extension.
- C22. Cohen, J., & Pranis, E. (1990). *GrowLab: Activities for growing minds*. Burlington, VT: National Gardening Association.

- C23. Colella, O. (1994). *Naturaleza*. Buenos Aires, Argentina: Sigmar. Original title: My first look at nature.
- C24. Cooking with the three sisters (199-). Ithaca, NY: Cornell University.
- C25. Czufin, L. F. (Ed.). (1974). *Child's garden: A guide for parents and teachers* (Fifth printing, revised & expand ed.). San Francisco, CA: Chevron Chemical Co., Ortho Division.
- C26. Dennee, J., Peduzzi, J., & Hand, J. (1996). *In the three sisters garden*. Dubuque, IA: Kendall/Hunt Pub. Co.
- C27. Diehn, G., & Krautwurst, T. (1995). Kid style nature crafts: 50 terrific things to make with nature's materials. New York, NY: Sterling Pub.
- C28. Dietl, U. (1995). [Spirebogen til små grønne fingre. English] Plant-and-grow project book (1st paperback ed.). New York, NY: Sterling Pub. Co.
- C29. Donald, K. B. (1991). *Planting seeds, growing minds: A horticultural K-6 curriculum*. Sacramento, CA: California Association of Nurserymen. Includes bibliographical references (p.29).
- C30. Feltwell, J. (1991). Beekeeping: A practical guide to beekeeping in the school grounds. Winchester, England: Learning Through Landscapes Trust.
- C31. Feltwell, J., & Learning Through Landscapes Trust. (1991). *Slugs, snails and earthworms*. Winchester, England: Learning Through Landscapes.
- C32. Forte, I., Pangle, M. A., & Tupa, R. (1975). *Pumpkins pinwheels and peppermint packages* (Student ed.). Nashville,TN: Incentive Publications.
- C33. Gale, F. C., & Gale, C. W. (1975). Experiences with plants for young children. Palo Alto, CA: Pacific Books.
- C34. Gamboa de Vitelleschi, S. (1991). *Aprender jugando con la naturaleza* (4th ed.). Maipu, Argentina: Editorial Bonum.
- C35. Gardening science manual (1971). Brooklyn, NY: Board of Education of the City of New York. Reprinted 1990.
- C36. Getting started: A guide for creating school gardens as outdoor classrooms (1997). Berkeley, CA; Santa Cruz, CA: Center for Ecoliteracy; Life Lab Science Program.
- C37. Gjersvik, M. (1997). *Green fun: Plants as play.* Willowdale, Ontario, Canada: Firefly Books.

- C38. Gjersvik, M. (1974). Green fun. Riverside, CT: Chatham Press.
- C39. Guy, L. A., Cromell, C., Bradley, L. K., & Maricopa County Master Gardener School Gardening Task Force. (1996). Success with school gardens: How to create a learning oasis in the desert. Phoenix, AZ: Arizona Master Gardener Press in cooperation with the University of Arizona, Maricopa County Cooperative Extension.
- C40. Hamilton, L. (1998). *Child's play in nature*. New York, NY: Berkeley Pub. Group.
- C41. Haplern, R. R. (1993). Green planet rescue: Saving the earth's endangered plants. New York, NY: F. Watts. Includes bibliographical references (p. 62) and index.
- C42. Harrison, P. (1995). Grasses the staff of life: An integrated curriculum designed for grades 1 & 2, grades 3 & 4 F.M. Norris, (Ed.). Fort Worth, TX: Botanical Research Institute of Texas.
- C43. Harrison, P. (1995). *Pumpkins more than just a jack-o-lantern: An integrated curriculum designed for grades 1 4* F.M. Norris, (Ed.). Fort Worth, TX: Botanical Research Institute of Texas.
- C44. Harrison, P. (1996). Beans, peas and bluebonnets: An integrated curriculum for elementary learners F. Norris, (Ed.). Fort Worth, TX: Botanical Research Institute of Texas. Includes bibliographical references.
- C45. Harrison, P. (1996). Cattails and horsetails-- life in a wet environment: An integrated curriculum for elementary learners F.M. Norris, (Ed.). Fort Worth, TX: Botanical Research Institute of Texas.
- C46. Harrison, P. (1996). Looking at leaves: An integrated curriculum for elementary learners F.M. Norris, (Ed.). Fort Worth, TX: Botanical Research Institute of Texas.
- C47. Hemenway, H. D. (1903). How to make school gardens: A manual for teachers and pupils. New York, NY: Doubleday, Page.
- C48. Herd, M. (1997). Learn and play in the garden (1st for the United States and Canada ed.). Hauppauge, NY: Barron's.
- C49. Hogan, K. (1994). Eco-inquiry: A guide to ecological learning experiences for the upper elementary/middle grades. Dubuque, IA: Kendal/Hunt.
- C50. Hunken, J. Madama, J., & Pacelli. P. (1982). Ladybugs and lettuce leaves (Rev. ed.). Washington, D. C.: Center for Science in the Public Interest.

- Prepared by Project Inside/Outside, Somerville Public Schools.
- C51. Ingram, M. (1993). Bottle biology: An idea book for exploring the world through plastic bottles and other recyclable materials. Dubuque, IA: Kendall/Hunt Pub. Co.
- C52. Jaffe, R., Appel, G., & Life Lab Science Program. (1990). Growing classroom: Garden-based science. Menlo Park, CA: Addison-Wesley Pub. Co.
- C53. Jaffe, R., & Cadoux, M. (1982). Growing classroom: A garden-based science and nutrition curriculum for 2nd through 6th grades. Santa Cruz, CA: Friends of the Harvest.
- C54. Junior master gardener handbook: Level 1 (1999). College Station, TX: Texas Agricultural Extension Service, Texas A&M University System.
- C55. Junior master gardener teacher/leader guide: Level 1 (1999). College Station, TX: Texas Agricultural Extension Service, Texas A&M University System.
- C56. Jurenka, N. E. A., & Blass, R. J. (1996). Beyond the bean seed: Gardening activities for grades K-6. Englewood, CO: Teacher Ideas Press.
- C57. Jurenka, N. E. A., & Blass, R. J. (1996). *Cultivating a child's imagination through gardening*. Englewood, CO: Teacher Ideas Press.
- C58. Katz, A. (1986). *Naturewatch: Exploring nature with your children*. Reading, MA: Addison-Wesley Pub. Co.
- C59. Kepler, L. (1996). Windowsill science centers. New York, NY: Scholastic Professional Books.
- C60. Kite, L. P. (1995). Gardening wizardry for kids. Hauppauge, NY: Barron's.
- C61. Liem, T. L. (1992). Turning kids on to science in the home. Chino Hills, CA: Science Inquiry Enterprises.
- C62. Lingelbach J. (Ed.), . Hands-on nature: Information and activities for exploring the environment with children (1986). Woodstock, VT: Vermont Institute of Natural Science.
- C63. Lovejoy, S. (1991). Sunflower houses: Garden discoveries for children of all ages. Loveland, CO: Interweave Press. Includes bibliographical references (p. 138-139) and index.
- C64. Lovejoy, S. (1994). *Hollyhock days: Garden adventures for the young at heart.* Loveland, CO: Interweave Press.

- C65. Lovejoy, S. (1999). Roots, shoots, buckets & boots: Gardening together with children. New York, NY: Workman Pub.
- C66. MacGarry, A. (1994). *Green technology: By teachers, for teachers* D. Randle, (Ed.). Machynlleth, Powys, United Kingdom: Green Teacher Resources.
- C67. Mackey, L. (1994). Math in the garden. Monterey, CA: Evan-Moor.
- C68. Milne, L. (1996). Nature crafts: Have fun with nature!: More than 80 great projects. Westport, CT: Reader's Digest Young Families.
- C69. Milord, S. (1989). Kids' nature book: 365 indoor/outdoor activities and experiences. Charlotte, VT: Williamson Pub. Co.
- C70. Mitchell, C., & Miller, N. (1977). Keep it growing: activities manual designed for classroom use by teachers of severely mentally impaired, trainable mentally & developmental learning students ... in Wayne County. Wayne, MI: Wayne County Intermediate School District.
- C71. Molden, S. (1999). Growth through nature: A preschool program for children with disabilities H. Loomis, (Ed.). Sagaponak, NY: Sagapress. Includes bibliographical references (p. 112-113)
- C72. Moore, R. C., & Wong, H. H. (1997). Natural learning: The life history of an environmental schoolyard: Creating environments for rediscovering nature's way of teaching. Berkeley, CA: MIG Communications.
- C73. My first look at nature (1991). (1st American edition). New York, NY: Random House.
- C74. Ocone, L. (1983). Youth gardening book: A complete guide for teachers, parents, and youth leaders. Burlington, VT: Gardens for All, Inc.
- C75. Ocone, L. (1990). National Gardening Association guide to kids' gardening. New York, NY: Wiley.
- C76. Outdoor classroom: Educational use, landscape design & management of school grounds (1990). London: H.M.S.O.
- C77. Outdoor classroom: Experiencing nature in the elementary curriculum (1989). Indianapolis, IN: Indiana Dept. of Education.
- C78. Perkins School for the Blind. (1993). *Horticulture program*. Watertown, MA: The School.
- C79. Petrash, C. (1992). Earthways: Simple environmental activities for young children. Mt. Rainier, MD: Gryphon House.

- C80. Please, P. (1991). *Children's gardening: A month by month guide advancing educational gardening activities in schools.* Frome, United Kingdom: Horticultural Therapy.
- C81. Please, P. (1993). *Children's gardening*. Beech Hill, Reading, England: Thrive.
- C82. Pranis, E., & Hale, J. (1988). *GrowLab: A complete guide to gardening in the classroom.* Burlington, VT: National Gardening Association. Includes 1 videocassette + 1 guide
- C83. Rhoades, D. (1995). Garden crafts for kids: 50 great reasons to get your hands dirty. New York, NY: Sterling Pub. Co.
- C84. Richardson, B. (1998). Gardening with children. Newtown, CT: Taunton Press.
- C85. Rosenberg, M. S., & Edmond-Rosenberg, I. (1994). *Special education sourcebook: A teacher's guide to programs, materials, and information sources.* Rockville, MD: Woodbine House.
- C86. Roth, C. E. (1988). Beyond the classroom: Exploration of schoolground and backyard. Lincoln, MA: Massachusetts Audubon Society.
- C87. Rowe, S. (1991). *Seasons in the school grounds*. Winchester, England: Learning Through Landscapes.
- C88. Russell, H. R. (1973). Ten-minute field trips: Using the school grounds for environmental studies, a teacher's guide. Chicago, IL: J.G. Ferguson Pub. Co.
- C89. Russell, H. R. (1990). Ten-minute field trips: A teacher's guide to using the school grounds for environmental studies (2nd ed.). Washington, D. C.: National Science Teachers Association. Includes bibliographical references and index.
- C90. Russell, H. R. (1998). Ten-minute field trips: A teacher's guide to using the school grounds for environmental studies (3rd ed.). Arlington, VA: National Science Teachers Association. Includes bibliographical references and index.
- C91. Russo, M. (1993). Tree almanac: A year-round activity guide. New York, NY: Sterling. Includes index.
- C92. Salisbury, J. E. (1988). Green classroom program: A teacher's guide to a garden-based science curriculum. East Falmouth, MA: New Alchemy Institute. Includes bibliography (p. 164-166).

- C93. Sanera, M. (1999). Facts, not fear: Teaching children about the environment (Rev. ed.). Washington, D. C.: Regnery Pub. Includes bibliographical references and index.
- C94. Sanera, M., & Shaw, J. S. (1996). Facts, not fear: A parent's guide to teaching children about the environment. Washington, D. C.; Lanham, MD: Regnery Pub.; Distributed to the trade by National Book Network.
- C95. Schecter, D. (1997). Science art: Projects and activities that teach science concepts and develop process skills. New York, NY: Scholastic Professional Books.
- C96. School garden: A module on growing foods: A teacher's guide (1979). (Rev. ed.). Honolulu, HI: Curriculum Research and Development Group, University of Hawaii.
- C97. Scott, E., & Duffy, C. (1998). Dinner from dirt: Ten meals kids can grow and cook (1st ed.). Salt Lake City, UT: Gibbs Smith.
- C98. Sheehan, K., & Waidner, M. (1991). Earth child: Games, stories, activities, experiments & ideas about living lightly on planet earth C. Haralson, (Ed.). Tulsa, OK: Council Oak Books.
- C99. Sheehan, K., & Waidner, M. (1994). Earth child: Games, stories, activities, experiments & ideas about living lightly on planet earth C. Haralson, (Ed.), (Rev. and updated ed.). Tulsa, OK: Council Oak Books.
- C100. Sheehan, K., & Waidner, M. (1998). Earth child 2000: Earth science for young children: Games, stories, activities, and experiments (New, with teacher's guide ed.). Tulsa: Council Oak Books. Rev. ed. of Earth child. 1994.
- C101. Stoneham, J. (1996). Grounds for sharing: A guide to developing special school sites. Winchester, Hampshire, United Kingdom: Learning through Landscapes.
- C102. Taylor, B. (1992). Green thumbs up!: The science of growing plants (1st Random House ed.). New York, NY: Random House. Includes index.
- C103. Tilgner, L. (1988). Let's grow!: 72 gardening adventures with children. Pownal, VT: Storey Communications. Includes index.
- C104. Van Hage, M. A. (1996). *Little green thumbs*. Brookfield, CT: Millbrook Press. Curriculum Materials
- C105. VanCleave, J. P. (1997). Janice VanCleave's plants: Mind-boggling experiments you can turn into science fair projects. New York, NY: John Wiley.

- C106. VanCleave, J. P. (1997). Janice VanCleave's plants [electronic resource]: Mind-boggling experiments you can turn into science fair projects. New York, NY: John Wiley. Includes index.
- C107. Wood, R. W. (1991). Science for kids: 39 easy plant biology experiments. Blue Ridge Summit, PA: Tab Books. Includes index.
- C108. Young, K. (1990). Learning through landscapes: Using school grounds as an educational resource. Winchester, Hampshire, United Kingdom: Learning Through Landscapes Trust.
- C109. Zhorne, W. L. (1996). *Nature crafts for children*. Grand Rapids, MI: Baker Books.

4 THESES AND DISSERTATIONS

- D1. Ackley, D. R. (1985). Effect of a horticultural therapy program on cerebral palsied children. (Unpublished master's thesis). Oregon State University, Corvallis, OR. Includes bibliographical references (leaves 59-63).
- D2. Alexander, J. (1987). Design standards for play spaces for children with cerebral palsy. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 89-90).
- D3. Aquino Ramirez, G. B. (1995). Social and nutritional benefits of community gardens for Hispanic-Americans in New York City and Los Angeles. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 23-25).
- D4. Ashiabi, G. S. (1995). Food systems: Community Shared Agriculture, a means of empowerment and social vitality. (Unpublished master's thesis, Wilfrid Laurier University, Waterloo, Ontario, Canada). Includes bibliographical references (leaves 95-100).
- D5. Ashley, I. E. (1969). Analysis of opportunities for paraplegics in certain ornamental horticulture occupations. (Unpublished doctoral dissertation). University of Illinois at Urbana-Champaign, Champaign, IL. Includes bibliographical references (leaves 94-95).
- D6. Bachert, R. E. (1976). History and analysis of the school garden movement in America, 1890-1910. (Unpublished doctoral dissertation). Indiana University, Bloomington, IN. Includes bibliographical references (leaves 181-189).
- D7. Barker, S. L. (1992). Meaning of a youth gardening program: A naturalistic

- *inquiry*. (Unpublished doctoral dissertation). Indiana University, Bloomington, IN. Includes bibliographical references (leaves 180-182).
- D8. Barnes, M. (1994). Study of the process of emotional healing in outdoor spaces and the concomitant landscape design implications. (Unpublished master's thesis). University of California, Berkeley, Berkeley, CA. Includes bibliographical references (leaves 102-107).
- D9. Barnes, S. G. (1973). Comparison of group versus individual attitudes and behaviors in a community gardening project. (Unpublished master's thesis). Florida State University, Tallahassee, FL. Includes bibliographical references (leaves 137-141).
- D10. Bassett, T. J. (1979). Vacant lot cultivation: Community gardening in America, 1893-1978. (Unpublished master's thesis). University of California, Berkeley, Berkeley, CA. Includes bibliographical references (leaves 192-199).
- D11. Bayer, L. M. (1986). *Community gardening center for Salt Lake City*. (Unpublished master's thesis). University of Utah, Salt Lake City, UT. Includes bibliographical references (leaves [164]-173).
- D12. Becker, T. (1996). Sensory explorations in architecture through building: The role of intuitive knowledge in design. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [65]-67).
- D13. Beechel, J. (1974). *Interpretation for handicapped persons*. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves 56-60).
- D14. Berge, B. A. (1993). Landscape preferences and stress responses of ethnically diverse adolescents. (Unpublished master's thesis). Washington State University, Putnam, WA. Includes bibliographical references (leaves 106-115).
- D15. Berry, J. S. (1975). *Hortitherapy program for substance abusers*. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D16. Bopp, L. A. (1985). Analysis of subject matter content in gardening and nutrition for elementary school students. (Unpublished master's thesis). University of Arizona, Tucson, AZ. Includes bibliographical references (leaves 71-72).
- D17. Brende, B. B. (1981). Survey of horticultural therapy in U. S. psychiatric hospitals. (Unpublished master's thesis). California State University, Hayward, Hayward, CA. Includes bibliographical references (leaves 75-83).

- D18. Brennan, D. A. (1978). *Horticultural therapy activities for the rehabilitation of physically disabled children.* (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographies.
- D19. Bushorr, L. B. (1993). Effects of purposeful activity on pulse rate and perceived exertion in a well elderly population. (Unpublished master's thesis). D'Youville College, Buffalo, NY. Includes bibliographical references (leaves 48-51).
- D20. Butterfield, D. W. (1982). Outdoor spaces surrounding group homes for the developmentally disabled adult. (Unpublished master's thesis). University of Illinois at Urbana-Champaign, Champaign, IL. Includes bibliographical references (leaves 106-110).
- D21. Ching, J. H. (1988). Historical and contemporary perspectives on the use of horticulture in occupational therapy. (Unpublished master's thesis). University of Southern California, Los Angeles, CA. Includes bibliographical references (leaves 181-190).
- D22. Chronister, G. L. (1993). Effect of horticulture therapy and animal-assisted therapy on seriously emotionally disturbed elementary students in a public school setting: An ethnographic study. (Unpublished doctoral dissertation). Oklahoma State University, Stillwater, OK. Includes bibliographical references (leaves 217-221).
- D23. Chu, J. (1998). Social and environmental restoration through therapeutic community gardens. (Unpublished master's thesis). San Jose State University, San Jose, CA. Includes bibliographical references (leaves 133-137)
- D24. Clark, G. A. (1980). Selected factors influencing the success of a community garden. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 6).
- D25. Clark, G. A. (1993). Effect of horticulture therapy and animal-assisted therapy on seriously emotionally disturbed elementary students in a public school setting: An ethnographic study. (Unpublished doctoral dissertation). Oklahoma State University, Stillwater, OK. Includes bibliographical references (leaves 217-221).
- D26. Conner-Ogorzaly, M. (1996). *Qualitative study of an innovative program to support children's learning of botanical conceptions*. (Unpublished doctoral dissertation). University of Texas at Austin, Austin, TX. Includes bibliographical references (leaves 327-340).
- D27. Corlett, J. L. (1999). Landscapes and lifescapes: Three generations of Hmong women and their gardens. (Unpublished doctoral dissertation). University of California, Davis. Includes bibliographical references (leaves 324-360).

- D28. Crandall, R. L. (1975). *Increasing the effectiveness of horticultural therapy activities in a geriatric setting.* (Unpublished master's thesis). Kansas State College, Manhattan, KS.
- D29. Daubert, J. R. (1981). Use of horticulture therapy at a public garden, park district, or county extension agency. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 9).
- D30. DeKay, M. (1992). *Daylighting and community gardening land use design tools*. (Unpublished master's thesis). University of Oregon, Eugene, OR. Includes bibliographical references (leaves 296-307).
- D31. DeMarco, L. W. (1997). The factors affecting elementary school teachers' integration of school gardening into the curriculum. (Unpublished doctoral dissertation). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 193-200).
- D32. Denny, B. P. (1992). Benefits of horticultural vocational training programs for individuals with autism. (Unpublished master's thesis). University of Toledo, Toledo, OH. Includes bibliographical references (leaves 72-86).
- D33. Doutt, K. M. (1990). Development of a manual for special educators for the purpose of teaching science through the use of horticultural activities: A thesis presented to the faculty of the Graduate School, Tennessee Technological University. (Unpublished master's thesis). Tennessee Technological University, Cookeville, TN. Includes bibliographical references (leaves 80-83).
- D34. Doxon, L. E. (1985). Measurement of physiological stress by personality type of developmentally disabled adults in horticultural training. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 37-38).
- D35. Duesing, S. (1997). *Interdisciplinary learning using farming and gardening is sound pedagogy*. (Unpublished master's thesis). Southern Connecticut State University, New Haven, CT. Includes bibliographical references (leaves 153-158).
- D36. Farmer, C. S. (1977). Patient/staff behavioral responses to flower arrangements within a private psychiatric hospital dining room. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 14).
- D37. Ferguson, M. (1989). *Therapeutic horticulture for the disabled*. (Unpublished research report, Charles Sturt University Mitchell, Bathurst, New South Wales, Australia). Included bibliographical references (leaves 129-132).

- D38. Fliegel, J. C. (1988). Some influences of horticultural and social activities on blood pressure of elderly people. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliography (leaf 44)
- D39. Floyd, J. A. (1972). Investigation into the physical and psychological response of the visually handicapped to some selected woody and herbaceous plant material. (Unpublished master's thesis). Clemson University, Clemson, SC. Includes bibliographical references (leaf 18).
- D40. Foncham, S. A. (1998). Attitudes toward and interest in community gardening in two low-income neighborhoods. (Unpublished doctoral dissertation). Texas Tech University, Lubbock, TX. Includes bibliographical references (leaves 88-90).
- D41. Francis, J. (1999). Gardening in inner Melbourne: Reaping personal, community, and environmental benefits. (Unpublished master's thesis, Monash University, Melbourne, Victoria, Australia). Includes bibliographical references.
- D42. Franco, M. (1983). Vegetable production in reclamated urban landfill sites. (Unpublished master's thesis). Rutgers University, Newark, NJ. Includes bibliographical references (leaves 27-30).
- D43. Gallagher, M. J. (1984). *Measuring arthritic hand skill performance during horticultural activities*. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 24).
- D44. Ghose, S. (1999). Healing dimensions of hospital gardens: Three case studies assessing the use, therapeutic benefits and design recommendations. (Unpublished master's thesis). University of Texas at Arlington, Arlington, TX. Includes bibliographical references (leaves 88-94).
- D45. Gignoux, L. C. (1987). Landscape design preferences of older people. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 144-145).
- D46. Gilkey, A. F. M. (1997). Trauma and the garden: Inquiry into the presence of response differences among traumatized and non-traumatized viewers to garden design influenced by aspects of trauma. (Unpublished master's thesis). University of Texas at Arlington, Arlington, TX. Includes bibliographical references (leaves 97-101).
- D47. Gilmour, D. J. (1982). Garden manual for instructing developmentally disabled adults. (Unpublished master's thesis). California State University, Sacramento, Sacramento, CA.

- D48. Gilreath, P. R. (1976). *Hortitherapy program for the visually handicapped*. (Unpublished master's thesis). Clemson University, Clemson, SC. Includes bibliographical references.
- D49. Glasner, A. G. (1987). Community gardening in San Francisco. (Unpublished master's thesis). San Francisco State University, San Francisco, CA. Includes bibliographical references (leaves 104-110).
- D50. Glasscock, M. T. (1981). Effect of horticultural therapy intervention on the self-concept of emotionally handicapped children. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D51. Godwin, L. d. (1994). Contributions of community gardening to participating households: A case study from Tempe, Arizona. (Unpublished master's thesis). Arizona State University, Tempe, AZ. Includes bibliographical references (leaves [65]-70).
- D52. Grant, C. F. (1994). Healing garden: Incorporating garden experience in hospitals and other health care facilities. (Unpublished master's thesis). University of Georgia, Athens, GA. Includes 8 fold-outs. Includes bibliographical references (leaves 121-128).
- D53. Gridley, M. M. (1997). Gardens for teaching: A case study in integrating behavioral factors in the design of outdoor educational environments for children. (Unpublished master's thesis). State University of New York, Syracuse, Syracuse, NY. Includes bibliographical references (leaves 81-86).
- D54. Griffin, M. R. (1998). Community vegetable gardening: A means to reduce cancer risk and related medical expenses in Trenton, New Jersey. (Unpublished bachelor's thesis). Rutgers University, Newark, NJ.
- D55. Habib, D. L. (1996). New spirit rising among us: Urban youth environmental activists. (Unpublished doctoral dissertation). University of Massachusetts at Amherst, Amherst, MA. Includes bibliographical references (leaves 196-202)
- D56. Hall, D. J. (1996). Community gardens as an urban planning issue. (Unpublished master's thesis, University of British Columbia, Vancouver, British Columbia, Canada). Includes bibliographical references (leaves 80-83)
- D57. Haller, R. L. (1978). Comparisons of horticultural and non-horticultural activities in the rehabilitation process of mildly retarded adults. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf vi).

- D58. Hashisaki, K. S. (1988). Experience of place: Significant childhood and adult environments. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [133]-142).
- D59. Hassan Loyan, B. N. (1990). Horticultural experience increases garden size and economic investment within a community garden program. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliography (leaves 7-8).
- D60. Hassan Loyan, B. N. (1995). Educational models for community garden programs in the United States and their potential application for sub-Saharan Africa. (Unpublished doctoral dissertation). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 68-72).
- D61. Hawkins, K. (1991). Therapeutic landscape: Nature, architecture, and mind in nineteenth-century America. (Unpublished doctoral dissertation). University of Rochester, Rochester, NY. Includes bibliographical references (leaves 354-374).
- D62. Herbach, G. J. (1998). Community gardening organizations' relationship to the public planning function: A story of shared goals. (Unpublished master's thesis). University of Wisconsin Madison, Madison, WI. Includes bibliographical references (leaves 82-86).
- D63. Hill, C. O. (1980). Resident acceptance of outdoor gardening in two geriatric institutions. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 46-47).
- D64. Hillenbrand-Nowicki, C. (1993). Effect of plants on human perceptions and behavior within an interior atrium. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 88-91).
- D65. Hiott, J. A. (1975). *Hortitherapy program for the mentally handicapped*. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D66. Hogan, D. (1986). Study to determine the feasibility of placing trained mentally retarded individuals into horticultural jobs in the Philadelphia area. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliography (leaf 27).
- D67. Hoge, G. W. (1990). Attitudes, subjective norms, perceived behavioral control, intentions, and behavior toward providing special recreation. (Unpublished doctoral dissertation). Indiana University, Bloomington, IN. Includes bibliographical references (leaves 91-96).

- D68. Horne, D. C. (1974). Evaluation of the effectiveness of horticultural therapy on the life satisfaction level of aged persons confined to a rest care facility. (Unpublished master's thesis). Clemson University, Clemson, SC. Includes bibliographical references (leaves [84]-88).
- D69. Hugo, A. P. (1996). Horticultural therapy with adults with serious and persistent mental illness: Implications for social work. (Unpublished master's thesis). College of St. Catherine; College of St. Thomas, St. Paul, MN. This paper was written as part of a joint MSW program of the College of St. Catherine and the University of St. Thomas. Includes bibliographical references (leaves 61-63).
- D70. Hunt, M. E. (1974). Relationship of knowledge of the urban physical environment and satisfaction with the urban environment. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D71. Hunter, N. L. (1971). Horticulture programs in prisons. (Unpublished bachelor's thesis). California Polytechnic State University, San Louis Obispo, CA.
- D72. Ikagawa, T. (1994). Residential gardens in urban Honolulu, Hawaii: Neighborhood, ethnicity, and ornamental plants. (Unpublished doctoral dissertation). University of Hawaii, Honolulu, HI. Includes bibliographical references (leaves 189-198).
- D73. Jedelsky, B. (1989). Freiraumplanung fur geistig-, korperlich-, sinnes- und mehrfach behinderte menschen. (Unpublished doctoral dissertation, Universitaet fuer Bodenkulture Wien, Wien, Austria).
- D74. Jewell, J. R. (1906). *Agricultural education, including nature study and school gardens*. (Unpublished doctoral dissertation). Clark University, Worcester, MA. Includes bibliographical references and index.
- D75. Kaplan, S. A. (1999). *Horticultural therapy at Casa Colina: A program for the disabled.* (Unpublished master's thesis). Texas A&M University, College Station, TX. Includes bibliographical references (leaves 35-39)
- D76. Kaufman, D. D. (1982). Assessment of program success in horticultural therapy for special populations. (Unpublished master's thesis). Southern Illinois University at Carbondale, Carbondale, IL. Includes bibliographical references (leaves 44-45).
- D77. Kelley, A. C. (1995). Evaluating potential for intergenerational gardening programs established by Master Gardeners of Johnson County, Kansas. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references.

- D78. Kim, E. (1999). Human psychophysiological and self-rated emotional responses to geranium visual stimuli during recovery from stress. (Unpublished doctoral dissertation). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 44-48).
- D79. Kovary, M. M. (1999). Healing landscapes: Design guidelines for mental health facilities. (Unpublished master's thesis). Cornell University, Ithaca, NY. Includes bibliographical references.
- D80. Krell, S. L. (1984). Acquisition of horticultural work skills by mentally retarded adults using social reinforcement. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 18-19).
- D81. Laviana, J. E. (1985). Assessing the impact of plants in the simulated office environment: A human factors approach (horticultural therapy, interior plantscaping, design). (Unpublished doctoral dissertation). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 66-72).
- D82. Leavitt, C. (1976). Community gardening: its origins in the reform movements of the late Nineteenth century and its current practice in Northampton, Massachusetts. (Unpublished master's thesis). University of Massachusetts, Amherst, MA.
- D83. Leger, J. G. (1997). Use of horticulture as an adjunct to psychotherapeutic treatment: An exploratory study: A project based upon an independent investigation. (Unpublished master's thesis). Smith College School for Social Work, Northhampton, MA. Includes bibliographical references (leaves 82-87).
- D84. Leiker, N. M. (1975). Considerations for the development of horticultural therapy diagnostic evaluations within a psychiatric setting. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D85. Leonard, J. R. (1997). Growing community through community gardens: Guidelines for using community gardens as a tool for building community. (Unpublished master's thesis). Ball State University, Muncie, IN. Includes bibliographical references (leaves 82-85).
- D86. Lin, H. (1995). Science discourse in one American kindergarten classroom. (Unpublished doctoral dissertation). University of Illinois at Urbana-Champaign, Champaign, IL. Includes bibliographical references (leaves 178-193).
- D87. Lindholm, G. (1995). Skolgarden: Vuxnas bilder; barnets miljo. (Unpublished Agr. dr., Sveriges Lantbruksuniversitet, Alnarp, Sweden).
- D88. Long, C. M. (1999). Experiencing a place, envisioning a landscape, empowering

- a people: The story of developing a community venture with the Siksika Nation Youth Entrepreneurial Development Society. (Master's thesis, University of Manitoba, Winnipeg, Manitoba, Canada). Includes bibliographical references (leaves 223-227).
- D89. MacGilvra, G. E. (1997). Value of community gardens to the people of their communities. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [111]-114).
- D90. MacQueen, M. S. (1996). *Challenge: Designing meaningful experiences with QuickTime VR*. (Unpublished master's thesis). Michigan State University, East Lansing, MI. Includes bibliographical references (leaves 53-54).
- D91. Mandeville, M. (1974). Some of the psychological aspects of establishing a horticultural therapy and rehabilitation program for use in a women's prison. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D92. Martin, J. L. (1994). *Plants; a practical prescription: Horticultural therapy; a partial remedy for social, medical, and educational difficulties.* (Unpublished master's thesis). West Texas A & M University, Canyon TX. Includes bibliographical references (leaves 91-98).
- D93. May, P. A. (1977). Horticulture and the senior citizen: An exploration into horticultural therapy. (Unpublished master's thesis). Cornell University, Ithaca, NY. Includes bibliographical references (leaves 83-86).
- D94. McAndrew, W. P. (1980). Analysis of horticultural therapy activities in licensed nursing homes. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D95. McDowell, M. J. C. (1997). Role and application of horticultural therapy with institutionalized older people. (Unpublished master's thesis, McGill University, Montreal, Quebec, Canada). Includes bibliographical references (leaves 106-113).
- D96. McGuinn, C. P. (1999). Assessment of a vocational horticulture training curriculum at a school for juvenile offenders. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 66-69).
- D97. McInerney, G. K. (1958). History of the School Garden Association of New York with emphasis upon its contributions to school activities which originate from school garden work. (Unpublished doctoral dissertation). Columbia University, New York, NY. Includes bibliographical references (leaves 88-91).
- D98. McQuaid, S. (1981). "Can we plant milk?": Community gardening with

- urban children. (Unpublished master's thesis). Bank Street College of Education, New York, NY.
- D99. Merkle, J. (1993). Perceived benefits and economic aspects associated with community gardening. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes sample letters and questionnaires. Includes bibliographical references (leaves 37-38).
- D100. Miller, R. (1984). *Children's use of the outdoor environment*. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [81]-89).
- D101. Mitchell, L. S. (1967). *Empirical study of urban recreation units: Playgrounds as central places*. (Unpublished doctoral dissertation). Ohio State University, Columbus, OH. Includes bibliographical references (leaves 213-219).
- D102. Monroe-Santos, S. L. (1998). *Improving permanency for urban community gardens*. (Unpublished master's thesis). University of California, Davis, Davis, CA.
- D103. Moore, S. C. (1985). Self-reporting scale for indicating emotional states of developmentally disabled adult horticultural workers. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 34-38).
- D104. Moros, N. (1984). *Horticulture for the handicapped*. (Unpublished doctoral dissertation, Auckland Teachers College, Auckland, New Zealand). Includes bibliographical references (leaf 37).
- D105. Mugo, S. W. (1993). Citizens + vacant lots=community open space: A case study of the Union Settlement Community Garden, East Harlem, New York City. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 66-68).
- D106. Murphy, M. L. (1977). Some effects of flower/nonflower table centerpieces on psychiatric patient and staff behavior. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 14-15).
- D107. Murphy, P. M. (1982). Development and evaluation of domain referenced items for professionals in horticultural therapy. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D108. Myer, I. A. (1983). Assessing relationships between horticultural knowledge and mental well-being of female college students. (Unpublished master's thesis).

- Kansas State University, Manhattan, KS. Includes bibliography (leaves 24-26).
- D109. Noah, M. (1990). Benefits and educational needs in home gardening activities as perceived by the rural elderly in Tucker County, West Virginia. (Unpublished master's thesis). West Virginia University, Morgantown, WV. Includes bibliographical references (leaves 34-35).
- D110. Noyes, S. J. (1978). Effects of living and non-living horticultural materials on academic achievement and conduct of moderately retarded institutionalized individuals. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D111. O'Connor, P. F. (1979). Horticulture as a resource alternative in vocational education and rehabilitation: Its effects on various personality and behavior factors in adolescents with special needs. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D112. Ogorzaly, M. C. (1996). *Qualitative study of an innovative program to support children's learning of botanical conceptions*. (Unpublished doctoral dissertation). University of Texas at Austin, Austin, TX. Includes bibliographical references (leaves 327-340).
- D113. Ojeda-Kimbrough, C. A. (1996). *Investing in gardens: A critical assessment of community gardening programs in Los Angeles*. (Unpublished master's thesis). University of California, Los Angeles, Los Angeles, CA. Includes bibliographical references (leaves 32-33)
- D114. Olson, A. P. (1976). Development and implementation of an evaluated study of the effect of horticultural therapy on certain physically disabled patients. (Unpublished master's thesis). Clemson University, Clemson, SC.
- D115. O'Neill, L. N. J. (1985). *Horticulture therapy: Reestablishing a lost domain of occupational therapy*. (Unpublished master's thesis). University of North Carolina at Chapel Hill, Chapel Hill, NC. Includes bibliographical references (leaves [34]-35).
- D116. O'Reilly, J. A. (1999). *Community gardeners: Planting the seeds of a sustainable future.* (Unpublished master's thesis, Wilfrid Laurier University (Canada), Waterloo, Ontario, Canada). Includes bibliographical references (leaves 148-153).
- D117. Owen, P. J. *Influence of a botanical garden experience on human health.* (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 37-43).
- D118. Parrett, L. A. (1994). Analysis of horticultural therapy programs within

- Kansas long- term adult care facilities. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes a sample letter and a questionnaire. Includes bibliographical references (leaves 44-47).
- D119. Parsons, A. B. (1985). Designing a resource guide for horticultural therapy programs at botanical gardens and arboreta. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Includes bibliographical references (leaves 21-23).
- D120. Peck, S. J. (1978). Adult education in botanic gardens: environmental awareness through horticulture. (Unpublished master's thesis). University of Delaware, Newark, DE. Includes bibliographical references (leaves 103-109).
- D121. Pigg, S. M. C. (1992). Community gardeners in Lansing, Michigan: Education, motivation and their relation to feelings about pest control. (Unpublished master's thesis). Michigan State University, East Lansing, MI. Includes bibliographical references (leaves 129-135).
- D122. Pitt, E. J. (1984). Factors which can affect the preferences of the elderly for hand pruners. (Unpublished master's thesis). Virginia Polytechnic Institute and State University. Blacksburg, VA. Includes bibliographical references (leaves 42-45).
- D123. Plankington, H. S. (1973). *Horticulture as a work program for therapy*. (Unpublished master's thesis). University of Delaware, Newark, DE.
- D124. Predny, M. L. (1999). Assessing an intergenerational horticulture therapy program for elderly adults and preschool children [electronic resource]. (Unpublished master's thesis). Virginia Polytechnic Institute and State University, Blacksburg, VA. Retrieved from institutional database http://scholar.lib.vt.edu/theses/available/etd-041499-
- 172746/unrestricted/mpthesis.pdf; Includes bibliographical references (leaves 7-11, 21, 37, 52-54).
- D125. Priest, M. J. (1984). Assessing adult developmentally disabled client ability to discern horticultural crop quality. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D126. Pugh, M. P. (1976). *Comparative study: Two training programs for mentally retarded adults.* (Unpublished master's thesis). Sam Houston State University, Huntsville, TX. Includes bibliographical references (leaf 72).
- D127. Relf, D. (1976). Horticulture in a semi-self supporting training center for mentally retarded adults. (Unpublished doctoral dissertation). University of Maryland, College Park, MD.

- D128. Rice, J. S. (1993). *Self development and horticultural therapy in a jail setting*. (Unpublished doctoral dissertation). Professional School of Psychology, Sacramento, CA. Includes bibliographical references (leaves 186-219).
- D129. Rockaway, R. J. L. (1994). Garden designs for the elderly: A guide to designing successful outdoor landscapes. (Unpublished master's thesis). University of Georgia, Athens, GA. Includes five foldouts and four transparencies. Includes bibliographical references (leaves 80-86).
- D130. Rod, B. (1994). Community gardening in the Old North End neighborhood of Burlington, Vermont: An interest and needs assessment survey. (Unpublished bachelor's thesis). University of Vermont, Burlington, VT. Includes bibliographical references (leaves. 89-95)
- D131. Rodriguez, B. (1989). Soil condition and the productivity of two community garden plots in the city of Milwaukee, Wisconsin during the 1986 gardening season. (Unpublished master's thesis). University of Wisconsin Milwaukee, Milwaukee, WI. Includes bibliographical references.
- D132. Rogers, N. B. (1995). *Modern commons: Place, nature, and revolution at the Strathcona Community Gardens.* (Unpublished master's thesis, Simon Fraser University, Burnaby, British Columbia, Canada).
- D133. Sachs, N. A. (1999). *Therapeutic value of outdoor space in psychiatric healthcare facilities*. (Unpublished master's thesis). University of California, Berkeley, Berkeley, CA. Includes bibliographical references (leaves 96-103).
- D134. Salamy, V. (1995). *Healing gardens: Design guidelines for landscape architects.* (Unpublished master's thesis). Ohio State University, Columbus, OH. Includes bibliographical references (leaves 256-272).
- D135. Samyn, D. A. (1998). Therapeutic garden for depression: A design exploration of environmentally-influenced therapeutic effects. (Unpublished master's thesis). University of Illinois at Urbana-Champaign, Champaign, IL. Includes bibliographical references (leaves 81-85).
- D136. Sanyal, B. (1984). *Urban agriculture: A strategy of survival in Zambia*. (Unpublished doctoral dissertation).). University of California, Los Angeles, Los Angeles, CA. Includes bibliographical references (leaves 268-280).
- D137. Schrader, L. A. (1996). Relationships between community gardeners needs, horticultural species preferences and organic gardening methods. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes a sample letter and questionnaire. Includes bibliographical references (leaves 30-32).
- D138. Seyfried, S. A. (1982). Children and horticulture: Construction and

- evaluation of illustrated horticultural teaching aids. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliography (leaves 19-20).
- D139. Shafer, L. (1998). *Benefits of a community garden: Implications for a sustainable society.* (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [105]-110).
- D140. Sharpless, M. R. (1993). Fertile ground, narrow choices: Women on cotton farms of the Texas Blackland Prairie, 1900-1940. (Unpublished doctoral dissertation).). Emory University, Atlanta, GA. Includes bibliographical references (leaves [432]-462).
- D141. Sheffield, B. K. (1992). Affective and cognitive effects of an interdisciplinary garden-based curriculum on underachieving elementary students. (Unpublished doctoral dissertation). University of South Carolina, Columbia, SC.
- D142. Shoemaker, C. A. (1982). *Modeling as a prevocational horticultural training method with trainable mentally retarded adults.* (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 5-6).
- D143. Shoemaker, J. (1982). Horticultural skill ratings, demographic characteristics, economics, economic aspects, and social behaviors of community gardeners. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 24).
- D144. Sinclair, J. A. (1980). Gardening for mobility impaired individuals: Design standards, plants and tools. (Unpublished master's thesis). University of Massachusetts, Amherst, MA.
- D145. Smith, D. H. (1998). Cultivating the margins: A teaching garden and community building at Interbay. (Unpublished master's thesis). University of Washington, Seattle, WA. Includes bibliographical references (leaves [58]-59)
- D146. Spencer, D. (1995). *Adult education at botanic gardens: Planning environmental education*. (Unpublished doctoral dissertation). Cornell University, Ithaca, NY. Includes bibliographical references.
- D147. Stebbins, H. P. (1996). Comparison of community gardening efforts in Austin, New York City and San Francisco. (Unpublished master's thesis). University of Montana, Missoula, MT. Includes bibliographical references (leaves 75-77).
- D148. Stephens, C. A. (1998). Environmental education: A vehicle for integrating

- community members of all abilities and discovering a sense of place. (Unpublished master's thesis). Prescott College, Prescott, AZ.
- D149. Stober, P. L. T. (1994). Survey of horticultural therapy professional issues. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D150. Train, R. L. (1974). Effect of horticultural therapy in maintaining the life satisfaction of geriatrics. (Unpublished master's thesis). Kansas State University, Manhattan, KS.
- D151. Tull, D. (1995). Urban gardening as a tool for community empowerment: A comparative study of urban gardening programs for youth in Los Angeles and San Francisco. (Unpublished bachelor's thesis). Hampshire College, Amherst, MA. Includes bibliographical references (leaves 106-111).
- D152. Voorhees, D. D. (1997). *Use of horticulture by occupational therapists in Minnesota.* (Unpublished master's thesis). College of St. Catherine, St. Paul, MN. Includes bibliographical references.
- D153. Waidelich, W. D. (1995). Core occupational competencies for secondary agricultural education programs as identified by Ohio agricultural business and industry. (Unpublished doctoral dissertation). Virginia Polytechnic Institute and State University, Blacksburg, VA.
- D154. Waldemayer, C. R. (1992). Surveying greenhouse industry attitudes toward employability of people with mental disabilities. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references.
- D155. Waliczek, T. M. (1994). Community gardening and its influence on the perceptions of quality of life of community gardeners. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaves 35-37).
- D156. Waliczek, T. M. (1997). Effect of school gardens on self-esteem, interpersonal relationships, attitude toward school, and environmental attitude in populations of children: A dissertation. (Unpublished doctoral dissertation). Texas A&M University, College Station, TX. Includes bibliographical references (leaves 123-134).
- D157. Wells, J. A. (1995). *Study of people's attitudes toward plants*. (Unpublished doctoral dissertation). University of Missouri Columbia, Columbia, MO. Includes bibliographical references (leaves 211-213).
- D158. Westphal, L. M. (1999). Growing power? Social benefits of urban greening projects. (Unpublished doctoral dissertation). University of Illinois at

- Chicago, Chicago, IL. Includes bibliographical references (leaves 198-207).
- D159. Williams, P. N. (1987). Horticultural activities and demographic factors influence children's self-esteem. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliography (leaves 24-25).
- D160. Williams, S. (1992). Seeds of change: Community gardening in South Holyoke, Massachusetts. (Unpublished master's thesis). Hampshire College, Amherst, MA. Includes bibliographical references.
- D161. Wilson, S. L. (1995). *Visits to public gardens: Its meaning for avid gardeners.* (Unpublished doctoral dissertation). University of Tennessee, Knoxville, Knoxville, TN.
- D162. Wotowiec, P. J. (1975). Students' and parents' ratings of the tract garden program in the Cleveland public schools. (Unpublished doctoral dissertation). Ohio State University, Columbus, OH. Includes bibliographical references (leaves 142-144).
- D163. Wu, W. (1991). Study of the worker role in people with schizophrenia participating in a horticulture program. (Unpublished master's thesis). University of Southern California, Los Angeles, CA. Includes bibliographical references (leaves 74-79).
- D164. Young, T. G. (1991). *Nature and moral order: The cultural significance of San Francisco's parks*, 1865-1925. (Unpublished doctoral dissertation). University of California, Los Angeles, Los Angeles, CA. Includes bibliographical references (leaves 251-276).
- D165. Zareh, E. E. (1998). *Native American mythology and Jungian psychology as metaphors in horticultural therapy*. (Unpublished master's thesis). Kansas State University, Manhattan, KS. Includes bibliographical references (leaf 32).

5 MEDIA

- M1. Ackerman, D. (1995). *Natural history of the senses*. [Audio cassette]. Prince Frederick, MD: Recorded Books
- M2. American Community Gardening Association & Insight Productions (Producers). (1997). *Growing community from the roots up* [VHS]. Philadelphia, PA: Insight Productions. Also available on DVD.
- M3. Arcane Light Productions (Producer). (1993). Homeless garden project. [VHS]. Santa Cruz, CA: Arcane Light Productions.
- M4. Bennell, P. (Director). (1986). My urban garden. [VHS] New York, NY: National Film Board of Canada.
- M5. Bowman, A., Kainy, J., & Kutcher, N. (Producers). (1995). *Green dreams: Planting hope in urban neighborhoods*. [VHS]. Los Osos, CA: The Video Project.
- M6. Children's garden project. (199-).[VHS]. Durango, CO: VIDEO-11 distributor.
- M7. Cleveland Society for the Blind. (1961?). Gardening for blind persons [Record]. New York, NY: American Foundation for the Blind.
- M8. Clokey, J. (Producer), & Clokey, J. (Director). (1996). *Opportunity to grow*. [VHS]. Los Osos, CA: San Louis Video Publishing.
- M9. Clokey, J., & Harrigan, J. (Producers), & Clokey, J. (Director). (1994). Universal design in the landscape. [VHS]. Los Osos, CA: San Luis Video.
- M10. Community gardening. (1992). [VHS]. Houston, TX: Zimmerman &

- Associates. The New garden; show #213.
- M11. Conrad, K. J. and Sokkay, D. (Directors). (1993). *Horticultural therapy*. [VHS]. Gaithersburg, MD: American Horticultural Therapy Association.
- M12. Cooper, A. (Producer) & Cooper, A. (Director). (1988). *Coping after a stroke*. [VHS]. Caulfield, Victoria, Australia.
- M13. Fleet, K. (1980-1987?). *Gardening without sight* [Audio cassette]. London: Royal National Institute for the Blind.
- M14. Fleet, K. (1989?). *Gardening without sight* [Audio cassette]. Victoria Park, Western Australia, Australia.
- M15. Fleet, K. (1991). Havetips for blinde og svagtseende ... lyt, duft og fol dig frem [Audio cassette]. Tonder, Denmark INFOKO.
- M16. Funston, R. (1990). Garden programme for everyone including blind, visually impaired, and disabled people [Audio cassette]. Richards, P. (Ed.). Melbourne, Victoria, Australia: Royal Victorian Institute for the Blind.
- M17. Green thumb program: Long term care. (1980). [Audio cassette]. Carbondale, IL: Southern Illinois University.
- M18. Gröna mediciner: En växande tillgång = Green medicines: A growing asset. (1995). [VHS]. Princeton, NJ: Films for the Humanities & Sciences.
- M19. Hammond, L. (Producer). (1990s). Eden alternative: Building human habitats. [VHS]. Endwell, NY: JL Video Productions.
- M20. Handelsman, J. F. (1998). *Spiritual gardening]: [cultivating love through caring for plants].* [Audio cassette]. Boulder, CO: Sounds True.
- M21. Harrigan, J., & Clokey, J. (Producers), & Clokey, J. (Director). (1994). *Introduction to universal design*. [VHS]. Los Osos, CA: San Luis Video Pub.
- M22. Harrigan, J., & Clokey, J. (Producers), & Clokey, J. (Director). (1994). *Universal design in architecture.* [VHS]. Los Osos, CA: San Luis Video.
- M23. Hôpital Rivière-des-Prairies. Centre de communication en santé mentale, Amis du Jardin botanique de Montréal, & Douglas Hospital (Verdun, Québec). Newman Readaptation Centre (Producers), & Martin, R. (Director). (1995). Seasons of the mind horticulture and therapy. [VHS]. Montreal, Canada: CECOM, Centre de communication en santé mentale, Hôptial Rivière-des-Prairies.
- M24. Horowitz, J. (Producer). (1995). Greenhouse Project horticultural therapy.

- [VHS]. Fair Lawn, NJ: JHCC Visual Productions. Horticultural therapy advisor, Joel Flagler.
- M25. How to begin a school garden [Audio tape & slides]. Oakland, CA: ANR Communications. Includes script, audiotape, and 53 slides.
- M26. Howard, R., Borruso, L., Lawson, J. E., & Basehore, L. (Producers), & Howard, R. (Director). (1995). *Growing hope: The homeless garden project story*. [VHS]. Santa Cruz, CA: Homeless Garden Project.
- M27. Huse, D., Sears, J. R., University of Massachusetts Dartmouth. Center for Portuguese Studies and Culture and Massachusetts Cultural Council (Directors), (1997). *Urban cottage gardens of the Portuguese community*. [VHS]. New Bedford, MA: Spinner Publications.
- M28. Hynes, H. P. (Director). (1998). *Community gardens*. [VHS]. Long Beach, CA: California State University, Long Beach.
- M29. Latham Foundation (Producer). (1995). Garden therapy. [VHS]. Alameda, CA: The Latham Foundation.
- M30. Lerner, M. (Producer), & Lerner, M. (Director). (1984). *Designing environments for everyone*. [VHS]. Washington, DC: Landscape Architecture Foundation.
- M31. Levenson, G. (Director). (1997). *Pumpkin circle*. [VHS]. Santa Cruz, CA: Informed Democracy. Includes: Activities for students folded sheet. Also available on DVD in English and in Spanish.
- M32. Lifelong gardening for seniors. (1995). [VHS]. Menlo Park, CA: Mid-Peninsula Access Corp.
- M33. Minnesota Extension Service (Producer), & Weidner, M. (Director). (1993). *Horticultural therapy*. [VHS]. St. Paul, MN: Minnesota Extension Service.
- M34. My amazing garden. (1994). [VHS]. Minneapolis, MN: Metropolitan Banana Productions.
- M35. National Film Board of Canada (Producer), & Merritt, J. (Director). (1977). *Vacant lot.* [VHS]. Santa Rosa, CA: Arthur Mokin Productions.
- M36. National Gardening Association (Producer). (1991). Plant a question using the GrowLab teaching cycle. [VHS]. Burlington, VT: National Gardening Association.
- M37. National Treasures Productions (Producer). (1993). Elsa Bakalar

- portrait of a gardener. [VHS].Santa Rosa, CA: Arthur Mokin Productions.
- M38. Paper Tiger TV (Producer). (1990s). Play for keeps: The struggle to save our community gardens. [VHS]. New York, NY: Paper Tiger Television.
- M39. Pasquale, M., Randall, L., & Georgia Center for Continuing Education. (Producers). (1989). *Growing connection in therapy*. [VHS]. Athens, GA: University of Georgia.
- M40. Permanent raised bed gardening. (1997). [VHS]. Ardmore, OK: The Samuel Roberts Noble Foundation Inc.
- M41. Peter Vogt & Associates, & Vogt, P. S. (Producers), & Vogt, P. S. (Director). (1993). *Toward universal design*. [VHS]. Washington, DC: Universal Design Initiative, Assistive Technology Program, National Rehabilitation Hospital. Produced for the Rehabilitation Engineering Center on the Evaluation of Assistive Technology.
- M42. Photosynthesis Productions, & Cornell Waste Management Institute, Cornell University (Producers), & Gluck, D. H. (Director). (1996). *It's gotten rotten*. [VHS]. Oley, PA: Bullfrog Films. Teacher's guide by Nancy M. Trautmann and Marianne E. Krasny.
- M43. Poulsson, A., Armstrong, M., & Scott, M. (Producers) & Armstrong, M. (Director). (1990-1996?). *Everyone's business*. [VHS]. Montreal, Canada: National Film Board of Canada.
- M44. Raven's Hollow (Firm), & Airlie Foundation (Producers). (1983). Garden in the heart. [Motion Picture] Airlie, VA:
- M45. *School garden curriculum.* [Audio tape & slides]. Oakland, CA: ANR Communications. . Includes script, audiotape, and 74 slides.
- M46. Sheilds, A.J. (Producer). (1990). *Gardening for everyone*. [VHS]. Claremont, Western Australia, Australia: People & Plants Association.
- M47. Therapeutic gardens enriching the garden experience. (1995).[VHS]. Charlottesville, VA: University of Virginia; Medical Television Facility.
- M48. Therapeutic nature of gardening. (1994/1995). [VHS]. St. Paul, MN: Educational Development System, University of Minnesota.
- M49. Virginia Polytechnic Institute and State University. University Relations/Visual Communications. (Producer). (1991). Exploring people-plant interaction. part 1, the art of Rhonda Roland Shearer. [VHS]. Arlington, VA: University Relations-Visual Communications at Virginia Tech. From the

symposium: The Role of horticulture in human well-being and social development, April 19-21, 1990, Arlington, Virginia.

- M50. Virginia Polytechnic Institute and State University. University Relations/Visual Communications. (Producer). (1992). Role of horticulture in human well-being and social development, exploring people-plant interactions. part 2, the science and theory. [VHS]. Blacksburg, VA: Virginia Tech.
- M51. Way it is: Barrier-free design for the elderly and the disabled. (1975). [Audio cassette & slides]. Syracuse, NY: Syracuse University.
- M52. Wellington, N. (Producer), & Wellington, N. (Director). (1992). *Growing season.* [VHS]. Oley, PA: Bullfrog Films.

6 JOURNALS, PERIODICALS & SERIALS

Journals, periodicals & serials where a substantial portion of the content addresses one or more aspects of horticultural therapy

- S1. Advances in environment, behavior, and design. (1987-).
- S2. Community greening review. (1991). Vol. 1, no. 1 (1991)-.
- S3. Disabled Gardeners Association (Perth, Western Australia, Australia). (1981). *Newsletter.* (1981-1982).
- S4. Environmental Design Research Association. (1969). EDRA; proceedings of the annual environmental design research association conference. 1st-.
- S5. Great plains journal of horticultural therapy. (1979). Vol. 1, no. 1 (summer 1979)-.
- S6. Green teacher. (1991). Issue 22 (Feb. 1991)-.
- S7. Hortscience. (1966). Vol. 1- winter 1966-.
- S8. HortTechnology. (1991). Vol. 1, no. 1 (Oct./Dec. 1991)-.
- S9. HortTherapy. (1979). Vol. 1, no. 1- Mar. 1979-.
- S10. Journal of therapeutic horticulture. (1986). v. 1- .
- S11. People and plants: Magazine of the People and Plants Association of Australia. (1985-).

- S12. People and Plants Association of Australia. (1985). Newsletter. (1985-).
- S13. Society for Horticultural Therapy and Rural Training. (1979). *Growth point: The magazine of the Society for Horticultural Therapy and Rural Training Ltd.* (1979-).
- S14. Terra-peutic news: Magazine of the Disabled Gardeners Association of Australia, Inc. ([1982]-1984).

7

TITLES MOST-COMMONLY HELD

This could be called 'Core Titles'. The bibliography grew from an analysis of the collection at Kansas State University Libraries with collections at 4 other institutions (Rutgers University, Texas A&M University, VirginiaTech and University of Washington, Seattle) with programs in horticultural therapy or community gardening. By their very nature horticultural therapy and/or community gardening are interdisciplinary topics. So the 'core' list includes titles from other disciplines as well as titles specifically addressing horticultural therapy in all its aspects.

There are 2 lists below. The first list includes titles from the bibliography held by all 5 institutions. Since one of the 5 programs does not have a horticultural therapy program, the titles in List 1 does not include any titles dealing with horticultural therapy. It is List 2, where 4 of the 5 institutions checked had the titles in their collections, which includes titles devoted to horticultural therapy. Both lists are arranged alphabetically by the title. The Reference number (bolded) refers to the item number of the work within chapters 1-6..

List 1: Titles held by 5 institutions

Access by design. George A. Covington, 1997. 178

Accessible gardening for people with physical disabilities: a guide to methods, tools, and plants. Janeen R. Adil, 1994. 5

Advances in environment, behavior, and design [serial title, usually partial

holdings]. 1987. S1

Analytical behavioural geography. Reginald G. Golledge, 1987. 300

Applied ecological psychology for schools within communities: assessment and intervention. 1997.

Approach to environmental psychology. Albert Mehrabian, 1974. 514

Behavior and environment; the use of space by animals and men. Proceedings of an international symposium held at the 1968 meeting of the American Association for the Advancement of Science. International Symposium on the Use of Space by Animals and Men (1968: Dallas), 1971. **395**

Biophilia. Edward Osborne Wilson, 1984. 786

Biophilia hypothesis. 1993. 424

Biophilosophy. Bernhard Rensch, 1971. 614

Childhood's domain: play and place in child development. Robin C. Moore, 1986, 538

Children's engagement in the world: sociocultural perspectives. 1999. 303

Children's play spaces: from sandbox to adventure playground. Marguerite Rouard, 1977. **634**

Cognition and environment: functioning in an uncertain world. Stephen Kaplan, 1982. **420**

Cognition, social behavior, and the environment. 1981. 341

Complete playground book. Arlene Brett, 1993. 91

Creative outdoor play areas. Peggy L. Miller, 1972. 523

Culture and environment. Irwin Altman, 1980. 11

Design of children's play environments. Mitsuru Senda, 1992. 674

Designing for the disabled. [2d rev. and expanded ed.] Selwyn Goldsmith, 1967. 291

Designing sidewalks and trails for access [microform]. Julie B. Kirschbaum, 1999. 432

Enabling garden: a guide to lifelong gardening. Gene Rothert, 1994. 633

Environment and behavior: a dynamic perspective. Charles J. Holahan, 1978. **365**

Environment for aging: interpersonal, social, and spatial contexts. Russell A. Ward, 1988. **766**

Environmental Design Research Association EDRA: proceedings of the annual Environmental Design Research Association Conference [serial]. 1969. **\$4**

Environmental psychology: directions and perspectives. 1983. 234

Environmental psychology: man and his physical setting. 1970. 599

Experience of landscape. Jay Appleton, 1975. 27

Experience of nature: a psychological perspective. Rachel Kaplan, 1989. 416

Experience of place. Tony Hiss, 1990. 362

Experimental psychology of sensory behavior. John F. Corso, 1967. 176

Fundamentals of sensory physiology, 3rd rev. & exp. ed. 1986. 666

Garden blocks for urban America. Louise Carter Bush-Brown, 1969. 114

Geographies of disability. Brendan Gleeson, 1999. 286

Green history: a reader in environmental literature, philosophy, and politics. Derek Wall, 1994. **760**

Green nature/human nature: the meaning of plants in our lives. Charles A. Lewis, 1996. 474

Healing landscape: therapeutic outdoor environments. Martha M. Tyson, 1998. **738**

Horticultural therapy and the older adult population. 1997. 773

Horticulture as therapy: principles and practice. 1998. 688

HortScience [serial]. 1966. S7

Human behavior and environment: advances in theory and research. 1976.

Human relationship with nature: development and culture. Peter H. Kahn, 1999. 414

Human senses and perception. G. M. Wyburn, 1964. 796

Human spaces: life-enhancing designs for healing, working, and living. Barbara Crisp, 1998. **185**

Humanscape: environments for people. 1978. 418

Idea of wilderness: from prehistory to the age of ecology. Max Oelschlaeger, 1991. **570**

In search of nature. Edward Osborne Wilson, 1996. 787

Landscapes for learning: creating outdoor environments for children and youth. Sharon Stine, 1997. **707**

Landscapes of the mind: worlds of sense and metaphor. J. Douglas Porteous, 1990. **594**

Meaning of gardens: idea, place, and action. 1990. 253

Natural affairs: a botanist looks at the attachments between plants and people. Peter Bernhardt, 1993. **73**

Natural history of the senses. Diane Ackerman, 1990. 4

New organic grower: a master's manual of tools and techniques for the home and market gardener. Eliot Coleman, 1989. 157

On nature, 1984, **635**

Passing strange and wonderful: aesthetics, nature, and culture. Yi-fu Tuan, 1993. **736**

People-plant relationships: setting research priorities. 1994. 238

Perception and the senses. Evan L. Brown, 1979. 100

Perspectives on environment and behavior: theory, research, and applications. 1977. **708**

Philosophy, science, and sense perception: historical and critical studies. Maurice Mandelbaum, 1964. **492**

Place and placelessness. E. C. Relph, 1976. 612

Places for people: Architects design hotels, motels, restaurants, bars, clubs, community recreation facilities, camps, parks, plazas, playgrounds. 1976. **191**

Placeways: a theory of the human environment. E. V. Walter, 1988. 765

Plants for people. Anna Lewington, 1990. 473

Plants, people, and environment. 1979. 421

Play for all guidelines: planning, design, and management of outdoor play settings for all children. 1992. **541**

Playgrounds: the planning, design and construction of play environments Peter Heseltine, 1987. **353**

Power of place: how our surroundings shape our thoughts, emotions, and actions. Winifred Gallagher, 1994. **261**

Power of place: urban landscapes as public history. Dolores Hayden, 1995. **344**

Psychology of place. David V. Canter, 1977. 118

Public places and private spaces: the psychology of work, play, and living environments. Albert Mehrabian, 1976. **513**

Restorative gardens: the healing landscape. Nancy Gerlach-Spriggs, 1998. 274

Role of horticulture in human well-being and social development: a national symposium, 19-21 April 1990, Arlington, Virginia. 1992. **609**

School and home gardens. William Herman Dietrich Meier, 1913. 515

Secret life of plants. Peter Tompkins, 1973. 731

Space and spirit in modern Japan. Barrie B. Greenbie, 1988. 307

Toward a new philosophy of biology: observations of an evolutionist. Ernst Mayr, 1988. **503**

Trees of life: essays in philosophy of biology. 1992. 316

Understanding of nature: essays in the philosophy of biology. Marjorie Glicksman Grene, 1974. **315**

Universal access to outdoor recreation: a design guide. 1993. 744

With people in mind: design and management of everyday nature . Rachel Kaplan, 1998. **419**

Yard, street, park: the design of suburban open space. Cynthia L. Girling, 1994. 285

List 2: Titles held by 4 institutions

Able to garden: a practical guide for disabled and elderly gardeners. 1990. **590**

Accessible landscapes: designing for inclusion. Philip S. Evans, 1993. 227

Aging and milieu: environmental perspectives on growing old. 1983. 636

Agriculture in the city's countryside. C. R. Bryant, 1992. 102

Backyard market gardening: the entrepreneur's guide to selling what you grow. Andrew W. Lee, 1993. 468

Barrier-free exterior design: anyone can go anywhere. 1985. 623

Behavior and the natural environment. 1983. 19

Behavior settings: a revision and extension of Roger G. Barker's Ecological psychology. Phil Schoggen, 1989. **667**

Biophilosophy: analytic and holistic perspectives. R. Sattler, 1986. 651

Body, mind, and the sensory gateways. Felix Deutsch, 1963. 202

Brain, environment, and social psychology. J. K. Chadwick-Jones, 1979. **133**

Bridge of life: from matter to spirit. Edmund Ware Sinnott, 1966. 689

Building without barriers for the disabled. Sarah P. Harkness, 1976. 336

Children and gardens. Gertrude Jekyll, 1982. 401

Children and the environment, 1978. 18

Children on playgrounds: research perspectives and applications. 1993. 339

Children's play and playgrounds. Joe L. Frost, 1979. 258

Children's play in diverse cultures. 1994. **628**

City gardens: an open spaces survey in the City of London. Brian Plummer, 1992. 591

City of farmers: informal urban agriculture in the open spaces of Nairobi, Kenya. Donald B. Freeman, 1991. **256**

Complete book of community gardening. Jamie Jobb, 1979. 406

Complete book of patio gardening. Jack Kramer, 1970. 436

Contributions of the physical, biological, and psychological sciences in human disability. 1958. **165**

Creating community in the city: cooperatives and community gardens in Washington, D.C. Ruth H. Landman, 1993. 449

Creative playgrounds and recreation centers. Alfred Ledermann, 1959. 464

Designing for the disabled. Selwyn Goldsmith, 1976. 294

Designing for the handicapped: the mentally retarded, the mentally ill, the maladjusted, the blind, the deaf, those with learning difficulties, the gifted or exceptional child. 1971. 57

Disability and the environment. Vida Carver, 1978. 127

Ecobehavioral analysis and developmental disabilities: the twenty-first century. 1990. **668**

Ecological psychology: creating a more earth-friendly human nature. George S. Howard, 1997. **383**

Ecopsychology: restoring the earth, healing the mind. 1995. 631

Edible city resource manual. Richard Britz, 1981. 96

Elderly people and the environment. 1984. 14

Enabling environments: measuring the impact of environment on disability and rehabilitation. 1999. **704**

Environment and aging. M. Powell Lawton, 1980. 460

Environment and behavior: an international and multidisciplinary bibliography, 1970-1981. Lenelis Kruse, 1984. **446**

Environmental context of aging: life-styles, environmental quality, and living arrangements. 1979. **116**

Environmental interaction: psychological approaches to our physical surroundings. David V. Canter, 1976. **120**

Environmental psychology. Paul A. Bell, 1978. 68

Environmental psychology. Jeffrey D. Fisher, 1984. 236

Environmental psychology. Norman W. Heimstra, 1978. 346

Environmental psychology: people and their physical settings. 1976. 600

Environmental psychology. Norman W. Heimstra, 1974. 345

Experience of landscape, rev. ed. Jay Appleton, 1996. 28

Experiencing the environment. 1976. **766**

Experimental sensory psychology. 1975. 655

Feng shui in the garden: simple solutions for creating comforting, life-affirming gardens of the soul. Nancilee Wydra, 1997. **797**

Fundamentals of sensory physiology. 1978. 665

Gaia, a new look at life on earth. J. E. Lovelock, 1979. 483

Gardening on Main Street. Buckner Hollingsworth, 1968. 367

Gardens in healthcare facilities: uses, therapeutic benefits, and design recommendations. Clare Cooper Marcus, 1995. **166**

Green history of the world. Clive Ponting, 1991. **593**

Green medicine; the search for plants that heal. Margaret Kreig, 1964. 444

Grow it indoors; a practical, personal how-to and why guide to growing successful houseplants. Richard W. Langer, 1975. **454**

Growing with gardening: a twelve-month guide for therapy, recreation, and education. Bibby Moore, 1989. **537**

Handbook of community gardening. 1982. 548

Healing dimensions of people-plant relations: proceedings of a research symposium: March 24-27, 1994, University of California, Davis. 1994. **255**

Healing gardens: therapeutic benefits and design recommendations. 1999. **168**

Horticulture as therapy: a practical guide to using horticulture as a therapeutic tool. Mitchell L. Hewson, 1994. **355**

Inquiry by design: tools for environment-behavior research. John Zeisel,

1984. **805**

International landscape design: architecture of gardens, parks, playgrounds & open spaces. Margaret Cottom-Winslow, 1991. 177

Interpretation for handicapped persons: a handbook for outdoor recreation personnel. Jacque Marlene Beechel, 1975. **65**

Landscape as inspiration. Hans Dieter Schaal, 1994. 652

Landscape design for elderly & disabled people. Jane Stoneham, 1994. 710

Life worth living: how someone you love can still enjoy life in a nursing home: the Eden alternative in action. William H. Thomas, 1996. **726**

Man and nature; philosophical issues in biology. Ronald Munson, 1971. 545

MetroFarm: the guide to growing for big profit on a small parcel of land. Michael Olson, 1994. **572**

Mind and body spaces: geographies of illness, impairment and disability. 1999. 115

Nature's keepers: the new science of nature management. Stephen Budiansky, 1995. **107**

Person to person: a guide for professionals working with people with disabilities. Lindsay Gething, 1992. **276**

Person to person: a guide for professionals working with people with disabilities. Lindsay Gething, 1997. 277

Philosophy of biology today. Michael Ruse, 1988. 643

Physical environment and behavior: an annotated bibliography and guide to the literature. Joachim F. Wohlwill, 1981. **791**

Plants for people: the psychological and physiological effects of plants. Judith Keane, 1992. **422**

Plants of life, plants of death. Frederick J. Simoons, 1998. 687

Plants, people, and culture: the science of ethnobotany. Michael J. Balick, 1996. 45

Play for all guidelines: planning, design, and management of outdoor play settings for all children. 1987. **540**

Recovery of the measure: interpretation and nature. Robert C. Neville, 1989. **559**

Sacred depths of nature. Ursula Goodenough, 1998. 304

Second nature: a gardener's education. Michael Pollan, 1991. 592

Sensation and perception. Stanley Coren, 1979. 173

Sensation and perception: an integrated approach. Harvey Richard Schiffman, 1976. 657

Sensation and perception: a history of the philosophy of perception. D. W. Hamlyn, 1961. **333**

Sensory integration. 1978. 499

Sharing the harvest: a guide to community supported agriculture. Elizabeth Henderson, 1999. **350**

Square foot gardening. Mel Bartholomew, 1981. 56

Sustaining agriculture near cities. Tufts University. School of Nutrition, 1987. 737

Symposium on Horticulture for the People: XXth International Horticultural Congress, Sydney, Australia, 17 August 1978. 1981. **716**

Therapeutic activities for the handicapped elderly. Charlotte M. Hamill, 1980. 331

Therapeutic and adapted recreational services. Jay Sanford Shivers, 1975. **682**

Therapy through horticulture. Donald P. Watson, 1960. 771

Transitory gardens, uprooted lives. Diana Balmori, 1993. 48

Visual approach to park design. Albert J. Rutledge, 1981. 646

War garden victorious. Charles Lathrop Pack, 1919. 578

When children play: proceedings of the International Conference on Play and Play Environments, 1983: University of Texas at Austin. 1985. **392**

Worlds of sense: exploring the senses in history and across cultures. Constance Classen, 1993. **144**

8

INDEX

The following index of subject terms is based primarily on Library of Congress (LC) subject headings and in a few instances, keywords or phrases for which there was no equivalent LC subject heading. The numbers following each subject heading refer to a specific title in the main bibliography. If the number is preceded by a letter, that letter refers to the items in the following specific chapters. Items without a prefixing letter are from the Books chapter. The serial titles (magazines, journals and other continuations) are not included in this index as their contents do encompass many of the subjects listed below.

C = Curriculum Materials

D = Theses and Dissertations

M = Media

P = Publications with less than 50 pages

A ativity and anomal in advantion	250, 450
Activity programs in education	350, 459,
C5, C6, C95,	Allotment gardens – Germany
Adaptation (Biology)	317,
636, 717,	Allotment of land
Adult day care centers	D24,
806,	Allotment of land – Denmark
Adult education	P42,
D120, D146,	Allotment of land – Great Britain –
Adult learning	History
D161,	542,
Aesthetics, Comparative	Alternative agriculture – Canada
736,	459,
Aging – Environmental aspects	Alternative agriculture – Ontario
636,	(Province) – Goderich (Township)
Aging – United States	D4,
M17,	Alternative agriculture – United
Agricultural ecology	States
531, 532, 533, 534,	459,
Agricultural education	Alzheimer's disease – Patients –
404, 405, D60, D74, D153,	Recreation
Agricultural laborers with disabilities	211, 806,
391,	Alzheimer's disease – Patients –
Agricultural machinery – Design and	Rehabilitation
construction – Catalogs	87, 211, 773. 806,
6, 7,	Amusements
Agricultural machinery for people	401,
with disabilities	Animals
6, 7,	C23, C73,
Agriculture	Animals – Therapeutic use
96, 703,	M29,
Agriculture – Economic aspects	Aphasic persons – Rehabilitation
102,	
	773,
Agriculture – Economic aspects – United States	Aquatic plants
	549,
350,	Arbor Day – Study and teaching
Agriculture – Oregon – Willamette	(Elementary)
Valley	C29,
96,	Architectural design
Agriculture – Study and teaching	369, D12,
D35, P51,	Architectural surveys – United States
Agriculture – United States –	– Forms
Handbooks, manuals, etc.	302,
572,	Architecture – Environmental
Agriculture, Cooperative – Canada	aspects
459,	D30,
Agriculture, Cooperative – United	Architecture – Health aspects
States	185,

Architecture – Human factors	D43,
M41,	Arthritis – Patients
Architecture – Human factors –	P43,
United States	Association of ideas
53,	199,
Architecture – Maryland – Baltimore	Attitude change in children
- Human factors	D156,
622,	Autism
Architecture – Psychological aspects	D32,
185, 365, 513, 652, 805,	Awareness
Architecture – Study and teaching	98,
(Elementary)	
330,	B 1 B1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Architecture, Domestic – Designs	Bach, Edward, 1886-1936
and plans	47, 135, 384, 482, 756,
82,	Backpacking
Architecture, Domestic – Maryland	226,
 Baltimore – Conservation and 	Backyard gardens
restoration	322,
622,	Bakalar, Elsa
Architecture, Modern – 20th century	M37,
185,	Balcony gardening
Architecture and recreation	193, 352, 721, 761,
320,	Barrier-free design
Architecture and the aged	8, 23, 57, 63, 127, 216, 262,
M51,	286, 290, 291, 292, 293, 294, 320,
Architecture and the aged –	336, 369, 391, 458, 462, 491, 524,
Programmed instruction	623, 711, M9, M21, M22, M30, M41
275,	M51, P33, P47, P92, P93, P98,
Architecture and the physically	Barrier-free design – Canada
	9
handicapped	52,
710, 759,	Barrier-free design – Law and
Area planning & development	legislation – United States
D46, D60, D73, D87, D121,	29,
D132, D164,	Barrier-free design – Michigan –
Aromatherapy	Huron-Manistee National Forests
189, 778,	55, P6,
Aromatic plants	Barrier-free design – Programmed
9, 698, 699, 753,	instruction
Aromatic plants – Therapeutic use	275,
750, 751,	Barrier-free design – Standards –
Art – Study and teaching	United States
(Elementary)	53,
C95,	Barrier-free design – United States
Art and science	178, 301, 302, 704,
C95,	Barrier-free design for children with
Arthritis	disabilities

8, P65,	
Barrier-free design for children with	Biosphere
disabilities – United States	410, 483, 484, 485, 486,
29, 742,	Blind – Education
Barrier-free design for older people	544, C78,
117, 460, 462,	Blind – Education – Gardening
Barrier-free design for people with	240, 241, M13, M14, M15, P17,
mental disabilities	P18, P19, P20,
117,	Blind – Great Britain
Barrier-free design for people with	P75,
visual disabilities	Blind – Recreation
P75,	670, M7,
Barrier-free design for students with	Blind – Rehabilitation
disabilities – United States	551,
742,	Blind – Travel
Bars (Drinking establishments)	262,
191,	Body, Human (Philosophy)
Bedding plants	3,
97,106,	Botanical gardens
Beds (Gardens)	D117, D120, D146,
64, 84,	Botanical gardens – Designs and
Bees – Study and teaching	plans
C30,	P25,
Behaviorism (Psychology) – United	Botanical gardens – Australia
States	351,
31, 125, 311,	Botanists
Bible crafts	469,
C109,	Botany
Biointensive gardening	71, 360, 433, 434,
400,	Botany – Experiments
Biological diversity conservation –	C7, C10, C14, C16, C33, C56,
Philosophy	C57, C103, C105, C106, C107,
424,	Botany – Laboratory manuals
Biology – Experiments	C25,
C51,	Botany – Study and teaching
Biology – Philosophy	C22, M36,
304, 314, 315, 316, 387, 410,	Botany – Study and teaching –
483, 484, 485, 486, 490, 500, 503,	Activity programs
545, 614, 642, 643, 644, 651, 689,	C14, C56, C57, C91,
727, 786,	Botany – Study and teaching
Biology – Religious aspects	(Elementary)
304,	C42, C43,
Biology – Social aspects	Botany – Study and teaching
500 ,	(Elementary) – Activity programs
Biology – Study and teaching –	C29,
Activity programs	Botany – Study and teaching
C51	(Elementary) – United States

Child environmentalists - United D26, D112, Botany - Study and teaching States (Preschool) - Handbooks, manuals, D55, etc. Child psychology 18, C33, Child psychology - Cross-cultural Botany, Economic 433, 434, 473, studies Botany, Medical 303, 444, M18, Child psychotherapy Botany projects D22, D25, C9, C10, C28, C105, C106, Children 401, **Boy Scouts** Children - Health and hygiene 577, Building, Wooden 124, 194, 618, Children-Nutrition-Study and Building laws – United States teaching (Elementary) 301, C52, C53, Butterflies - Study and teaching Children and adults D124, C6, Children and the environment 141, 142, D156, Camps for people with disabilities Children with disabilities 224, P65, P99, 320, Campus planning Children with disabilities -524, Education Cancer – Nutritional aspects 648, C101, D54, Children with disabilities- Education Carbonated beverage bottles (Preschool) - Curricula - United States - Case studies Cardiopulmonary system - Diseases 530, C71, Patients Children with disabilities -P23, Recreation Casa Colina 589, Children with mental disabilities -D75, M8, Rehabilitation Causation 169, D18, D50, 425, 426, 427, Cerebral palsied children Children with social disabilities -D1, D2, Education – United States Cerebrovascular disease D26, M12, Children with mental disabilities -Child care services Education (Elementary) P24, 310, Children's gardens Child development 141, 339, 123, 130, 131, 132, 190, 224, Child development – Cross-cultural 265, 267, 321, 430, 448, 481, 520, studies 521, 629, C3, C12, C13, C14, C19, 303, 628, C21, C25, C53, C56, C57, C63, C65,

C80, C81, C82, C84, C92, C103, D7, City planning – United States –

D98, M6, P22,	History – 20th century
	167,
Children's gardens – Handbooks,	Civic Garden Centre (Toronto,
manuals, etc.	Ont.)
88, C36, C74, C75, P16,	P27,
Children's gardens – History	Civilization, Modern – 20th century –
D6,	Psychological aspects
Children's playhouses	287,
218,	Climbing plants
Children's rooms	92,
218,	Clinical health psychology
Chrysanthemums	649, 650,
271,	Cognition
Church work with older people	118, 341, 420,
25,	Cognition disorders – Treatment
Cities and towns – Interpretive	10,
programs – California – Los Angeles	Cognition in infants
344,	740,
Citizens' associations – Australia	Collective farms – United States
134,	350,
City and town life	Community centers – Washington
D70,	(State) – Seattle – Designs and plans
City and town life – Bibliography 67,	D145,
City and town life – Psychological	Community development M2,
aspects	Community development, Urban
776 ,	D40,
City and town life – United States	Community development, Urban –
122,	California – Los Angeles
City children	D113,
396, 538,	Community development, Urban -
City dwellers – Nutrition	Canada
198,	D116,
City of London (England).	Community development, Urban –
Corporation	Illinois – Chicago
590,	D29,
City planning	Community development, Urban
216, 286, 396, 612,	Indiana – Muncie
City planning – 1945-	D85,
317, D30,	Community development, Urban –
City planning – Italy – Turin	Maryland – Baltimore
93,	622,
City planning – New York (State) –	Community development, Urban –
New York – History	New York (State) – New York
252,	581,
	Community development, Urban –

Washington Metropolitan Area Boston – History 449, 769, Community development, Urban – Community gardens - Massachusetts Washington (State) - Seattle Holyoke P39, D160, Community development, Urban -Community gardens - Michigan -Zimbabwe Lansing 504, D121, Community gardens Community gardens - Michigan -Washtenaw County 149, 212, 250, 265, 406, 475, 548, 672, 696, 733, D9, D23, D24, P61, D30, D42, D56, D60, D62, D82, Community gardens - New Jersey -D88, D98, D102, D137, D143, Trenton D155, D158, M5, M10, M2, M28, D54, M35, P69, Community gardens - New York Community gardens - Arizona -(State) - New York D3, D105, D147, M38, Tempe Community gardens - New York D51, (State) - New York - History Community gardens - Austria -Vienna 252, Community gardens - Ohio -34, Community gardens – Bibliography Cleveland 214, 489, 602, Community gardens - California Community gardens - Ontario (Province) - Goderich (Township) D151, Community gardens - California -Los Angeles Community gardens - Ontario D3, D113, (Province) - Toronto Community gardens - California -575, San Francisco Community gardens - Quebec D49, D147, (Province) - Montreal Community gardens - Denmark 584, 402, P42, Community gardens – Social aspects Community gardens -- Economic 243, D3, D4, D40, D116, D132, M38, aspects Community gardens - Switzerland D59, D99, Community gardens - Great Britain 162, 768, Community gardens - Texas -Community gardens - Handbooks, Austin manuals, etc. D147, Community gardens - Texas - Waco 161, Community gardens - History Community gardens - Utah - Salt 769, Community gardens - Indiana -Lake City Muncie D11, D85, Community gardens – United States Community gardens - Massachusetts 149, 233, 350, 389, 430, 550,

D.00 D.140	
D89, D139,	453, 479, 494, 509, 511, 522, 527,
Community gardens – United States	549, 576, 586, 587, 598, 607, 639,
- Anecdotes	640, 684, 691, 701, 706, 718, 719,
342,	720, 721, 732, 788, 794, 795, 798,
Community gardens United States	799, 800, 801, P11, P38, P71, P84,
– History	Container gardening –
D10, D98,	Encyclopedias
Community gardens – Vermont –	164,
Burlington	Container gardening – Great Britain
D130,	423,
Community gardens – Victoria –	Container gardening – Pictorial
Melbourne	works
D41,	75, 358, 359, 411,
Community gardens – Washington	Container gardening – Quebec
Metropolitan Area	(Province) – Montreal
449,	584,
	· · · · · · · · · · · · · · · · · · ·
Community gardens – Washington	Cooking (Beans)
(State)—Seattle	C24,
288, 381,	Cooking (Corn)
Community gardens – Washington	C24,
	· · · · · · · · · · · · · · · · · · ·
(State) – Seattle – Designs and plans	Cooking (Flowers)
D12, D133,	203,
Community gardens – Wisconsin –	Cooking (Herbs)
Milwaukee	90,
	*
D131,	Cooking (Squash)
Community Shared Agriculture	C24,
D4,	Cooking (Vegetables)
Compost – Study and teaching	C97,
(Secondary)	Cooperative societies
	*
M42,	449,
Conduct of life	Cooperative societies –
138,	Saskatchewan – Moose Jaw
Conservation of natural resources	M43,
	· · · · · · · · · · · · · · · · · · ·
421,	Cosmology – History
Conservation of natural resources –	630,
Problems, exercises, etc.	Cottage gardens – Massachusetts
C98,	M27,
Consumer cooperatives	Counseling
	9
475,	558,
Container gardening	Courtyard gardens
2, 33, 44, 60, 61, 62, 70, 74, 77,	352,
80, 85, 90, 126, 152, 153, 154, 156,	Creative activities and seat work –
182, 184, 193, 206, 207, 208, 209,	Handbooks, manuals, etc.
219, 220, 221, 235, 237, 239, 272,	C32,
289, 306, 335, 347, 354, 356, 357,	Criminals – Rehabilitation
358, 359, 379, 407, 408, 412, 413,	242, D128, M52, P13,
423, 431, 437, 438, 441, 443, 452,	Crop yields

D59,	D34,
Crowding stress	Disability studies
395,	115,
Cucurbitaceae – Study and teaching	Discrimination against people with
(Elementary)	disabilities – Great Britain
C43,	257,
Customer relations	Dried flower arrangement
260,	679, 680,
Cut flowers	Dwellings – Barrier-free design –
679, 680,	United States
···, ···,	704,
	, , ,
Day care centers	
310,	Earthworms
Day care centers for the aged –	C31,
Activity programs	East Bay Frenchy-American School
211,	(Berkeley, Calif.)
Daylighting	C14,
D30,	Ecology
Decoration and ornament – Plant	421,
forms	Ecology – Experiments
M49,	C98,
Delridge (Seattle, Wash.)	Ecology – Study and teaching
288,	C92,
Depressed persons – Care	Ecology – Study and teaching
D135,	(Elementary)
Desert gardening	C5, C49,
C39,	Ecology – Study and teaching
	=: : : : : : : : : : : : : : : : : : :
Developmental disabilities –	(Elementary)
Environmental aspects	C88, C89, C90,
668,	Ecosystem management
Developmentally disabled –	107,
Employment – United States	Edible landscaping
20,	P53,
Developmentally disabled – Housing	Education – United States
D20,	524,
Developmentally disabled –	Education, Cooperative –
Psychology	Handbooks, manuals, etc.
D103,	385,
Developmentally disabled –	Education, Elementary – Curricula
Recreation	D31,
476,	Education, Preschool
Developmentally disabled –	83,
Rehabilitation	Education, Preschool – Activity
D125, M44, P77,	programs – United States – Case
Developmentally disabled – Training	studies
of	C71.

Education, Urban – Michigan –	Environmental education – United
Detroit Metropolitan Area	States
763,	D55,
Educational games	Environmental engineering
519,	462,
Educational innovations – United	Environmental ethics
States	736,
D26, D112,	Environmental health
Educational sociology	779,
D87,	Environmental management
El Mirasol Educational Farm, Santa	419,
Barbara, California	Environmental policy – History
160,	760,
Electric power production	Environmental policy – United
P30,	States
Endangered plants	507, 508,
C41,	Environmental protection
Entrepreneurship	421, C99,
D88,	Environmental protection –
Environment	Problems, exercises, etc.
514, 624,	C98,
Environmental design	Environmental protection – Study
513,	and teaching – United States
Environmental education	D55,
C62, C87, C93, C94, C100,	Environmental psychology
D146,	16, 17, 18, 19, 31, 40, 59, 68,
Environmental education – Activity	69, 86, 116, 118, 119, 120, 121, 125,
programs	133, 145, 150, 234, 236, 251, 261,
C66, C79, C92, C99,	300, 305, 307, 311, 345, 346, 362,
Environmental education –	365, 366, 396, 414, 416, 417, 418,
California – Berkeley – Case studies	419, 420, 460, 461, 513, 514, 538,
C72,	599, 600, 312, 631, 636, 646, 667,
Environmental education –	668, 671, 704, 708, 709, 726, 739,
Curricula	762, 764, 765, 766, 767, 775, 776,
D148,	790, D8, D58, D64,
Environmental education – Great	Environmental psychology –
Britain	Bibliography
C108,	446, 447, 791,
Environmental education – Study	Environmental psychology – Cross-
and teaching	cultural studies
D148,	11, 12, 13, 14, 15,
Environmental education – Study	Environmental psychology – History
and teaching (Elementary) –	630, 790,
Michigan – Detroit Metropolitan	Environmental psychology –
Area	Research
763,	805,

Environmental responsibility 580, 790, Exercise therapy Environmental sciences 111, 112, C99, Environmental sciences -Experiments Experiments C8, C10, C105, C99, Environmental sciences -Philosophy – History Fairview Gardens (Santa Barbara, 760, Calif.) - History Environmental sciences - Study and 1, teaching Family recreation 210, C93, C94, C100, Environmental sciences - Study and Farm layout teaching (Elementary) 391, C79, C88, C89, C90 Farm produce – United States – Environmentalism Marketing 676, 350, Environmentalism - Psychological Farmers – California – Biography aspects 1. 145, 150, 631, Farms Ergonomics and human-machine P94, relations Farms - Great Britain 758, 768, Essences and essential oils -Farms, Small - Ontario (Province) -Goderich (Township) Therapeutic use 750, 751, D4, Ethics Farms, Small - United States -Handbooks, manuals, etc. 251, Ethnic neighborhoods - Hawaii -572, Honolulu Fathers and sons D72, 502, Ethnobotany Female offenders 45, 433, 434, D91, Ethnophilosophy Feng shui 797, 427, European Year of the Environment, Floriculture 1987 403, 626, 627, 613, Flower arrangement Evoked potentials 203, 335, C3, (Electrophysiology) Flower gardening 681, 97, 203, 235, 335, 579, 679, Evolution (Biology) – Philosophy 680, 316, 503, Flowers Exceptional children 203, 586, 679, 680, Flowers - Drying P22, Exercise for older people 679, 680,

Flowers – Folklore	Games
203,	C34, C48,
Flowers – Odor	Garden animals
626, 627,	C60,
Flowers – Psychological aspects	Garden borders
D36, D106,	64, 84, 97, 546,
Flowers – Therapeutic use	Garden cities
46, 47, 203, 213, 384, 482, 512,	382,
656, 678, 697, 756,	Garden ecology
Flowers – Therapeutic use –	337, 493,
Handbooks, manuals, etc.	Garden structures
188,	183, 197, 321, 322,
Folklore – United States	Garden structures – Design and
619,	construction
Food habits – New Jersey – Trenton	618,
D54,	Garden structures – Designs and
Food supply	plans
475, 749, P78,	D11,
Food supply – Ghana – Accra	Garden tools
566,	186, 264, 327, 692, 783, P62,
Fountains – Missouri – Kansas City	P85,
529,	Gardeners
Fragrant gardens	342, 755, 769, D137,
9, 698, 699, 753, 754,	Gardeners – Attitudes
Freshwater ecology – Study and	D7, D121, D155, D161,
teaching (Elementary)	Gardeners – United States –
C45,	Biography
Freshwater plants – Ecology – Study	M37,
and teaching (Elementary)	Gardening
C45,	66, 130, 131, 132, 146, 147,
Fruit	155, 170, 172, 205, 212, 228, 229,
C60,	230, 265, 266, 360, 367, 370, 380,
Fruit – Propagation	431, 439, 442, 488, 515, 516, 522,
P34,	548, 592, 616, 632, 701, 712, 713,
Fruit – United States – Marketing	728, 735, 746, 747, 761, 770, 774,
468,	780, 784, 797, 800, 801, C3, C11,
Fruit-culture	C35, C48, C60, C63, C64, C80, C83,
2, 182, 468, 684, 690, 691, 700,	C103, C104, D143, M5, M6, M10,
C60, P34,	M34, M37, M40, P89,
Fruits – Marketing	Gardening – Australia – Equipment
700,	and supplies
Fund raising	722,
P55,	Gardening – Bibliography
1 55,	489,
	Gardening – Environmental aspects
Gaia hypothesis	337,
410, 484, 486,	•
, , ,	

Gardening – Equipment and	Gardening – Study and teaching –
supplies	United States
186, 264, 692, 783, 758, C82,	D6,
Gardening – Experiments	Gardening – Study and teaching
472, C7, C14, C16, C56, C57,	(Elementary)
C107,	567, C29, C54, C55, C82, D16,
Gardening – Folklore	Gardening – Study and teaching
619,	(Elementary) – Activity programs
Gardening – Handbooks, manuals,	C54, C55,
etc.	Gardening – Study and teaching
C54, C55,	(Secondary) – United States
Gardening – History	D156,
363,	Gardening – Texas – Waco
Gardening – Minnesota	249,
M48,	Gardening – Therapeutic use
Gardening – New Jersey	21, 41, 50, 51, 75, 76, 99, 101,
P79,	105, 110, 129, 139, 166, 168, 181,
Gardening - New Zealand	196, 199, 242, 244, 255, 259, 263,
43,	268, 269, 273, 281, 284, 324, 325,
Gardening – North America	337, 338, 355, 361, 371, 372, 373,
C18, C26,	374, 375, 376, 378, 385, 448, 505,
Gardening – Ontario – Toronto –	518, 525, 526, 535, 536, 537, 543,
History	544, 551, 552, 553, 554, 555, 556,
P27,	557, 564, 565, 571, 573, 579, 582,
Gardening – Philosophy	603, 609, 611, 654, 670, 675, 683,
493, 563, 592, 780, 797,	688, 693, 697, 715, 726, 730, 738,
Gardening – Psychological aspects	745, 771, 773, 804, 807, C70, C78,
565, D108,	D1, D15, D17, D18, D21, D22,
Gardening Psychological aspects -	D23, D25, D28, D32, D33, D34,
Testing	D39, D43, D44, D46, D47, D48,
D103,	D50, D52, D57, D63, D65, D68,
Gardening – Religious aspects	D69, D71, D75, D76, D77, D83,
282,	D84, D91, D92, D93, D94, D95,
Gardening – Research	D96, D107, D110, D111, D114,
472,	D115, D118, D119, D123, D125,
Gardening – Social aspects	D126, D128, D135, D141, D142,
D38,	D144, D149, D150, D157, D159,
Gardening – Social aspects – Hawaii	D163, D165, M8, M11, M16, M19,
– Honolulu	M24, M20, M29, M33, M39, M46,
D72,	M47, M48, M50, M52, P3, P13, P15,
Gardening – Study and teaching	P22, P24, P37, P41, P44, P46, P49,
C4, C21, C50, C92, D35, D80,	P50, P54, P55, P57, P64, P77, P80,
D141, M36, M45, P30,	P84, P88, P89,
Gardening – Study and teaching –	Gardening - Therapeutic use -
Activity programs	Australia
C67,	P76,

Gardening – Therapeutic use –	Gardening – United States
Bibliography 280, 422, 501, 608, 748, P58,	574, 619, Gardening – Vermont – Burlington
P72,	D130,
Gardening – Therapeutic use –	
Handbooks, manuals, etc. 385, P87, P90,	Gardening – Victoria – Melbourne D41,
Gardening – Therapeutic use –	Gardening – Virginia – Handbooks,
Illinois D119,	manuals, etc.
Gardening – Therapeutic use –	610, Gardening – Vocational guidance
Illinois – Chicago	P44,
D29,	Gardening – West Virginia – Tucker
Gardening – Therapeutic use –	County
Kansas	D109,
D108,	Gardening for older people
Gardening – Therapeutic use –	43, 49, 94, 95, 136, 137, 139,
Minnesota	375, 377, 390, 448, 498, 544, 552,
D152, Gardening – Therapeutic use – New	583, 588, 590, 633, 653, 713, 715, 722, 757, 793, 803, D21, D45, D129,
York (State) – New York	M17, M19, M32, P14, P23, P66, P70,
P60,	P74, P85,
Gardening – Therapeutic use – New	Gardening for older people –
Zealand P76,	Bibliography 279,
Gardening – Therapeutic use –	Gardening for older people –
North Carolina	Equipment
D119,	D122,
Gardening – Therapeutic use – Ohio D119,	Gardening for people with disabilities
Gardening – Therapeutic use –	5, 43, 49, 123, 129, 136, 137,
South Carolina – Columbia	146, 147, 196, 215, 231, 232, 240,
180,	241, 244, 259, 263, 325, 326, 368,
Gardening – Therapeutic use –	374, 377, 390, 480, 498, 551, 553,
Study and teaching	590, 613, 633, 639, 640, 653, 670,
M11, P56,	673, 693, 757, 758, 792, 793, 803,
Gardening – Therapeutic use –	804, 807, 808, C19, D21, D37, D39,
Study and teaching – Australia –	D104, D127, M7, M12, M13, M14,
Victoria	M15, M16, M24, M32, M33, M46,
332,	P10, P9, P14, P15, P17, P18, P19,
Gardening – Therapeutic use –	P20, P21, P22, P23, P24, P28, P29,
United States 20,	P31, P38, P41, P43, P46, P48, P57, P64, P66, P67, P70, P71, P74, P80,
Gardening – Therapeutic use –	P84, P85, P86, P88, P90, P95, P96,
United States – Case studies	Gardening for people with
274,	disabilities – Australia
۵، ٦,	722, P76,
	122, 1 10,

Gardening for people with	24,
disabilities – Australia Victoria –	Gardens – Pictorial works
Doncaster	179, 516, 669, 677,
428, 429,	Gardens - Psychological aspects
Gardening for people with	24, 254, 338, D161,
disabilities – Bibliography	Gardens – Religious aspects
279, P58,	253, 398, 399, 547,
Gardening for people with	Gardens – Social aspects
disabilities – Florida – Marianna	253, 254, 322,
P77,	Gardens – Styles
Gardening for people with	398, 399, 560,
disabilities – Great Britain –	Gardens – Styles – History
Equipment and supplies – Catalog	363,
148,	Gardens – Symbolic aspects
Gardening for people with	253, 547,
disabilities – New Zealand	Gardens, Hmong – California –
P76,	Sacramento
Gardening to attract wildlife	D27,
104,	Gentrification – New York (State) –
Gardens	New York
205, 469, 755, P63,	M38,
Gardens – Australia – Design	Geographical perception
722,	300, 305, 594, 624, 776,
Gardens – Denmark	Gifford Pinchot National Forest
402,	(Wash.)
Gardens – Design	54, P5,
64, 155, 179, 246, 253, 255,	Girls – Societies and clubs
321, 322, 398, 399, 431, 510, 517,	577,
546, 560, 597, 603, D129, D134,	
Gardens – Hawaii – Honolulu	Gleaning
	475,
D72,	Grasses – Study and teaching
Gardens – History	(Elementary)
317, 363,	C42,
Gardens – Juvenile literature	Green movement – History
401,	760,
Gardens – Massachusetts	Green technology
M27,	C66,
Gardens – Massachusetts –	Greenhouse gardening
Northampton	M43, P67,
D82,	Greenhouse management
Gardens – New York (State) – New	403, M43,
York	Greenhouses
48,	P64,
Gardens – Ontario – Toronto –	Greenhouses – Design and
History	construction – New Jersey
P27,	P59,
Gardens – Philosophy	Greenhouses - Designs and plans

D11, P67,	90, 182, C60,
Group psychotherapy for youth	Herbal cosmetics
200,	203,
Group relations training	Herbal remedies
98,	384,
Growth (Plants)	Herbs
C102,	90, C60,
Growth (Plants) – Experiments	Herbs – Therapeutic use
C102,	41, 525, 526, 697, M18,
G102,	Hermeneutics
	559, D13,
Hand – Diseases	HETA-Psych (Test)
D43,	D103,
Handelsman, Judith F., 1948-	Hispanic Americans
334,	D3,
Handicapped employment	Historic sites – Interpretive
385,	programs – California – Los Angeles
Handicraft	344,
C27, C32, C38, C43, C48, C68,	Hmong (Asian people) – Agriculture
C91, C103, C104, P12,	D27,
Hanging baskets	Hmong American women –
206, 237, 270, 511,	California – Sacramento – Social life
Hanging plants	and customs
206, 237, 438, 441,	D27,
Harris County Center for the	Holidays
Retarded, Houston, Texas	C32,
D126,	Holistic medicine
Health	482,
616,	Homeless Garden Project
Health facilities	459, 695, M3, M26,
D52,	Homeless persons – Employment –
Health facilities – Design and	New York (State) – New York
construction	P60,
166, P2,	Homeless persons – Georgia –
Health facilities – Landscape	Macon – Social conditions
architecture	243,
166, D44, D135, P66,	Homeless persons' gardens –
Health facilities – Landscape	California – Santa Cruz
architecture – United States – Case	459, 695, M3, M26,
studies	Homeless persons' gardens – New
274,	York (State) – New York
Health facilities Psychological	48,
aspects	Homelessness – Georgia – Macon
M19,	243,
Health professionals	Homeopathy - Materia medica and
78,	therapeutics
Herb gardening	46, 47, 213, 384, 656, 678,

Horticultural products - Rating of Human beings – Effect of environment on D125, Horticultural service industry --14, 15, 59, 261, 419, 729, D117, Pennsylvania – Philadelphia Human ecology D66, 3, 11, 12, 13, 15, 59, 222, 414, Horticultural writers - United States 421, 507, 508, 787, Biography Human ecology - Asia 334, 717, Horticulture Human ecology - Japan -555, 609, 716, D14, D24, Philosophy D107, P88, 307, Horticulture - Great Britain -Human ecology – Philosophy History 145, 383, 424, 570, Human ecology - Psychological 772, Horticulture – Recreational use aspects D68, 287, 383, Horticulture – Social aspects Human ecology - Religious aspects 255, 393, 609, 675, M50, Horticulture - Study and teaching Human ecology - Study and 242, 536, C78, D59, teaching (Elementary) Horticulture - Study and teaching C17, (Elementary) Human engineering 543, C29, D138, 513, D122, Horticulture – Therapeutic use Human geography D78, D124, M44, M47, 286, 601, Horticulture - United States -Human geography - Methodology Handbooks, manuals, etc. Human geography - Moral and 572, $Hospitals-Landscape\ architecture$ ethical aspects 639, 640, 736, Human-plant relationships Hotels 71, 238, 255, 421, 474, 675, 191, House plants 687, 782, D148, D157, 58, 81, 452, 454, 721, 794, 795, Human-plant relationships -Dominican Republic - Rio Limpio House plants in interior decoration – Psychological aspects D64, Human-plant relationships -House plants in office decoration Research 238, Huron-Manistee National Forests Housing, Cooperative - Washington Metropolitan Area (Mich.) 449, 55, P6, Human behavior 4, M1, Human beings – Cross-cultural Indian cookery studies C18, C24, 11, 12, 13,

- W	
Indians of North America –	Kolonihaveforbundet for Danmark
Agriculture	– History
103, C18, C26,	P42,
Indians of North America – Folklore	
C17, C18, C26,	Land use – Planning
Indians of North America – Food	D30,
C26,	Land use, Urban
Indians of North America –	102, 725, 737,
Mythology	Land use, Urban – Africa
D165,	318,
Indoor gardening	Land use, Urban – Kenya – Nairobi
452, 453, 721, 794, 795, C60,	256,
C69, C104, P34	Landscape
Interdisciplinary approach in	594,
education	Landscape – Health aspects
D35,	779, P2,
Intergenerational relations	Landscape – Therapeutic use
448, D77, D124,	779, D44,
Interior decoration – Psychological	Landscape architecture
aspects	23, 82, 177, 212, 321, 396, 620,
185,	645, 647, D20, D134, M30,
Intersensory effects	Landscape architecture – Design
681,	D79,
	Landscape architecture –
	Environmental aspects
Japan – Civilization	M30,
32,	Landscape architecture – Japan
Job analysis	307,
D153,	Landscape architecture – Missouri –
Journalists – Biography	Kansas City – History
502,	529,
Jungian psychology	Landscape architecture – Pictorial
D165,	works
Juniper Gardens Children's Project	177,
31, 125, 311,	Landscape architecture –
Juvenile delinquents	Psychological aspects
D96,	652, D58, D61, D79,
	Landscape architecture – Public
	opinion
Kevin Heinze Garden Centre –	D45,
History	Landscape architecture –
428, 429,	Therapeutic use – United States
Kitchen gardens	D61,
C14,	Landscape architecture – United
Knowledge, Theory of	States
492, 765, 776,	621,
	•

344,

Landscape architecture for people

with disabilities Low-maintenance gardening 227, 409, 641, 710, 711, M9, 487, Landscape assessment – Japan 307, Landscape design Manners and customs 478, 710, 711, D46, D129, 4, M1, Landscape design – Sweden Master gardeners D87, P50, Landscape gardening Master Gardeners of Johnson 33, 82, 183, 260, 322, 440, 487, County, Kansas 510, 516, 546, 597, 603, 755, 769, D77, Materia medica, Vegetable D14, P53, Landscape gardening – British 135, Columbia – Case studies Mathematics - Study and teaching (Elementary) – Activity programs 615, Landscape in art C67, Medical geography 27, 28, 601, 779, Landscape in literature 594, Medical personnel - Great Britain -Language arts (Elementary) Attitudes C6, 257, Learning by discovery Medicinal plants C33, 525, 526, M18, Learning disabled – Services for – Medicinal plants - Folklore Great Britain 687, 364, Medicine, Herbal Leaves – Collection and preservation 656, 678, - Study and teaching (Elementary) Medicine, Psychosomatic - Cases, clinical reports, statistics C46, Leaves - Study and teaching 202, (Elementary) Melwood Horticultural Training C46, Center Leisure - Great Britain - Planning D126, Mental health Leisure - Study and teaching 47, 110, 113, 135, 138, Mental health facilities - Design and 476, Local transit stations - Great Britain construction P75, 121, D79, Long-term care facilities Mental illness – Diagnosis 78, 79, D84, Long-term care facilities -Mental retardation - Rehabilitation Recreational activities 169, 39, 308, 343, 596, Mentally ill Long-term care of the sick D69, 78, 79, Mentally ill - Care Los Angeles (Calif.) – History 268, 273, 372,

Mentally ill – Employment – New	Natural history – Study and teaching
York (State) – New York	(Early childhood) – Activity
P60,	programs
Metaphysics	C40,
559,	Natural history – Study and teaching
Metropolitan areas – Asia	(Elementary) – Activity programs
1	
283,	C88, C89, C90,
Michigan State University. 4-H	Natural resources
Children's Garden	593,
D90,	Naturalism – Religious aspects
Mind and body therapies	304
113,	Nature
Minorities – California – Los	C23, C34, C73,
Angeles – History	Nature – Effect of human beings on
344,	107, 414, 507, 508, 570, 593,
Modeling – Therapeutic use	676,
D142,	Nature – Psychological aspects
Motels	19, 26, 113, 138, 145, 150, 287,
191,	383, 414, 416, 417, 591, 625, 630,
Mulching	631, 775, D8,
712, 714,	Nature – Religious aspects
Mythology – Psychological aspects	304, 635,
26,	Nature (Aesthetics)
	736,
S	Nature, Healing power of
National parks and reserves	113, 482, 558, D61, D134,
D13,	Nature and nurture
National parks and reserves – United	133, 141, 142, 636,
States	Nature and nurture – United States –
743,	Case studies
Native plant gardening	C71,
723,	Nature conservation
Native plant gardens	107, 760, 786,
723,	Nature conservation – Philosophy
Native plants for cultivation	424,
723,	Nature conservation – Religious
Natural gardens – United States	aspects
493,	305,
Natural history	Nature conservation – Study and
593, C58,	teaching (Elementary)
	= '
Natural history – Experiments	C77,
C8,	Nature craft
Natural history – Outdoor books	146, 147, 753, C3, C27, C37,
C69,	C40, C60, C63, C64, C68, C83,
Natural history – Study and teaching	C104, C109, P12,
729, C62,	Nature study

404, 405, C5, C58, C62, C69, 726, C77, D74, Nursing homes - Recreational Nature study – Activity programs activities C17, C40, C88, C89, C90, 108, 211, 596, D94, Nature study – Study and teaching Nursing homes – Texas – Waco (Preschool) - Curricula - United 249, States - Case studies Nutrition 530, C71, 616, Nutrition - Study and teaching Nature trails for people with visual disabilities (Elementary) 244, D39, D16, Naturopathy Nutrition extension work - Sudan 525, 526, P32, Neighborhood Garden Association Neighborhood improvement Occupational therapy 10, 76, 99, 110, 163, 169, 331, programs 372, 373, 386, 455, 456, 457, 582, D40, New business enterprises - United 734, D21, D76, D92, D107, D123, States D163, P44, P49, 468, Occupational therapy - Minnesota New business enterprises – United D152, States - Handbooks, manuals, etc. Occupational therapy – Trends D115, New York (City) - School Garden Occupational therapy for children 128, 169, 568, 569, Association D97, Occupational therapy for older Nurseries (Horticulture) people 260, D154, 87, 211, 343, 535, 773, 806, Nurseries (Horticulture) - Economic D19, Older people – Dwellings – Design aspects 565, and construction Nursery dealers 117, D154, Older people – Dwellings – Planning Nursery growers D154, Older people - Dwellings - United Nursery schools States 310, 14, 116, 460, 461, 767, Nursery stock Older people – Health and hygiene 221, 248, 343, D19, D38, Older people – Housing – British Nursing home care Columbia - Case studies 726, Nursing home patients - Mental 615, health Older people – Institutional care – 726, Canada Nursing homes – Psychological D95, aspects

Older people – Institutional care –	167,
United States	Open spaces – United States –
461,	Planning
Older people – Mental health	285,
535,	Organic farming
Older people – Nutrition – Study	157, 159, 319, 468, 532, 534,
and teaching	714,
324,	Organic farming – California – Santa
Older people – Psychology	Barbara
D150,	1,
Older people – Recreation	Organic gardening
25, 94, 95, 245, 580, 596, D28,	157, 158, 159, 160, 337, 532,
D68, D93,	561, 714, 770, D137, M42,
Older people – Rehabilitation	Organic gardening – Australia –
248, 331, 552,	Hastings (Vic.)
Older people – Social conditions –	P35,
Congresses	Ornamental climbing plants
636,	237,
Older people – United States	Ornamental horticulture
* *	
M17,	D5,
Older people – United States –	Ornamental plant industry
Psychology	270,
767,	Outdoor education
Older people – United States –	83, 685, 707, C76, C87, C101,
Social conditions	P51,
116,	Outdoor education – California –
Older people – West Virginia –	Berkeley – Case studies
Tucker County	C72,
D109,	Outdoor recreation
Ontology	25, 65, 523, P8,
500,	Outdoor recreation – Equipment
Open spaces – California – Los	and supplies
Angeles	C2,
D113,	Outdoor recreation – Planning
Open spaces – Design and	744,
construction	Outdoor recreation – Therapeutic
	1
D73,	use
Open spaces – England – London	D133,
590,	Outdoor recreation for children
Open spaces – Italy – Turin	83, 210, 226, 339, 519, 540,
93,	541, 577, 617, 685, D100,
Open spaces – New York (State) –	Outdoor recreation for children –
New York – History	Environmental aspects
252,	396,
Open spaces – Psychological aspects	Outdoor recreation for children with
776,	disabilities
Open spaces – United States	C2, D148,
1 1	,,

252,

Paleoethnobotany

313,	Parks – Planning
Paraplegics	647,
D5,	Parks - Psychological aspects
Parks	646,
645,	Parks – United States – Design and
Parks – Access for the physically	plans
handicapped – California –	388,
Handbooks, manuals, etc.	Parkways – Missouri – Kansas City
595,	529,
Parks – Barrier-free design	Patio gardening
42, 227, 388, 613, P45,	193, 352, 431, 436, 586, 706,
Parks Barrier-free design – Alaska	774, 798, 799,
741,	Peasantry – Dominican Republic –
Parks – Barrier-free design – Indiana	Rio Limpio
D67, Parks Barrior from decian	89,
Parks – Barrier-free design –	Pediatric neurology
Maintenance and repair	681,
P45,	People with disabilities
Parks – Barrier-free design – Texas	23, 115, 127, 163, 165, 225,
P81, P82, P83,	276, 277, 278, 286, 491, 524, C70,
Parks – Barrier-free design – United	D75,
States 740	People with disabilities – Australia
388, 740,	Case studies
Parks – Barrier-free design –	276, 277, 278,
United States – Handbooks,	People with disabilities –
manuals, etc.	Employment
30, 705,	D5, D37, M8,
Parks – California – Interpretive	People with disabilities –
programs – Handbooks, manuals,	Employment – Saskatchewan –
etc.	Moose Jaw
595,	M43,
Parks – California – San Francisco	People with disabilities –
D164,	Government policy – United States
Parks – Design and construction	432,
646, 647,	People with disabilities – Great
Parks – Designs and plans	Britain
42, 388, D164,	216, 257,
Parks – England – London	People with disabilities – History
590,	286,
Parks – Italy – Turin	People with disabilities – Housing -
93,	British Columbia – Case studies
Parks – Missouri – Kansas City –	615,
History	People with disabilities Life skills
529,	guides
Parks – New York (State) – New	673,
York – History	

People with disabilities – Medical	People with mental disabilities –
care – Great Britain	Education
257,	D80,
People with disabilities – New	People with mental disabilities –
Zealand	Employment
43, D104,	D66, D154,
People with disabilities – Public	People with mental disabilities –
opinion	Housing – Design and construction
276, 277, 278,	117,
People with disabilities – Recreation	People with mental disabilities –
65, 204, 215, 661, 662, 682,	Recreation
743, D13, D67, D144, M12, P1, P8,	D73,
P99,	People with mental disabilities –
People with disabilities – Recreation	Rehabilitation
 United States 	140, 394, 611, D57, D76,
109, P81, P82, P83,	D110, D126, D127, D142, P37,
People with disabilities - Recreation	People with visual disabilities
 United States – Handbooks, 	284, D48, M13, M14, M15,
manuals, etc.	P21,
705,	People with visual disabilities –
People with disabilities - Recreation	Recreation
 United States – Planning 	807, 808, P29,
109,	Perception
People with disabilities - Recreation	70, 100, 174, 175, 295, 296,
Trails – Washington (State) –	297, 298, 299, 312, 333, 470, 471,
Gifford Pinchot National Forest	624, 657, 658, 659, 660, 796,
54, P5,	Perception – Physiological aspects
People with disabilities –	173,
Rehabilitation	Perception (Philosophy)
165, 194, 195, 331, 455, 456,	3,
457, 704, 734, D76, D114, M8, P9,	Perception, Disorders of
P76,	151,
People with disabilities - Services for	Perception in children – Testing
65, 127, 661,	681,
People with disabilities - Services for	Perennials
– Great Britain	531, 533, 534,
364,	Perfumes
People with disabilities - Services for	626, 627,
 United States 	Permaculture
662, 742,	66, 531, 532, 533, 534, P52,
People with disabilities Vocational	P53,
guidance	Personal space
D66,	395,
People with mental disabilities	Pests – Control
564, C70,	D121,
People with mental disabilities –	Pets – Therapeutic use
Care	725, D22, D25, M19,

Philosophy, American – 20th C105, C106, Plants - Folklore century 559, 71, 73, 435, 619, 687, Philosophy of nature Plants - Miscellanea 3, 32, 107, 383, 424, 425, 426, 435, 427, 559, 570, 635, 676, 686, 787, Plants - Mythology Philosophy of nature - History 71, 687, 630, Plants - Psychological aspects Physical anthropology D64, Plants – Social aspects Physical education for people with 71, disabilities Plants – Social aspects – Dominican Republic – Rio Limpio 320, Physical fitness 89, Plants - Study and teaching 616, Physiology (Elementary) 666, C29, Plants - Therapeutic use Physiology, Comparative 499, 603, Places – United States Plants, Cultivated - Folklore 435, 167, Plant conservation Plants, Ornamental 75, 84, 106, 206, 209, 217, 347, C41, Plant conservation - Citizen 356, 479, 487, 510, 586, 724, 794, 795, participation C41, Plants, Ornamental – Hawaii – Plant containers Honolulu 126, 156, 423, 576, 639, 640, D72, Plants, Ornamental – History P38, P84, Plant ecology - Study and teaching (Elementary) - Great Britain Plants, Ornamental – Pictorial works C1, C17, 75, Plant introduction - Study and Plants, Potted teaching (Elementary) - Great 208, 217, 272, 598, Britain Plants, Protection of C1, 403, Plants, Useful Plant propagation 221, C102, D31, 473, Plant varieties Plants, Useful - Dominican Republic Rio Limpio 272, Plants 89, 71, 99, 731, C4, C23, C38, C50, Plants, Useful – Study and teaching (Elementary) - Great Britain Plants - Ecology - Study and teaching (Elementary) Plants and crops – Therapeutic use C44, D37, Plants as noise barriers Plants – Experiments

183,	D122,
Plants for play environments	Psychiatric hospitals – Landscape
539,	architecture – United States
Plastic scrap – Recycling	D61,
C51,	Psychic trauma
Play	D46,
353, 392, 519, 538,	Psychobiology
Play Cross-cultural studies	133,
628,	Psychological tests
Play – Philosophy	D103,
258,	Psychological tests for children
Play environments	681,
83, 190, 218,	Psychology
Play environments – Designs and	251,
plans	Psychology – Experiments
D2,	176,
Playgrounds	Psychology – Research
91, 353, 392, 463, 464, 465,	708,
466, 467, 523, 589, D101, P68, P73,	Psychology and philosophy
Playgrounds – Design and	630,
construction	Psychoanalysis
143, 258, 478, 540, 541, 674,	202,
D2,	Psychometrics
Playgrounds – Equipment and	176,
supplies	Psychotherapy
143, 353, 589, 634,	D78,
Playgrounds – Planning	Psychotherapy, Group
143, 330, 353, 540, 541, 674,	505,
P39,	Public art – California – Los Angeles
Playgrounds – Social aspects	344,
339,	Public buildings – Barrier-free design
Pollution	P65,
593,	Public buildings – Barrier-free design
Power lawn mowers	United States
327,	53, 301,
Power tools	Public health – New Jersey –
327,	Trenton
Primitive societies	D54,
425, 426, 427,	Public history – California – Los
Prisoners – Education	Angeles
242, D71,	344
Problem youth	Public spaces
D96,	122,
Professional projects – Urban	Public spaces – California – Los
planning	Angeles
288,	344,
Pruning – Equipment and supplies	утт,
r runnig – Equipment and supplies	

Public spaces – Barrier-free design – 741, United States Recreation areas and people with 740, disabilities – Australia – Sydney Public spaces – Psychological (N.S.W.) 562, aspects Recreation areas and people with disabilities - Colorado Public spaces – United States – Planning – Case studies P97, 122, Recreation areas and people with Pumpkin disabilities - Great Britain C43, M31, 364, Recreation areas and people with disabilities - Texas QuickTime P81, P82, P83, D90, Recreation areas and people with Quietude disabilities - United States 230, 740, 777, Recreation areas and people with disabilities - United States -Raised bed gardening Handbooks, manuals, etc. 181, 184, 196, 528, 546, 640, 30, 301, 746, 747, M40, P70, P71, P95, Recreational therapists Rare plants 79, C41, Recreational therapy 35, 37, 108, 204, 284, 682, D48, Recreation 111, 112, 580, D164, D68, D94, P64, Recreation areas Recreational therapy – Dictionaries 191, 523, 538, D101, 36, Recreation areas – Access Recreational therapy – Evaluation 245, 740, 759, 111, 112, Recreational therapy – Management Recreation areas - Design and construction 245, 647, 674, 744, Recreational therapy - Study and Recreation areas - Indiana teaching (Higher) 38, P4, Employees – Attitudes Recreational therapy for older D67, Recreation areas – Planning people 647, 674, 39, 308, 343, 580, 596, Recreation areas - Planning -Recreational therapy for youth Michigan – Huron-Manistee 200, National Forests Recycling (Waste, etc.) 198, M42, P34, 55, P6, Reformatories for women Recreation areas and people with disabilities D91, 744, D67, P99, Rehabilitation Recreation areas and people with 804, D21, P49, P57, disabilities – Alaska

Rehabilitation – Virginia – Fairfax	Schizophrenics
County	D163,
556,	School buildings – Design and
Regional planning 362,	construction – New Jersey P59,
	School environment
Regional planning – Asia 283,	
Rehabilitation	707, P36, School environment – California –
204,	Berkeley
Religion	•
425, 426,	C72, School facilities – Planning
Relocation (Housing)	648,
636,	School facilities – United States
Rest homes – Recreation	742,
D68,	School farms
Restaurants	96, P94,
191,	School gardens
Retirement communities – Virginia –	170, 172, 265, 309, 348, 349,
Radford – Designs and plans	370, 397, 404, 405, 515, 520, 521,
D45,	567, 703, 781, C12, C13, C14, C36,
Ritual	C39, C47, C96, D31, D74, D156,
26,	M25,
Road markings	School gardens – Design
262,	D53,
Roads – Interchanges and	School gardens – New Jersey
intersections	P59, P79,
262,	School gardens – Ohio – Cleveland
Roof gardening	D162,
352, 621, P11,	School gardens – Study and teaching
Royal Botanic Gardens, Kew –	(Elementary)
History	C52, C53,
201,	School gardens – Sudan
Rural development projects – Sudan	P32,
P32,	School gardens – Sweden
Rural schools	D87,
P73,	School grounds
,	328, 729, C20, C30, C87, P68,
	School grounds – California –
San Francisco League of Urban	Berkeley – Design – Case studies
Gardeners	C72,
M5,	School grounds – Design
Sanctuary gardens	478, 707, C101,
104, 230, 246, 415, 506, 597,	School grounds Designs and plans
D97,	C76,
Satisfaction	School grounds – Psychological
D70,	aspects
	707,

School management and Self-help of people with disabilities organization P85, Sense (Philosophy) D87, School psychology 3, 671, Sense organs Science – Experiments 499, 666, C59, C61, Sense organs - Aging 151, Science – Philosophy 686, Senses and sensation Science – Problems, exercises, etc. – 4, 72, 98, 100, 151, 173, 174, 175, 176, 179, 202, 205, 295, 296, Handbooks, manuals, etc. 297, 298, 299, 312, 340, 470, 471, C15, Science - Study and teaching 492, 499, 594, 624, 655, 657, 658, 659, 660, 663, 664, 665, 666, 796, C86, Science - Study and teaching -Handbook, manuals, etc. Senses and sensation - Crosscultural studies D33, Science - Study and teaching - New 144, Senses and sensation - Testing York C35, Science - Study and teaching Sidewalks - United States - Design (Elementary) and construction C5, C6, C52, C53, C59, C95, 432, Shearer, Rhonda Roland D86, D112, Science -- Study and teaching M49, (Preschool) Sheltered workshops C15, 394, Science projects Shrubs C10, C105, 620, Siksika Nation Scientific recreations D88, C61, Screens (Plants) Slugs (Mollusks) 183, C31, Sculptors – Interviews Small business - United States M49, Seasons Small gardens - Germany -C91, Frankfurt am Main Self-esteem in children 317, D159, Snails Self-help devices for people with C31, disabilities Social behavior in animals 6, 7, 194, 195, P1, 785, Self-help devices for people with Social change disabilities - Great Britain - Catalogs 593, Social environment 148, Self-help groups 514, 475, Social groups

Streets – Missouri – Kansas City –
Design and construction – History
529,
Stress (Physiology)
D14,
Stress (Physiology) – Measurement
D34,
Stress (Psychology) D14,
Stress management 654,
Stress relaxation
D78,
Student movements – United States
D55,
Suburbs – United States – Planning
285,
Sustainable agriculture
737, C66,
Sustainable agriculture – Canada
459,
Sustainable agriculture – United
States
459,
Sustainable development
749, C66, D139, P40,
Table setting and decoration –
Psychological aspects
D106,
Tales – North America
C17,
Teaching – Aids and devices
C56, C57, D138,
Teenagers with disabilities – United
States – Attitudes
445,
Terra-cotta
423,
Territoriality (Zoology)
395,
Therapeutic community
121,
Therapeutics – Dictionaries
36,

Trails - Barrier-free design - United Urban agriculture - Africa 318, States 432, Urban agriculture - Africa, West Tree planting 694, Urban agriculture – Asia 328, Trees 283, Urban agriculture – Australia – 620, C91, Trees – Psychological aspects Hastings (Vic.) 585, P35, Trees - Study and teaching Urban agriculture – Bibliography P91, C20, Trees – Study and teaching Urban agriculture - Canada (Elementary) 477, P26, C29, Urban agriculture – Ghana – Accra Tropical plants 566, 452, Urban agriculture - Government Truck farming policy - Canada 157, 159, 319, 323, 403, 468, P26, 496, 497, 604, 605, 606, 700, P7, Urban agriculture – Kenya Truck farming - Great Britain P40, 247, Urban agriculture - Kenya - Nairobi 256, Truck farming - Handbooks, Urban agriculture - Oregon manuals, etc. Willamette Valley 495, Truck farming – Nigeria – Western 96, State Urban agriculture – Quebec 752, (Province) - Montreal **Tuberculosis** 584, Urban agriculture - Research -386, Tuberculosis – Therapy Canada 386, 477, Urban agriculture - United States D89, Underachievers - South Carolina Urban agriculture - United States -Handbooks, manuals, etc. Unemployed – England – York 572, Urban agriculture - Washington 637, 638, Union Settlement Community (State) – Seattle Garden (East Harlem, New York, 288, N.Y.) Urban agriculture - Zambia D136, D105, United States. Americans with Urban agriculture - Zimbabwe Disabilities Act of 1990 504, 705, M21, Urban animals - United States -Urban agriculture Guidebooks 96, 102, 198, 223, 725, 737, C86, 749, P53, P69, P78, P94, Urban beautification

P53,	527, 528, 561, 604, 605, 606, 690,
Urban beautification – Bibliography	691, 746, 802, C14, C60, C97, D42,
214,	M4, M40, P34,
Urban beautification – Maryland –	Vegetable gardening – Great Britain
Baltimore	247,
622,	Vegetable gardening – Great Britain
Urban beautification – New York	– History
(State) – New York	187,
P60,	Vegetable gardening – Handbooks,
Urban beautification – Pennsylvania	manuals, etc
– Philadelphia	495, C36, C74, C75,
114,	Vegetable gardening - Massachusetts
Urban environment	– Boston
D158,	769,
Urban gardening – United States –	Vegetable gardening – Michigan –
Anecdotes	Washtenaw County
342,	P61,
Urban gardening – Washington	Vegetable gardening – New Jersey –
(State) Seattle	Trenton
381,	D54,
Urban gardens	Vegetable gardening – Nigeria –
677, P63,	Western State
Urban geography	752,
286,	Vegetable gardening – Switzerland
Urban planning	702,
D46, D56, D60, D62, D73,	Vegetable gardening – United States
D87, D121, D132,	578,
Urban planning – California – San	Vegetable gardening – West Virginia
Francisco	– Tucker County
D164,	D109,
Urban plants – United States –	Vegetables
Guidebooks	C60,
C86,	Vegetables – Great Britain
Urban plants – Social aspects –	247, Vegetables – Marketing
Congresses 782,	319, 700,
Urban poor – Zimbabwe	Vegetables – Propagation
504,	P34,
Urbanization – Asia	Vegetables – United States –
283,	Marketing
Urns	468,
306,	Vegetation dynamics – Management
,	725,
	Video recordings for the hearing
Vegetable gardening	impaired
2, 56, 157, 158, 159, 171, 182,	M34,
192, 323, 400, 408, 443, 450, 451,	Virtual reality
	•

D90, D163, Vocational education Work environment D96, D81, Vocational rehabilitation Work environment - Barrier-free 385, 394, D96, P77, design - United States Volunteer workers in recreation – 53, Working-men's gardens Colorado P97, 581, 637, 638, 672, D82, D143, Working-men's gardens - Great Britain - History Water conservation - Australia -187, 542, Citizen participation Working-men's gardens - Italy -134, Turin Water gardens 93, 549, World War, 1914-1918 - Food Web sites supply - United States D90, 578, Wheatland Hills Retirement Center D45. Wilderness areas Youth - Michigan - Detroit 570, Metropolitan Area – Attitudes Wilderness areas - United States 763, Youth with social disabilities 305, 777, Williams Fork River (Colo.) -D96, Recreational use P97, Window gardening Zambia – Economic conditions 72, 329, 412, 724, 774, D136 Women 71, Women – California – Los Angeles -History 344, Women - Health and hygiene 384, Women - Psychology 16, Women college students - Kansas -Manhattan – Attitudes D108. Women farmers - Texas -Blacklands - History D140, Women in agriculture – Texas – Blacklands - History D140,

Work – Psychological aspects

ABOUT THE AUTHOR

Diana Farmer is a librarian at Kansas State University. She has a Masters of Library Science. In 1991 she became a subject bibliographer for Agriculture, Engineering and the Sciences. As part of her responsibilities she worked with the Department of Horticulture (as it was named at that time) to develop a collection to support the Horticultural Therapy program. Although her responsibilities changed over the years, her relationship with the Department of Horticulture, Forestry and Recreational Resources continued as has her collection building responsibilities.

She became a member of the United States Agricultural Information Network (USAIN) in 1991. She has been active in the organization—serving as President, as a Delegate to the USAIN Executive Council, and chairing or serving on various committees.